

Middlesex University Research Repository:

an open access repository of Middlesex University research

http://eprints.mdx.ac.uk

Shaw, Jennifer, 2004.

Police perceptions of their encounters with the mentally ill in the community.

Available from Middlesex University's Research Repository.

Copyright:

Middlesex University Research Repository makes the University's research available electronically.

Copyright and moral rights to this thesis/research project are retained by the author and/or other copyright owners. The work is supplied on the understanding that any use for commercial gain is strictly forbidden. A copy may be downloaded for personal, non-commercial, research or study without prior permission and without charge. Any use of the thesis/research project for private study or research must be properly acknowledged with reference to the work's full bibliographic details.

This thesis/research project may not be reproduced in any format or medium, or extensive quotations taken from it, or its content changed in any way, without first obtaining permission in writing from the copyright holder(s).

If you believe that any material held in the repository infringes copyright law, please contact the Repository Team at Middlesex University via the following email address: eprints@mdx.ac.uk

The item will be removed from the repository while any claim is being investigated.

POLICE PERCEPTIONS OF THEIR ENCOUNTERS WITH THE MENTALLY ILL IN THE COMMUNITY

JENNIFER SHAW

PhD 2004

POLICE PERCEPTIONS OF THEIR ENCOUNTERS WITH THE MENTALLY ILL IN THE COMMUNITY

1241

JENNIFER SHAW

A thesis in partial fulfilment of the requirements for Middlesex University for the degree of Doctor of Philosophy

2004

School of Social Sciences

Police Perception of the Mentally Ill in the Community

Contents

Acknowledgements	page 3
Abstract	page 4
Chapter One: Introduction	page 5
Chapter Two: Literature Review	page 16
Chapter Three: Methodology	page 90
Chapter Four: Findings – The Risk Agenda and the Police	page 121
Chapter Five: Findings – The Informal Management of the Mentally Ill	page 148
Chapter Six: Findings – The Service Response to the Mentally Ill	page 173
Chapter Seven: Discussion- The Police Experience of the Mentally III	page 206
Appendices	page 254
Bibliography	page 258

Acknowledgements and Dedication

In undertaking this research, several individuals and organisations provided invaluable help and support. The various police forces involved in the research provided me with essential information and assistance. Specific thanks and acknowledgment should be made to the two forces that additionally allowed their officers to be interviewed and to the individual officers concerned. They showed me considerable hospitality and patience. Thanks also to numerous friends who have 'lived' this thesis with me.

This work is dedicated to my parents, in acknowledgment of their support and guidance.

Abstract

This thesis aims to provide an examination of the police experience of mental illness. The thesis highlights several of the dilemmas of the police, the perception of their roles and the methods used by the officers to deal with the mentally ill in the community. The police have consistently faced social problems in the community, including mental illness. There have been many changes and reforms of mental health policy over the past twenty years. This includes the introduction of community care policy. It has been acknowledged that these changes have impacted on the police and numbers of mentally ill being dealt with by the police. This thesis involves an examination of previous research into the police and the mentally ill, the policy and social context of this involvement, including an examination of pertinent mental health policy, and the practice of the police when facing a complex social problem.

Operational officers from two police stations were interviewed to gain insight into their perceptions of the mentally ill. Senior, managerial officers also participated to provide an overview of the policies of several forces. Various issues were raised concerning the involvement of the police with the mentally ill. These included the dilemma of dealing with a vulnerable group in a climate where risk is at the forefront. The police adopt various methods for managing the mentally ill. The overwhelming approach, however, continued to involve the use of police discretion. The pragmatic use of discretion appeared to be the most useful mechanism, regardless of the policy and procedure of their specific forces.

Chapter One: Introduction

1.1 Introduction

The development of the police as a research subject has witnessed numerous themes and several central issues. One of these issues concerns the link between policy and police practice. What the police can and should do is often in variance to what they actually do in practice. This linked to what is also possible within their particular communities. Questions over the role and function of the police are similarly associated, such as the distinction between law enforcement and order maintenance and the notion of keeping the peace as a feature of police work and the many forms this takes [Neocleous, 2000].

The police fulfil many functions in society, including fighting crime and protecting people and property. Overall, they generally deal with and manage problematic situations. This also includes police involvement with the mentally ill in the community. Within this role and in other roles, they rely on the use of their discretion. It is this police tool that assists in deciding on action and practice. The police negotiation of their environment is also influenced by their understanding of their main roles and tasks. This examination of the role and practice of the police has proved to be a recurring theme [Reiner, 1992a, 1992b, Waddington, 1993, 1999].

There is a long history of the police association with the mentally ill and in dealing with social problems generally [Bean et al, 1991]. Questions arise regarding the nature of this police work and the appropriateness of their involvement with the mentally ill. As Reiner [1992b] states

"A perennial chestnut of debate around the police role has been whether the police are best considered as a force, with the primary function of enforcing the criminal law, or as a service, providing balm for a sea of social troubles." [Reiner, 1992b, page 139]

The 'culture' of the police is often the central feature that links several of the main themes surrounding police practice, whether it is the issue of discretion, the autonomy of the officer on the street or the translation of policy into practice. When examining the police world, therefore, different areas are highlighted. These areas provide an awareness into practice of the police and their involvement with both suspects and the victims of crime.

The mentally ill are a definite group who raise several issues for the police. These involve questions over how the police manage this group of mentally ill individuals effectively. One of the main areas centres on the care and control of the mentally ill by the police. Recent changes in mental health legislation have presented a complex framework for this involvement. These shifts have centred on various areas, including fundamental policy changes. Recent developments have included the introduction of community care and the transformation of mental health services. Equally, issues around personality disorder and the emphasis on risk have meant that the area of mental illness is increasingly complicated.

The need to manage the mentally ill in the community is demonstrated in the introduction of supervision registers, the Mental Health Act 1995, the proposals to manage and treat dangerous and severe personality disordered individuals, and the publication of the White Paper in December 2000, 'Reforming the Mental Health Act'. It is this wider social policy framework that influences police encounters with the mentally ill. However, mental illness is a nebulous concept, open to subjectivity.

Organisations, such as the police, have to manage these diverse meanings and respond appropriately. This issue is important when looking at the police experience of dealing with the mentally ill.

The police have additionally faced modifications in their operational structure. These include the development of community policing, the emphasis on performance indicators and increasing demand on resources. In light of these various external policy changes and internal alterations, it is useful to examine the police management of the mentally ill in the community.

1.2 Aim of the Research

The aim of this research was an examination of the police perception of mental illness and their involvement with the mentally ill in the community. It gathered evidence from the wider social policy context of mental health. It does not, however, ignore the experience of the officers and their practice within specific locations. Equally important was the 'official' policy response of police services to the issue of the mentally ill in the community.

Operational and managerial officers were interviewed to provide details of their involvement with the mentally ill. This also included examining the policy of various police forces and looking at how they co-ordinated police practice regarding the mentally ill in their communities. Several police forces were approached, with two agreeing to allow the operational officers to be involved in the research. Numerous forces aided the process by providing relevant documents and access to officers responsible for the force's implementation of policy on mental illness. The process involved both individual and group interviews. The research attempted to provide an overview of the officers' understanding and behaviour towards the mentally ill.

The research aimed to examine the process of police involvement with the mentally ill, from their first contact through to the police solution of the incident. A main objective was to learn the nature of the police involvement with the mentally ill and how they managed such incidents. This also involved examining what the police considered to be the most appropriate solution to the social problem faced. From this, key themes were identified, which highlighted some of the dilemmas faced currently by these police officers when managing the mentally ill in the community.

1.3 Rationale for the Research

The choice of this research problem arose from a variety of considerations, both professional and personal. My previous research into the experience of black mentally disordered offenders offered some insight and interest into the dichotomies of the criminal justice system generally, as both a control and a welfare agency. The police are at the first stage of the criminal justice process. They are, therefore, at a difficult but intriguing position. Issues around their ability to identify mental illness and their practice at the interface of the mental health and the criminal justice systems suggest that the police face various dilemmas. It was these aspects which further increased my interest into the practice of the police when involved with the mentally ill.

My employment in Ashworth Hospital also provided me with an awareness of the nature of mental illness and an understanding of the diagnostic and prognostic dilemmas associated with such apparent socio-medical constructs. In having a certain understanding of mental illness, what became more interesting were the similarities in experience between the police as a social agency and Ashworth Hospital. This was in terms of culture, external assumptions and public image. My knowledge of both the mental health system and the criminal justice arena was beneficial in understanding the

experiences of the officers in this study. While this recognition is apparent, the aim was to be as independent as possible.

The police with their culture, history and development are an intriguing subject for research. The various roles and functions, the uniform with its authority and power and the police as social agents all contribute to this fascination with their actions and practice. How officers interact with the public, their interpretation of incidents and the various situational factors that characterise their involvement with the mentally ill take on further significance. For to understand this experience of the officer is to provide a perspective on the practice of policing itself. The relationship of the police to their environment, therefore, involves examining their perceptions and reactions to the issue of mental illness in modern policing.

1.4 Outline of the Thesis

To question the role and tasks of the police in their relationship with the mentally ill is also to question the function fulfilled in modern society and their involvement in the social problems. What role the police complete in their encounters with the mentally ill and how they interact with this specific group are areas which highlight several issues.

This thesis attempted to examine the main issues relevant for the police in their dealings with the mentally ill. Questions occurred as to the core issues identified by the police in their everyday practice. The research looked at previous research into not only the police, but also mental health policy generally.

One cannot ignore the process of labelling and its influence both on the recipient and the professionals involved in the mental health system. This is within a climate of fear and a belief in the link between mental illness and violence. To try and understand the police understanding of mental illness, we need to understand the wider perspective and the social assumptions attached. When recognising these features, it is equally relevant to examine the treatment of the mentally ill, which can contribute to their involvement with the police. These issues are outlined in Chapter Two. This chapter provides a context for the current involvement of the police with the mentally ill and an examination of the key themes that have emerged. This includes exploring the meanings attached to the notion and perceptions of mental illness. Such areas as the problems of diagnosis, the current management of mental illness and the issue of risk are discussed.

Chapter three outlines the methodology and methods involved in the research. This not only outlines the research process and various issues involved in researching the police, but also attempts to outline a theoretical perspective from the work of Pierre Bourdieu. This particularly highlights his notions of 'habitus' and 'field'. These are suggested as potential perspectives on the practice of the police in their management of the mentally ill.

There are key issues raised that attempt to provide insight into the practice of the police. It could be suggested that the action of the police and their reaction to the mentally ill is the result of a 'habitus' or set of dispositions that guide them in their practice. It is also suggested that this may be a consequence of the socialisation process in becoming a police officer. Linked to the notion of police culture, Bourdieu's notions of 'field' and 'habitus' attempts to understand police action as a response to the dilemmas faced within a specific social context.

The findings from this research are divided into three main chapters. Two of the chapters deal with the discussions with operational officers, whilst the chapter six

considers the service response to the mentally ill. Chapter four examines the risk agenda and the perception of the police. This has consequences for how the police view the mentally ill. It also influences the subsequent action taken when faced with an individual with mental illness. There are dilemmas involved within this approach. This centres on the vulnerability of the mentally ill whilst maintaining an awareness of the risk issues involved. The police role as welfare agents means that officers have to use various methods to manage this conflict.

Chapter five outlines the informal methods of management employed by the police in their involvement with the mentally ill. Involved in this is the police use of their skills of persuasion and the predominance of the established police methods. Included within the management issues as discussed by the police is the consideration of outcomes. The police are aware that managing the situation not only involves the immediate factors involved, but also the placement of the mentally ill in more appropriate settings. This is far from straightforward.

Chapter six looks at the service response to mentally ill and their policy regarding mental illness. This primarily centres on the discussions with managerial officers and the relevant documents and policy developments. Five police forces participated in the research. What the findings point to are the diversity of responses and the different priorities of each police force. The reasons for such differences are various and highlight the complexity of managing the mentally ill in the community.

In the final chapter, the main issues raised from the research are highlighted. These include the police perception of the mentally ill and the methods employed to 'make sense' of the situation, such as the use of discretion. This is within the context of police policies and the influence of the police culture. Possible recommendations outlined in

the chapter focus on the practical aspects of managing the mentally ill in the community and emphasise certain areas, such as training and the development of local services.

1.5 Key Issues

There are several key issues that have emerged throughout the research, which additionally raise further questions. These include

- The continuation of the police role with the mentally ill
- The policies of the police
- The issue of risk, violence and care of the mentally ill
- The use of discretion and the informal management of the mentally ill.

The police continue to play a major part in caring and controlling the mentally ill in the community. It appears that officers still rely on the core skills of policing, including officer discretion. This is in light of the policies and schemes that have been developed to assist the police in this process. The police officers in this study maintain a pragmatic approach to policing the mentally ill. By using discretion, the police officer is an agent of interpretation and choice. Discretion guides and defines the actions of the officer on an individual level. This is also part of the culture of the police. Waddington [1999] illustrates the link between this culture and the existence of discretion. He goes on to state that,

"It derives from the discovery that police work is rarely guided by legal precepts, but that police officers exercise extensive discretion in how they enforce the law."

[Waddington, 1999, p287]

In attempting to understand the role of discretion, one is also attempting to understand the culture of the police. However, one should recognise that the definition of discretion is multi layered, dependent on attitude, experience and perceptions. Similarly, the notion of a police culture is not an all encompassing social construction, but is influenced by the particular force under inspection and their own ideological basis [Waddington, 1999]. How the police as an organisation transfer their main principles and philosophy is also an area of importance.

The issue of training as a method of not only imparting knowledge and awareness, but also influencing the actions of officers cannot be ignored. De Lint [1998] and Goldsmith [1990] illustrate possible ways this may be achieved. These include the use of individual judgement in a structured manner and the negotiation between the various echelons of the police in determining the ownership and acknowledgement of policy. The involvement of the individual 'street cop' [Reiner, 1992b] in policy development aims for a convergence between the objectives and expectations of the 'managerial' officers and the experience of the officer in the community. This raises several issues.

The police as an organisation attempt to answer the dilemmas and problems posed by the mentally ill in the community. To look at the organisational response is to look at the 'official' answer, directed from both governmental and local levels. The police as a service strive to formulate consistent and structural responses. These highlight what the police managers see as the main components in the management of the mentally ill and range from the encouragement of inter agency working through to individual designated police officers responsible for the issue of mental illness. The diversity of areas and the police forces within them is illustrated within the diversity of schemes and the objectives identified.

One has to question, however, whether these measures are practical and effective. The situations faced by officers in the community are complex and time consuming.

Officers are required to view each case individually, examining the various factors

involved in the situation. The subjective meanings of the individual officers involved in this process of assessing situations and finding suitable solutions takes on further importance when mental illness is suspected. The police negotiation of their social environment may still depend on discretion as a working tool.

In practice there remains a divergence between the experience and practice of the individual officer and the demands and the strategies of the organisation. As Grimshaw and Jefferson [1987] acknowledge, the methods of transferring policy to practice involve a variety of possible outcomes. There appears no smooth transition from policy to practice, but rather a complex process that involves the recognition of discretion as a maintaining force of autonomy for individual officers in the community.

The significant features identified by the police to their involvement with the mentally ill includes potentially contradictory elements;

- the presumed link between violence and mental illness
- the issues of managing the mentally ill, including the options available
- the police as social commentators
- their understanding of the dilemmas and vulnerabilities of the mentally ill in the community.

The 'symbolic mentally ill' identified by the police places emphasis on the above features. They are usually seen as both potential perpetrators and victims. The portrait of the mentally ill becomes a starting point for the police process. How they then proceed is dependent on a number of additional features, including local resources. The police experience of mental illness involves components evident in the police encounters in society generally. These include such features as the social and environmental context, the behaviour presented [and particularly 'strange' behaviour,

actions and events], the police priorities in situations and their perception of these encounters. The 'first aid' approach of policing the mentally ill is one aspect of the police experience. This has become increasing relevant since the advent of community care and the issue of personality disordered individuals in the community.

1.6 Conclusion

The police experience of mental illness points to the continuing role of the officer as a short term, problem solver. To be a police officer requires individuals to transform into information providers and negotiators of social problems. It implies an understanding of the dynamics that define the social problem. The officers additionally aim for attainable and identifiable solutions. The interface between the social environment of the police, their roles and tasks and the socio-medical world of mental illness implies a conflict of expectations, in interactions and in outcomes.

It is not unusual to remark on the diversity of the police role. It is a constant feature and area of discussion as to whether the police are involved in order maintenance or law enforcement [Morgan and Newburn, 1998, Waddington, 1993]. Questions additionally arise as to whether they should be proactive in the prevention of crime or remain responsive to societal demands and situations. The police have to examine how they react to these situations and the various approaches available, including the idea of community policing. These areas go beyond the scope of this research. What is raised, however, are several interesting areas of debate that highlight the complexity of mental illness and the perceptions of officers in managing this particular social problem.

Chapter Two: Literature Review

2.1 Introduction

Since the formation of the police service, officers have been involved in a variety of social problems, occurrences and dilemmas. Amongst these is the issue of the mentally ill in the community. Here, officers deal with a variety of problems via a variety of methods. The expectations on the police are various; they are expected to act as carers, as social workers, working as extensions to the mental health system, in addition to adopting a community and problem solving approach to policing an isolated group.

Consequently, the recent developments in mental health policy have several implications for the police; public perceptions of mental illness, risk and violence, the practice and expectations of services, and the management of diverse social needs in a climate of fear and pessimism.

The police have a principal role concerning the mentally ill and their inclusion in the criminal justice system. More fundamentally, issues around the police management of the mentally ill have become relevant. The police, aware of the rising numbers they come into contact with who are also mentally ill, realise that this contact,

"..is forming a significant part of modern policing as more and more hospitals and other institutions close in favour of community care alternatives."

[Roughton, 1994, page 4]

Evidence suggests that the effect of community care, linked to the wider policy of diverting the mentally disordered away from the criminal justice system, has been to provide the police with additional responsibilities and involvement in mental health policy [Berkeley, 1995, Roughton, 1994, Stephens et al, 1994]. It is a complex process, with police association with the mentally ill challenging their normally perceived role

as 'law enforcers'. It furnishes the police with an assortment of dilemmas. This includes acknowledging the role of the police within the mental health arena. For,

"The policehave to decide if they are peacekeeping or law enforcing, and question their occupational culture that omits references to welfare from its contents."

[Thomas, 1994, page 51]

Predominant issues throughout this exploration highlight the following:

- the very notion of mental illness and its definition
- the various mental health policy measures introduced
- the emphasis on risk in mental health policy seen in the suggested reforms of the Mental Heath Act
- the impact on the police of these policy objectives in every day encounters with the mentally ill and the management techniques they employ.

All this occurs under the dilemmas posed by issues of care and control. These points have become increasingly relevant as they highlight the complex nature of mental health, social policy and the implementation of objectives. The police form a central role as the only 24-hour service who have to manage these issues, regardless of training, understanding and willingness. It is suggested that the police provide a fundamental service for the mentally ill. The nature of mental health provision proposes that this will continue.

2.2 The Methodology and Scope of the Literature Review

When undertaking this literature review, various sources were used. These covered criminological and psychological references, in addition to areas of mental illness and policing generally. The databases included Aditus, Psychinfo, Psychlit, Cochrane, CINAHL, Medline, and the Kings Fund, with the following used as key words

Table One: Key Words

Police key words	Mental health and illness	Connected areas
	key words	
Police	Psychiatry	Bourdieu
Police authority	Madness	habitus and social field
Police danger	Mental health policy	Symbolic interactionism
Police culture	Mental illness	Culture
Police and force	Diversion schemes	Emotions
Police structure	Supervision registers	Attitudes
Police policy	Personality disorder	Symbols
Diversion	Community care	Interactions
Police uniform	Care programme approach	Interpretations
Police powers	Psychiatric admission	Roles
Discretion	Reform of the mental health act	Identity
Police rules	Mental illness and culture	
Arrests		
Community policing		

Articles and books were examined from 1960s to current research and publications. The aims for taking this approach included establishing an overview of relevant police research, key areas in mental health and social policy, and identifying key themes that have been consistent over the last thirty to forty years. The journals referred to include the following:

Table Two: Key Journals

Police and crime	Mental health and illness	Connected areas
American Journal of Police	British Journal of Psychiatry	American Journal Of
Behavioural Sciences and the	Community Care	Sociology
Law	Criminal Behaviour and Mental	American Sociological
British Journal of Criminology	Health	Review
Criminology	Health Service Journal	British Journal of
Criminal Justice Review	Hospital and Community	Sociology
Criminal Justice and Behaviour	Psychiatry	Culture and Psychology
Howard Journal	International Journal of Law and	Current Anthropology
Issues in Criminology	Psychiatry	Current Issues in
Journal of Criminal Justice	Issues In Criminological and Legal	Contemporary Problems
Law and Contemporary Problems	Psychology	Deviant Behaviour
Police	Journal of Advanced Nursing	Human Studies
Police Magazine	Journal of Forensic Psychiatry	Social Problems
Police Review	Psychiatric Quarterly	Sociocriticism
Stanford Law Review	Medicine, Science and the Law	Sociological Theory
	Mental Health Nursing	Sociological Quarterly
	Nursing Times	Studies in Symbolic
	Nursing Standard	Interactionism
	Openmind	Symbolic Interaction
	Psychiatric Bulletin	Theory and Psychology
		Urban Life

This table demonstrates the diversity of sources, including professional papers, in addition to more academic journals. It highlights the potentially wide-ranging themes when examining the area of the police and the mentally ill. While there are distinct areas, some overlap does occur. This is especially in terms of criminality and mental health. The main use of these journals demonstrates the predominance of distinct and singular research studies into subjects such as the police. When conducting the literature review, it became evident that the main contributions to the identified areas came from North America. The reasons for this may be various, including the nature of the areas examined and the databases used.

Several key books were referred to, such as Morgan et al [1998] and Reiner [1992]. However, in identifying the relevant areas it became apparent that journals and research papers offered more insight into developments, both in policing and mental health policy. The areas that were not primarily focused on in detail included police structure, issues around substance misuse and mental illness, police performance, abstract psychological issues such as role identity and objects, more detailed analysis of sociological explanations for behaviour, specific therapeutic interventions with the mentally ill and wider issues around the mentally ill and psychiatry. While an overview was required in several of the above areas within the defined areas of the research, it was considered that extensive reference could potentially detract from the main issues to be seen in the interactions between the police and the mentally ill.

2.3 The Context of Mental Illness

Mental illness as a medical disease carries with it connotations of instability, a lack of control and odd behaviour to such an extent that it generates apprehension in those experiencing and encountering it. It is often perceived as extremes, with little

consideration given to the wide range of illness associated with mental states [Leff, 2001a]. To distinguish between mental health and illness is to question the wider perceptions of normality [Fernando, 1991]. To understand mental illness, one has to place it within a cultural context. How one culture determines mental illness is not necessarily transferable to other cultures. It appears to be a matter of interpretation.

Mental illness cannot be seen in isolation from mental health and notions of normality. An important point centres on the inclusion of the individual in wider society, with the various social norms and displays of appropriate behaviour. This society will determine the type of mental illness, its response to the 'abnormality' and the general notion of mental health. Questions around the construction of deviance and abnormality centre on how we define what is permissible and what is not. Whether behaviour is seen as 'rule breaking' or, more sympathetically, as a manifestation of distress influences how that behaviour is then managed. The effect of this labelling is apparent for both society and individual alike [Morrell, 2000]. One also has to recognise that mental illness has to be understood within an historical framework [Cohen, 1988, Fernando, 1995]. The history of mental health institutions and treatments prescribed are examples of these changing perceptions. The closure of psychiatric hospitals and care in the community are recent advancements in mental health practice. The combination of ideologies, culture and history cause difficulties in establishing a constant understanding and definition of mental illness and mental health.

The distinction between mental health and mental illness is one that is not only difficult to determine, but involves subjective definitions of normality. By definition mental health determines mental illness [Fernando, 1991]. The correlation of these two notions attempts to assess how optimum health can be achieved and maintained. To

provide an individual with a psychiatric diagnosis and prognosis is to make a social judgement, rather than an objective and medical assessment of their condition [Johnstone, 1989]. As there is no definitive formal physiological measure of mental illness, any conclusions and determinants of illness are achieved through descriptions and judgements. Attempts to minimise the internalised procedure of diagnosis, to provide a semblance of scientific objectification in mental illness diagnosis and to instigate a viable and clear system have resulted in the implementation of universal schemes of identification.

The formal systems of diagnosis are embodied in the Diagnostic and Statistical Manual of Mental Disorders [DSM] and the International Classification of Diseases [ICD]. Kirk and Kutchins [1994] outline the initial claims of the DSM system, of its insistence in a scientific foundation through evidence and the belief in its' diagnostic reliability. The psychiatric profession has viewed DSM as presenting an image of legitimacy and power via established language, research and uniformity [Kirk and Kutchins, 1994]. Any argument that expounds the effectiveness of such a system of diagnosis, and its subsequent influence upon the treatment options available, has to be aware of the changes that have taken place within DSM. In the course of its history, hundreds of revisions have occurred, each claiming to increase its level of scientific credibility. In questioning the revised editions of DSM, it is noted that

".. unless past claims by proponents of DSM are scrutinised in light of the scientific evidence we are unable to evaluate the extent to which science and systematic evidence support the process of continual revision."

[Kirk and Kutchins, 1994, page 72]

In assessing the reliability of DSM it becomes apparent that, as with other methods of assessment, it is prone to interpretation and reinterpretation and equally dependent on

the clinician. The research by Kirk et al [1994] indicates that not only are the claims of objectivity, and hence scientific reliability, premature, but that questions of methodology need to be addressed.

2.4 The Role of Psychiatry

It has been argued that psychiatry will explicitly control individuals who do not fit the collective sense of normality [Fernando, 1995]. Johnstone [1989], following the social deviant labelling arguments of Goffman [1961] and Scheff [1966], sees psychiatry and the use of the medical model as being the main instruments in labelling deviance as 'illness'. In doing so, it effectively and legitimately controls those labelled. Psychiatry is also determined by, and follows, the social and ideological climate. It acts in a manner relevant to the period and in conjunction with the perceived norms. This is an argument put forward by others, including Francis [1996] who maintains that the medical and welfare establishments have the aim of 'policing' and supporting the social order.

From its inception, psychiatry has acted as a means of social coercion and constraint [Fernando, 1995, Rose, 2001]. The use of the medical model to justify the control of those labelled deviant, different or irrational is a point that is often used in the critique of psychiatry. While there has been a recognition of the influence of social factors, there is still an over reliance on the medical model. This traditional view of mental illness diminishes the wider issues involved in constructing and defining mental illness. The emphasis is firmly placed on biological and disease notions of illness [Littlewood, 1991].

This presents difficulties when looking at cultural distinctions in the presentation of mental illness and resulting treatment modalities. What is apparent in Littlewood's [1991] argument is the importance of the moral values inherent in psychiatry. Included within this argument is the pattern of relationships between, what he terms, the 'colonial' psychiatrist and 'subject peoples'. The consequence of medicalising episodes and events is to ignore cultural and individual experiences. Psychiatry, confirming the arguments regarding the use of DSM, cannot be viewed as a 'pure' science, with objective measures and criteria. For,

"The attempt to exclude social meaning and context from psychiatric theory itself does not automatically render our discipline value free and universal. Indeed, the reverse."

[Littlewood, 1991, page 700]

The supposed universality of psychiatry's theories and methods poses obstacles for the implementation of a reliable trans-cultural system of diagnosis and treatment. In the pursuit of the general and transferable notions of illness and disease, psychiatrists often have a firm belief and bias towards ascertaining similarities across cultures [Kleinman, 1987]. However, evidence from other studies [such as the World Health Organisation's examination of schizophrenia] demonstrates differences between cultures rather than clear similarities. As Kleinman [1987] claims, it is the professional beliefs that accentuates similarities and diminishes what appears to be culture specific.

The framework of mental health practice demonstrates an imbalance, both in diagnosis and legislation. It favours the professional status and opinion of the medical profession to the potential detriment of patients. As Crepaz-Keay [1997] states,

"Any legislation which relies so heavily on something as subjective as a psychiatrist's diagnosis of mental disorder has no place on the statute books."

[Crepaz-Keay, 1997, page 5]

An appraisal of the medical model, with the relative autonomy of the psychiatrist, has criticised the subjectivity enshrined in such an approach. The remarks of Crepaz-Keay

[1997] echo the anti-medical focus of much of the critique of the system. In particular, the legislation demonstrates the difficulty in defining and balancing the rights and needs of the patient against medical perception, opinion and behaviour. It transpires that increasing restrictive measures, such as supervision registers, serve the domination of psychiatry and the medical system generally. As Dallaire et al [2001] argue, the systems of criminal justice and mental health preserve their role as maintainers of social order. They continue to justify their position as guardians of social normality, partly in response to mental illness being seen as "somewhere between 'badness' and 'illness'" [Dallaire et al, 2001, page 133].

The perpetual image of the mentally ill person that has dominated the popular view is one of the dangerous schizophrenic, considered too 'unstable' to be maintained in the community [Swanson et al, 2000]. There is also a stronger element involved, for

"society fears people who are unpredictable or incomprehensive far more than those who are simply dishonest career villains." [Murphy, 1991, page 198]

It is this fear and paranoia of mental illness and the mentally ill which has contributed to their detachment from wider society and the imposition of controls. It is a fear that does not appear to have any firm grounding in reality.

2.5 The Fear of Violence and the Risk Agenda

The combined perceptions of fear, danger and mental illness have an extensive history. The westernised idea of mental illness is associated with a number of negative beliefs [Cohen, 1988]. It is also surrounded by the specific view that psychiatric patients are both violent and dangerous [Department of Health, 1998, Neuberger, 1997]. In response to this fear, assessment and treatment options are often designed to contain, becoming more punitive rather than providing care and assistance.

Recent ideas of mental illness have tended to connect mental illness with violence. This has led to the concentration on risk assessment and risk management in clinical practice [Mullen, 2001]. These issues imply a balancing of public safety with individual human rights [Szmukler, 2001]. This prominence on public safety is evident in the reforms of the 1983 Mental Health Act [Birmingham, 2001, Home Office and Department of Health, 2000]. The current risk agenda has not only changed the perception of mental illness, but has also redefined how that mental illness is managed. The concept of risk contains many preconceived ideas, notions and procedures. Essentially, it is a clinical and legal paradigm [Dallaire et al, 2001]. There is a danger for the clinician that if any risk becomes real, the condemnation of their actions will be immense. As Mullen [2001] states, a blame culture is active within the mental health field. This is coupled with

"..increasing punitive attitudes towards deviance of all kinds.
...the mentally disordered all too easily are placed into the role of potential perpetrator in need of containment with the mental health professional being held responsible for ensuring such control"

[Mullen, 2001, page xv]

As definitions of mental disorder change over time [Busfield, 2001], the current emphasis on risk has to be seen in light of recent public concerns and moral panics [Morrall, 2000].

The need for assessment and management to contain and restrict serves several functions. It enables the removal or distancing of the cause of this fear, thereby minimising anxiety. Only when there is a serious incident is this paranoia prompted [Peryer, 1995]. This reinforces the public panic towards the mentally ill, reaffirming the belief that the mentally ill now pose an increased danger [Neuberger, 1997]. While 'danger' is a frequently used term, there is limited consensus as to what constitutes

'dangerousness' or 'dangerous behaviour'. The link between mental illness, dangerousness and violence is firmly ingrained into the public conscious [Harrison, 1997a]. This is reinforced by combining dangerousness and risk with unpredictability [Dallaire et al, 2001]. As a consequence such connotations have powerful influences on services, perceptions and individuals. This frequently distorts reality.

When looking at the overall level of violence in society, the amount committed by the mentally ill is small [Boyd Report, Confidential Enquiry into Homicides and Suicides by Mentally Ill, 1994,]. The number of homicides committed by psychiatric patients is also relatively slight when compared with the overall number of six to seven hundred murders per year [Boyd et al, 1994]. Munro et al [2000] confirm this finding, noting that the rate of homicides by the mentally ill has shown no increase since the implementation of community care. Leff [2001b] goes further and argues that there has been a reduction in the level of violent acts committed by the mentally ill.

The major change has centred on the public perceptions of increased risk. This illustrates that the fear of violence and serious offending by the mentally ill is greater than the reality [Eaton, 1996]. Underpinning these areas is the public necessity for, what Bynoe [1992] terms, 'certainty and absolute security', an assurance that cannot be effectively achieved. The lack of a realistic foundation in the portrayals of the mentally ill, however, does not alter the perceptions concerning mental illness. It can often carry associations with criminal activity especially with involvement in the criminal justice system. Laberge and Morin [1995] note not only the increased numbers of mentally ill involved in the criminal justice process, but also the subsequent increase in the criminality of the mentally ill, which is further confirmed by Bonovitz et al [1981].

Any discussions surrounding the mentally ill and crime, however, involves an examination of the inferred dichotomy between the concepts of 'mad' and 'bad'. In some instances, there is a clear distinction. The mentally ill are not conscious of their actions or behaviour and are not fully aware of the consequences of any actions or offences committed. Subsequently, they may not be viewed as corrupt or criminal. This distinction, however, does not appear in the cases of those with personality disorder, where a myriad of images and assumptions are produced. This particularly evident when discussing the case of the 'psychopath'.

The notion of the 'psychopath' has social and clinical implications [Reed, 1996]. The introduction of proposals for the indefinite detention of severe personality disordered individuals is based on perceived risk, rather than actual behaviour or offences [Mullen, 1999]. The fear of psychopathy and personality disorder, in its various guises and categories, has contributed to a development in risk management and the incarceration of individuals based on assumptions of what they might do. While in the domain of mental health services, the suggestions put forward combine notions associated with the criminal justice system, primarily of anti social behaviour and high levels of risk.

Complications around notions of dangerousness and violence make the overall situation difficult to assess. The debate on mental illness and violence and the lack of agreement concerning definitions has repercussions not only in possible outcomes for the mentally ill, but also in the wider perceptions of mental illness. The part that the media play in accelerating and reinforcing these views is important to consider [Busfield, 2001]. While it is not fair to argue that the media as a whole have helped to form distorted images, a significant majority have contributed to the image of the mentally ill as potentially dangerous [Linehan 1996b]. The effect of this discrimination on the

mentally ill is seen in the combination of the media's statements with public assumptions through to the actions of legislators and clinicians. It corroborates the view that the mentally ill require some form of control [Swanson, 1996].

This ultimately reinforces the maintenance of the mentally ill away from society, making them invisible. Swanson [1996] goes as far to state that the wider community is not prepared to accept these individuals, a point partly illustrated by the failure of community care policy. It is not only the discerned dangerousness or the inadequacies of risk assessment in care in the community practice that has produced a sense of failure [Francis, 1996]. A contributing factor has been the relatively recent exposed nature of mental health policy. This has helped present a distorted picture. As Francis argues,

"the comfortable walls which separated madness from the community for two hundred years are beginning to crumble."

[Francis, 1996, page 4]

The erosion of the distance between the mentally ill and the outside community has resulted in making them visible [Munro et al, 2000]. Whatever the result of this visibility and increasing reliance on risk assessment, the mentally ill have to be seen as equally vulnerable. They face an array of discriminate treatment regimes aimed at control and segregation.

2.6 The Vulnerability of the Mentally Ill

Much research has examined the experiences and situation of the mentally ill [e.g. Harrison et al, 1989, McGovern and Cope, 1989]. It tends to focus on the factors that led to admission and the utilisation of certain sections of the Mental Health Act 1983. It is clear that the reliance on the use of drugs, in hospitals and in the community, and the deaths of psychiatric patients illustrate the predicaments many experience [Rickford, 1995b, Rogers et al 1993, Teevan, 1996, Waters, 1995b, Whiteley, 1996a]. The

restrictions in consent to treatment, the lack of information provided to patients [Rogers et al, 1993] and the overall disdain placed upon mental illness have left many without the ability to question and determine their own situation.

One of the most constraining elements of mental health legislation centres on the use of compulsory detention. Over a five-year period, detentions under the Mental Health Act 1983 increased by forty five per cent [Ogden, 1997, Whiteley, 1997c]. One feature has been the increase in the number of men detained, especially under Part II of the Act. The rationale for this may be partly as a consequence of the failure of community care. However, this explanation is too simplistic [Ogden, 1997]. While it is a contributory factor, there are other elements that have influenced the rise in formal detention.

The decrease in the number of hospital beds available for the assessment and treatment of mentally ill patients has meant that patients need to be seriously ill before they are admitted. This has meant that the hospital's atmosphere is far from caring or beneficial for treating mental illness [Ogden, 1997]. The result of this has been to make patients reluctant to be admitted on a voluntary basis. They reach a level of crisis, a level that may demand the attention of the emergency services, including the police. The professionals, aware of public perceptions, are reluctant to then discharge people into the community.

There is an additional difficulty in discharging patients when there is a limited supply of adequate housing and services that can maintain the mentally ill within the community [Leff, 2001b]. In this sense the policy of community care has been a factor in this increase. For the increase in the number of men being detained, Ogden [1997] argues that the power of the public view and its association with violence has meant that more are being detained as an antidote to this concern. As Leff [2001b] argues, this

perception, in conjunction with the realities of mental health policy, has provided a rationale for explaining the failure of community care. How this 'reality' for the mentally ill differs from their assumed dangerousness, instability and incapacity can be seen by the treatment imposed. Any treatment is fraught with complications, contradictions and uncertainties. The dichotomy between care and control demonstrates the problems in balancing safety with care.

What exacerbates the situation of the mentally ill and increases the difficulties for agencies are economic and social factors [Leff, 2001b]. Mental illness cannot be merely seen as a medical concept, to be treated via medical solutions. A wider picture of the socio-economic factors has to be taken into account, including homelessness and unemployment. A number of studies have examined the social and demographic features of the mentally ill who come into contact with the criminal justice system. They have identified significant areas in their respective study groups.

Research examining the use of section 136 notes the demographic features of identified groups of mentally ill [Rogers and Faulkner, 1987, Roughton, 1994]. A significant proportion are homeless, unemployed and likely to be relatively young, usually between seventeen and thirty-five years old. Psychiatric involvement and the use of in-patient facilities are equally relevant themes, with many experiencing previous in-patient treatment [Rogers and Faulkner, 1987]. Robertson [1988] confirmed that those with schizophrenia also suffer from grave social segregation, with many experiencing limited social support. However, these examinations do not assess the full extent of the involvement of the mentally ill in the two systems. This is no clear indication of the numbers involved, receiving formal or informal assistance. The exact numbers of

mentally ill processed or the circumstances that lead to such involvement may not be extensively known.

The situation of black mentally ill people, furthermore, demonstrates a more acute form of discrimination. They experience austere forms of treatment, both medically and socially [Bhui et al, 2003]. They are subject to a process of labelling and stereotyping that not only stigmatises their mental illness, but also includes assumptions around potential criminality and ethnicity. Francis [1993] argues that black people are extensively over classified as mentally ill; they are over five times more likely to be diagnosed than the indigenous British population. There is a reliance on stereotypes to classify and categorise the black and ethnic populations, with images of hostility, violence and danger [Wilson, 1997]. Therefore,

"Perceived differences between racial and ethnic groups conjure up much racial imagery that is unconsciously introduced into social encounters."

[Bhui, 2003, page 10]

The use of compulsory admission for black people averages sixty per cent, where they are also more likely to be detained as 'offender patients' [Cope, 1990, Sashidharan, 1994, McGovern et al, 1987]. This is particularly the case for young black males, where the rate is twenty five times that of the comparable white group [McGovern et al, 1987]. The black mentally ill are more likely to have access to services via non-medical agencies, such as the police. They are more likely to face not only increased use of medication, but also higher doses [Sashidharan, 1994]. The argument put forward is that these experiences are a result of the assumptions around race and madness, reinforced by the use of explicit coercive psychiatric practices.

We cannot ignore how mental illness is perceived and the resulting management of those labelled. There is a disparity between the general view of the mentally ill and the reality for many diagnosed. The vulnerabilities of the mentally ill are secondary to the apprehension around danger, violence and risk. The image of the mentally ill has allowed various developments in practice to emerge. Since the advent of community care, several new initiatives have continued to place emphasis on risk and the management of challenging behaviour. The consequences are wide ranging. To fully understand the implications of recent mental health policy, it is beneficial to examine developments in legislation and practice.

2. 7 Mental Health Policy

Mental health policy and practice has changed significantly in the last thirty years.

From the endorsement of community care, the introduction of supervision registers, the development of community mental health teams, the modernisation programme of the NHS to the launch of the National Service Framework and proposed reforms of the Mental Health Act, there have been fundamental changes in the ideology and provision of services. Several additional themes have equally transferred into mental health policy, notably modernisation, social exclusion, partnerships and service user involvement [Higgitt et al, 2002]. The rise of mental health practice in recent decades as a controversial area of social policy can be witnessed in the presentation of community care.

2.8 Community Care Policy

Community care conceptualised the ideology of several generations of policy makers. It covers a divergent array of objectives and arrangements. Its foundation and development towards its current form can be traced back to the administrations of the

early 1950s and 1960s [Goodwin, 1990, Hadley and Clough, 1996]. Community care policies are also connected to the idea of a 'welfare state', with services being based on equality of distribution and delivery. With the formulation of policies and services to cover wide sections of public life, the concept of community care had not only the opportunity to expand, but also to verify itself as a major ideological and practical solution to a variety of social dilemmas.

Community care has a contentious history. Several key components are identified enabling its expansion in mental health policy [Goodwin, 1990, Hadley and Clough, 1996]. These elements range from the emergence of new types of medical treatment for the mentally ill and the potential cure of mental illness to more fundamental issues, such as the growth in resources. On a practical level, this move towards community care envisaged an increase in staff and resources to coincide with the wider expectations of public services. The significant feature of these changes included the integration of mental health into the wider health and general welfare systems [Goodwin, 1990]. This was also linked to the recognition of mental disorder as a social problem, which was receptive to medical involvement and solution. The reliance on the medical model in the community was further heightened by the use of new drug treatments. Underpinning these approaches was the principle of 'normalisation' or 'social role valorisation' [Hadley and Clough, 1996, page 11]. Included in the political, ideological and economic climate was the awareness and belief in consumer choice. By integrating services, a process was envisaged that allowed for more independence and social integration in the community.

However, other ideological and philosophical intentions surrounded community care. These rested on the basic conjectures concerning the care of the mentally ill. This was reinforced by an increasing emphasis on the discharge of patients from mental hospitals. The main critique of the treatment of mentally ill centred on institutionalisation and the arguments of the 'anti psychiatry' movement. It primarily raised issues concerning the detrimental effect of mental institutions. This realisation and the publicised episodes of cruelty against patients challenged the justification of psychiatric hospitals [Goodwin, 1990]. A critique of the welfare state at the end of the 1970s and the beginning of the 1980s pointed to a general acceptance of community care as a distinct policy objective. The ideology of the Conservative government in the 1980s further allowed community care to be firmly established within a wider political framework. The impetus for community care, therefore, increased.

The culmination of this shift was the National Health Service and Community Care Act 1990. By questioning the role of the state as the main provider of social and health care, the ideology of mental health policy reinforced the principles of community care. It aimed to provide justification for community care policy, while appreciating the previous decades' failures. The fundamental principles behind the 1990 Act included choice, empowerment and partnerships between clients, service providers and agencies [Vaughan and Badger, 1995]. The central objective of community care was to enable clients to live as normally as possible by supplying services that were not only flexible, but also based on a concept of minimal intervention [NACRO, 1991, 1993b].

The process designed to put community care into practice involved care management in social services and the care programme approach in the health service. The systems encouraged a more comprehensive assessment process, intending to meet the needs of

the client effectively. Many of the elements of the care programme approach previously existed within the health service [Hamiliton and Roy, 1995]. The inclusion of such components as clinical and social plans and user consultation introduced 'fundamental changes' [Hamiliton and Roy, 1995, page 35]. Golding [1996] adds that this approach ensured that people with chronic and recurring mental health problems were identified and provided with the most appropriate package of care.

The replacement of a relatively uncomplicated system of mental health care provision with a large entangled administrative and financial organisation led to difficulties in managing care in the community for both social service and health staff [Hadley et al, 1995]. The result of the 'fragmentation of responsibility and funding' has been to leave the mentally ill with little effective service provision [Hadley et al, 1995]. As no lead agency has distinct responsibilities, the question of accountability is obscured. With this lack of direct accountability, the issue of definite boundaries compounds the workings of community care.

The care management and care programme approaches were recognised as the main constituents of this situation, with similarities in approach and reproduction of work [Hadley et al, 1995]. The situation has, therefore,

"...also accentuated the differences between health and social services teams and [has] probably made joint working more, rather than less difficult."

[Hadley et al, 1995, page 1557]

Prior to the formation of current policy it was acknowledged that the distinction between health and welfare was difficult to maintain. Whether such a distinction was considered necessary is debatable. Nevertheless, the philosophy of multi agency work today encourages a convergence of agencies' objectives and working to the benefit of

the client and their needs. The aim of improving agency co-ordination was an important theme in the foundations of community care [Goodwin 1990].

One factor that has intensified the situation for authorities is the obligation to provide care for the mentally disordered offender [Reed, 1992]. The proposals of the Reed Report [1994] and the resulting policy of diversion from the criminal justice process have exacerbated the situation, placing pressure on the health service [Deahl, 1996]. Nevertheless, these procedures result in increasing the numbers of individuals who come into contact with the police, as limited resources fail to maintain a substantial level of care and service. The desired effect of diverting the mentally ill away from the penal system is limited by the lack of provisions outside the system. Such inadequacies in policy have achieved the opposite result by attracting more mentally ill to the police in the search for care and assistance [Roughton, 1994].

Those who come into contact with diversion schemes often perceive the police as an agency where they can access treatment. These actions involve the police inadvertently into the process of caring for the mentally ill, bringing them to the forefront of the widely publicised care 'crisis' [Bradley-Taylor, 1996, Rickford, 1995a, Ritchie et al, 1994, Whiteley, 1996b]. The result of this has been a reappraisal by the police of their methods of work with the mentally ill and with other agencies.

There are shortfalls between the expectations of community care policy and the reality of resources. There is an acknowledgement that community care has failed to reach its objectives [Leff, 2001b]. One criticism has been the lack of community and the limited emergence of services. The very notion of community carries diverse opinions and definitions. It also has implications for how psychiatry and the mental health system operate within that community [Rose, 2001].

The main criticism relates to the fact that the closure of psychiatric beds has not been matched by adequate services within the community [Healy, 1997]. A survey carried out by the Royal College of Psychiatrists found that bed occupancy in many hospitals was continuously above 105 per cent. Leff [2001b] notes that there now remain only fourteen specific psychiatric hospitals out of a 1975 total of one hundred and thirty. What has developed in response are more diverse systems of care, incorporating community mental health teams and crisis intervention. The fact that mental health policy is perceived to have failed, regardless of initiatives and developments, has led to further proposals that aim to 'tighten up' the uncertainties in the system.

2.9 Developments in Mental Health Policy

With many conflicting and demanding expectations placed on community care policy, it was evident that a form of resolution was required to stabilise the framework. These additional measures have the effect of developing control mechanisms which were once in the domain of the hospital [Rose, 2001]. These changes impinge on the mentally ill and their rights. The reluctance to question the foundations of community care places pressure on the introduction of such measures as supervision registers, supervised discharge and the reorganisation of the mental health care system to provide solutions to many of the problems identified.

The reasons for introducing a collection of measures to supplement and expand on community care are various. Some, such as the Mental Health Act 1995, have been partly in response to the inquiries into homicides by care in the community patients. The suggestions and proposals concerning the structure and practice of mental health policy aim to establish a more efficient system, utilising existing services and

diminishing the voids in provision. These comprise notions of risk, supervision and intensive treatment.

One process that has emerged in answer to these issues has been the introduction of supervision registers [1994]. One benefit of the register is the dissemination of knowledge between agencies regarding specific clients. The focus, however, centres on both the identification of the serious mentally ill and the maintenance of suitable treatment. Those to be included on the register are deemed to be at risk of harm to themselves or others. It was originally perceived that there was a discrepancy between those needing treatment and those in contact with services, partly due to the inadequacies of community care. Again this included a belief in the potential risk of a minority of mentally ill individuals [Bean, 2001]. The need for firm controls can be seen within the context of 'community safety' [Bean, 2001]. This and subsequent measures are a response to this requirement. Within this debate, several areas of concern appear when this mechanism is examined.

Coffey [1995] raises issues concerning the power relationship between client and professional and the possible dilemmas such a scheme imposes. The use of the power in the implementation of the register has ramifications for the concept of care. The profession of psychiatry, in particular, reinforces its position as social control agents [Coffey, 1995, Cohen, 1988, Fernando, 1991, 1995]. With the introduction of supervision registers, the imbalance of power and the administration of tighter controls over the client/patient continue. Advocates of the register have argued that such a process would not only protect the public by maintaining closer contact with the patients, but would assist in targeting the right services for the patient. Yet there is anxiety over the possible infringement of the patient's civil rights and the encroachment

of confidentiality that may arise with information passing between agencies more frequently. What makes the situation complicated, however, are the unclear criteria for inclusion on the register and the very notion of risk [Bean, 2001, Bindman et al, 2000]

Since the introduction of the register, its function and effectiveness are questionable. Bindman et al [2000] have concluded that, while all Trusts have implemented local registers, there is extensive variation in practice. More disturbingly it has failed in its initial intention; no definitive group of mentally ill are identified through its use and many of those placed on the register have no documented record of risk indicators [Bindman et al, 2000]. Issues around the assessment of risk continue, particularly in light of the limited developments in risk assessment. The benefits of supervision registers have been neglected in response to the preoccupation with risk and fear of violence. The fact that services may now dispense with the register in favour of a more thorough care programme approach illustrates that these mechanisms often have insurmountable problems [Bindman et al, 2000] and have limited value in clinical practice [Bean, 2001].

In conjunction with supervision registers, further legislative answers were developed, chiefly the Mental Health [Patients in the Community] Act 1995. Various obligations were placed on health and social services to ensure that patients discharged from hospital continue to receive the necessary treatment. The aim of 'supervised discharge' is to ensure a continuity of treatment and maintenance of contact with the mentally ill perceived to be in need of supervision. The ideas put forward by the 1995 Act were considered for a substantial period, prior to implementation. The particular notion of compulsory care in the community has existed for many years. In 1986, the Mental Health Act Commission put forward this proposal. In 1987, the Royal College of

Psychiatrists suggested the introduction of 'community treatment orders', reinforcing this with new recommendations in 1993 [Coffey, 1996]. These proposals are close to those put forward in the final act. Equally the principles behind the measures are reinforced by the suggested reforms of the 1983 Mental Health Act [HMSO, 2002].

Again the Act, and in particular supervised discharge, is characterised by variation in practice. Since 1996, the numbers being placed on this order have remained relatively small [Bean, 2001]. It does not address the fundamental issue of resources. While it is important that patients receive the necessary treatment, imposing more controls and constraints does not equate with effective services. Issues concerning supervised discharge echo the concerns shown with the implementation of supervision registers. The particular section that attracted consideration is the power to 'take and convey'. This power can be activated when the patient fails to fulfil any of the requirements of their discharge plan. This also influences the role of the police with the mentally ill. The incorporation of the police in the implementation of this power potentially occurs when the supervisor believes that the patient is reluctant to comply with the conditions, and nominates the presence of the police to assist in the patient's removal to their place of work or return to hospital. This section places an unwarranted, controlling aspect on the mentally ill, reasoning that they,

"...will be subject to the 'power of arrest' for no apparent purpose."

[Harris, 1995, page 5]

Both the supervised discharge and registers personify the importance of managing the mentally ill in the community through close supervision and monitoring. Yet they relinquish several of the expectations of a balanced mental health system. Rather it transpires that this process of supplementing existing procedures with further reform will continue;

"In a few years time, there will be complaints that the existing system is not tight enough and allows certain types of patient who are considered dangerous to slip through the net. The demand will be for new controls for this group."

[Bean, 2001, page 76]

The climate of mental health provision has been symbolized by repeated revisions and embellishments. The changes in mental health policy have not merely been in the legislative context. They have included modernisation programmes and organisational reforms.

It was evident that the national health service generally and mental health particularly required some form of modernisation. The National Service Framework [1999] aimed to provide an integrated system, encouraging partnerships between a variety of agencies and services [Bowler et al, 2001b]. The introduction of crisis intervention teams equally altered the provision and delivery of services.

Recent policy developments have also extended the remit of many services, particularly in the area of personality disorder [Department of Health, 2003]. While the main proposals are 'implementation guidelines', they urge services to initiate clinical models for the exclusion of individuals with personality disorder in mainstream services. This has been accompanied by resources and funding aimed at the development of specific services for those with personality disorder [NIMHE, 2004]. Equally the changes to the Mental Health Act 1983 have potential ramifications for mental health services and patients.

2.10 The Reform of the Mental Health Act

When examining the development of mental health policy, one pertinent factor is identified. Any suggestions for change and reform occur after one or two incidents,

usually homicide, which involves an individual with mental illness. In the case of the reforms of the mental health act it is the case of Michael Stone in the late 1990s. Michael Stone was convicted of the murder in Kent of a mother and daughter. He was seen as suffering a 'untreatable' personality disorder by professionals, after frequently presenting to services. Criticism was targeted at these professionals and their refusal to treat Stone as a result of his diagnosis [Steele, 1998b]. What raised concerns was their inability to manage individuals who posed a high risk and were seen as unreceptive to treatment.

The latest developments continue to stress the risk of the mentally ill and the need to control their actions and behaviour. While supervision registers and the 1995 Act targeted a specific section of the mentally ill, the proposals for the reform of the 1983 Act extend the remit of risk. The White Paper, 'Reforming the Mental Health Act' [HMSO, 2000] argues for new structure; a legal framework to implement the compulsory treatment of individuals in the community and the introduction of provisions for those considered high risk. The paper acknowledges the difficulty in defining mental disorder and the existing emphasis on the perception of treatability. It goes one step further in arguing for the transference and maintenance of the powers evident within the hospital environment to the community.

The potential realisation of compulsory powers in the community for mental health practitioners and the enforcement of treatment compliance are responses to the fears generated by the visibility of the mentally ill in the community. The dilemmas posed if such legislation is evoked are various, not only for public sector agencies, but also the rights of the patients themselves.

The proposals concede that, since the demise of psychiatric hospitals, the organisation of mental health provision has changed. The place for treatment occurs principally in the community [Grounds, 2001]. It therefore seems sensible to increase the powers in the community to reflect this shift. However, in light of the criticisms of community care and the nature of mental health policy generally, there is a growing suspicion towards these suggestions [Goodwin et al, 2002, MIND, 2001]. They focus on the familiar balancing of patients' rights with public protection. As with the proposals for the preventive detention of those with personality disorder, the White Paper argues for the compulsory treatment of mental illness against the will of the individual.

Ideologically, the reformed act would emphasise 'compulsion at the expense of safeguards' [Grounds, 2001, page 387]. How the final act will look is open to debate. However, it continues thirty years of increasing controls and management methods on a group who typify the experiences of the socially excluded.

This overview of mental health policy and community care demonstrates several key themes. The constraints on resources, the restrictions in agency working and the negative perspectives on mental illness and mental health policy have contributed to limitations of practice [Bowler et al, 2001]. Clear deficiencies have been identified. How these shortcomings are to be addressed concerns agencies such as the police. The effect of mental health policy on the police and their work occurs on various levels. It appears that the police are still being asked to fill the 'gaps' left by the mental health system [Bowler et al 2001].

2.11 The Police and Mental Health Policy – their 'formal role'

The police operate with various legal and governmental objectives. These include consideration of police resources, both legally and materially. Equally, the wider policy

context has relevance for the authority and remit of the police within the mental health arena. In addition to the legislative framework, seemingly non-related policy has an influence and impact on the work of the police. As within other areas of police work, such as child protection, the police encounter situations that extend the remit of their every day work. There is an acceptance of police involvement in society's many dilemmas and social problems. Therefore, the policy dictating the treatment of the mentally ill effects, on occasions indirectly, the experience of the police. The police have a policy framework in which their interactions with the mentally ill are not only influenced but also determined.

The previous analysis of mental health policy and community care has identified key areas connected with its practice and functioning. The closure of psychiatric hospitals, the concept of social integration of the mentally ill within the community, the shortfall between these expectations and the policy in reality and the problems of cross organisational work have all placed limitations on community care. Agencies have experienced various complications, reorganisations and challenges in providing a reasonable level of care. Individuals experiencing mental health problems have witnessed fluctuating levels of care, leaving some isolated. For the police as a social agency, these occurrences have presented them with additional challenges.

The police have an established and accepted role in dealing with social dilemmas. In their practice, they can 'be considered as gatekeepers between the mental health and criminal justice systems' [Doyle et al, 1994, page 118]. When dealing with the mentally ill, the police call upon the various aspects of their practice and skills, including assessment, discretion, quick decision-making and the requirement for multi agency work. Their role in dealing with the mentally ill raises questions around responsibility,

appropriateness and compatibility with their overall function. They are often seen as the 'secret social service', acting as a bridge between the mentally ill and relevant agencies [Thomas, 1994]. This is a function frequently associated with the police prior to recent mental health policy.

The starting point for this association is important. As Harrison et al [1989] discovered, the mentally ill are more likely to come into contact with the police at a time of crisis. This is particularly when there is difficulty in accessing other, more relevant, services.

Many fail to achieve access to more suitable treatment regimes [Birmingham, 2001].

Stephens et al [1994] further claim that the exposed instability and florid behaviour of the mentally ill generally, with their visibility in public places, escalates the probability of police contact. It is frequently acknowledged that the number of mentally ill coming to the attention of the police is increasing [Community Care, 1997, Police Review, 1996c, Roughton, 1994]. The actual figure is hard to ascertain. There is difficulty in gaining accurate figures for section 136 specifically [Rogers and Faulkner,1987, Bean, 2001] and the more informal contact between the mentally ill and the police. This is the result of a lack of monitoring of incidents and the variations in police practice.

The police involvement with the mentally ill covers both criminal and welfare themes [Bean et al, 1991]. For,

"...where a mentally disordered person has committed an offence it is often an individual officer's discretion which is the primary determinant of whether the case becomes a crime matter or a welfare matter."

[Bean et al, 1991, page 3]

The police are being called upon to deal with psychiatric patients who are both homeless and isolated [Berkeley, 1995, Leff, 2001, Roughton, 1994, Waddington,

1993]. For many the transition from institutionalised medical care to penal incarceration via the police is quick and inevitable. The police are constantly reacting to a situation they did not initiate. How they do this, however, is largely determined by individual forces, their involvement in formal schemes and the existing philosophy of the police culture. The involvement of the police as one of several public agencies should come as no surprise.

Stephens [1994] notes this prevalence. For sections of the mentally ill, the process may involve frequent associations with the police, culminating in the common phenomena of the 'revolving door syndrome'. Bean et al [1991] state that of the cases in their study, twenty five per cent were known to the police, with thirty per cent having a criminal record. The fact that the police were acquainted with a significant minority raises some concerns regarding their use as a 'social care agency', and the type of care being experienced by this group. Further, the sustained movement of the mentally ill from one agency to another is characterised by a low level of supervision in the community and inadequate treatment.

Gardner and Pugh [1996] found that those discharged from hospital to voluntary administered homes have a higher rate of police involvement and re-admission to hospital. Hennessy [1995] comments that the police are routinely in contact with the same group of mentally ill. In interviews with the police, Hennessy [1995] was surprised at the immense quantity of time the police spent dealing with the mentally ill. This was complicated by the fact that the police recognise these numbers are a fraction of the mentally ill in the community.

2.12 The Police and Mental Health Practice

Through mental health legislation and police powers, officers can formally become involved in the care and supervision of the mentally ill. Besides the more direct method of accessing police involvement through criminal activities, the mentally ill and the police can be brought together through the use of specific sections of the two Mental Health Acts. For example, section 136, as the part of the 1983 Mental Health Act, enables the police to remove a mentally ill person to a place of safety, namely a hospital, or if necessary, a police station [this is described further on page 55].

This section typifies the dilemmas for policing. While attempting to provide care, the police are often criticised for their involvement in mental health matters [Bean, 2001]. One has to remember that regardless of the legislation and the options available, the likelihood of police contact with the mentally ill will remain. It is debatable whether the police themselves choose to be involved further with the mentally ill. A central dilemma associated with section 136, therefore, questions

" not so much whether the police should be involved but how much."

[Bean, 2001, page 93]

The problems generated centre on the potential criminalisation of the mentally ill. To involve the police in mental health policy is to confuse the boundaries of practice.

While it is not a criminal procedure, section 136 translates to added involvement with criminal justice agencies, particularly through the use of the police station as a place of safety. The demarcation of powers and responses is confused.

Another factor evident in the use of section 136 is the identification of mental illness, which provides the basis for decisions on sectioning. The police are good identifiers of mental illness [Bean et al, 1991, 2001]. They predominantly rely on obvious

disturbances of behaviour as indicators, taking a 'pragmatic' approach by assessing the individual's appearance and speech pattern. Figures provided by Rogers and Faulkner [1987] demonstrate a high proportion of successful diagnosis on the part of the police, with officers achieving a rate of ninety per cent diagnostic accuracy. The reasons why some police forces have an ability to assess mental health successfully has more to do with the frequency of association, rather than any specific training initiatives. This is one argument that is relevant for the Metropolitan police force, where in 1995 1,500 people were detained via section 136 [Cherrett, 1996a, 1996b].

The police encounters with the mentally ill consequently involve encounters with other services and agencies. In this guise, the police become co-ordinators of actions and procedures. However, the path to an effective multi agency response is difficult to obtain. This lack of agency organisation has been a frequent criticism of police practice [Bean, 2001, Ritchie et al, 1994]. Roughton [1994] cites one example of the problems in communicating, co-ordinating and implementing procedures. In the detentions identified in the West Midlands area, an approved social worker presence was achieved in just over fifty eight per cent of cases. Accomplishing both medical and social services attendance occurred in only fifty one per cent of detentions. Berkeley [1995] and Rogers and Faulkner [1987] similarly found that approved social workers often failed to attend section 136 assessments. The reasons for this lack of social work involvement may be the result of social services being oblivious to their assessment role. The police may not be aware of the importance of social work input. Alternatively, the ethical, philosophical and culture backgrounds of the two agencies may suggest resistance and aversion to the implementation of inter agency co-operation and working. Regardless of the reasoning, the outcome remains the same; an incomplete assessment and procedure.

In response to this complicated scenario, several initiatives have emerged. These range from the alliance of the police and professionals from health and social services in the provision of assessment and diversion in police stations to joint procedures on section 136 [Carty, 1996, Rickford, 1995a]. To highlight the situation of the mentally ill and for those involved in the criminal justice system, the Reed Committee [Department of Health and the Home Office, 1992] examined the services utilised and provided recommendations for the care and treatment of this group, including the concept of diversion. The report made clear the necessary components of an effective approach to the care and treatment of the mentally ill in the criminal justice system. It echoed the objectives laid down in the Home Office circulars and the prerequisite that the prosecution of the mentally ill should occur only when necessary. One conclusion reinforced the notion that the services for the mentally ill need to be not only better coordinated, but also effectively used.

2.13 The Development of Diversion

The development of more formal schemes was an attempt to provide a division between care and control for the police in their work with the mentally ill. As Stephens et al [1994] note, the operation of such projects illustrate how the police can then concentrate on their 'real' role and practice. The police operate as 'case finders', facilitating the inclusion of other professions into the care and management of the mentally ill. Using a named police officer as co-ordinator, the police liaise with professionals in a manner that benefits all parties, including the mentally ill. In Britain this is signified by the formation of mental health liaison officers in the Metropolitan police. The designated officer negotiates with other agencies on issues relating to the treatment of the mentally ill and ensures the police are represented in discussions. There has been a call for the introduction of mental health liaison officers in all forces to increase the level of

interaction between the police, other agencies and the mentally ill [Bean, 2001, Police Review, 1996c]. The success of these schemes relies, however, on joint discussion and planning at a senior level.

In making these schemes the official policy response, the principle dictating juvenile justice was integrated, which recognised the dichotomy between welfare and control [Pitt, 1990]. In the 1990s, this translated into the policy of diverting the mentally ill. In one sense it has been an assertive objective. As Bean [2001] argues 'diversion has acquired something of a cult position' [page 110]. It is not a new policy. Birmingham [2001] sees the history of diverting the mentally ill as going back two hundred years. What has occurred over the last three decades is the recognition of alterative facilities for those deemed mentally ill and in contact with the criminal justice system.

The first Home Office circular 66/90 outlined an approach for local authorities in managing the mentally disordered offender. The emphasis was the removal of those who had either committed or where suspected of committing an offence from the criminal justice system and placed within the care of social services or the health sector. It was clear that diversion should occur on all levels; at the police station, in court and in prison. This basic principle realised the inappropriateness of managing a group with diverse social and medical needs within a system geared towards the establishment of guilt, the administration of justice and the legal sanctioning of punishment. It intended to take the 'psychiatric' out of the criminal justice system.

With regard to police involvement, the circular and accompanying documents outlined their existing responsibility with section 136 and the opportunities for accessing relevant services. It further proposed to define the aims of diversion and its application. It was anticipated that consensus on the most appropriate approach regarding the

mentally ill would also emerge. In reality the development of strategies and the implementation of specific schemes was not as widespread as anticipated. A survey carried out by Bingham in 1992 of 43 police forces found only 12 schemes in operation, with an additional 13 schemes planned in the remaining areas [in Staite et al 1994]. Conversely, Vaughan et al [1995] note the official figures from the Home Office that acknowledge the presence of sixty schemes. Birmingham [2001] further confuses the situation by stating that there now exist 40 diversion schemes. What can be seen from this development is an unclear model of diversion and unqualified reports of its achievements.

What is essential in understanding the promotion of diversion is the reliance on the two systems effectively working together. An additional circular (12/95) and the associated document, 'Mentally Disordered Offenders: Interagency Working' [HMSO, 1995] reinforced the inclusion of additional methods of management. It predominantly reiterated the notion of inter agency working. While illustrating the existing schemes, the circular attempted to act as a catalyst for encouraging further initiatives. Practical issues also emerged from the circular, such as the training needs of staff and the systems for transferring information between services. Other factors, however, affect the development and success of formalised schemes. The introduction of such schemes requires the confidence of the police in order to achieve its aims. The police have to perceive the scheme as credible. From their viewpoint, they have to know that the project can provide a quick and succinct response to their call for assistance. As Meehan [1995] has shown, the lack of involvement of the police, and their reluctance to utilise such facilities, can determine the outcome of any intervention.

Whether diversion has been successful in meeting its aims is debatable. There is confusion over the procedure for formal cautioning, particularly establishing guilt and the appropriateness of the response. Equally, as with juvenile justice, the role of the criminal justice agencies as gatekeepers to both systems remains contentious. Further, what defines effective practice is difficult to ascertain. Bean [2001] remarks that the objectives placed by individual agencies influence how diversion is perceived. Whether the policy has reduced the numbers in contact with the criminal justice system or promoted agency cooperation is at the centre. As Bean [2001] acknowledges, the complications inherent in the schemes indicate that, while the numbers progressing through the system have decreased, the ideals of agency working are limited. The professional ideologies and attitudes continue to dominate and ensure that effective communication is problematic. Similarly, the future of diversion is open to question. Recent changes in the NHS, in resources dedicated to diversion and in prison health care, all impact on the continuation of diversion [Birmingham, 2001]. A potential concern remains that

"many schemes, which find themselves isolated from mainstream psychiatric services, fail to achieve what they set up to do."

[Birmingham, 2001, page 204]

2.14 Further Developments in Practice

Regardless of the wider political and social context of practice, the police service is still required to incorporate a myriad of objectives, including the establishment of assessment arrangements, the development of a police co-ordinator of policy and the implementation of various schemes and training forums. Surrounding this is the balancing of public concern versus the social care of individuals. These changes recognise areas of practice that result from previous governmental objectives and

encourage the many components of the two systems of health and criminal justice to establish further methods of dealing and managing the mentally ill. What appears to be important throughout this, and reinforced by other policy objectives, is the interconnecting nature and mutual dependence of the police and other agencies.

Similarly, police involvement in the Criminal Justice and Court Act 2000 highlights the importance of seeing the whole system and the dependence and connection between each agency. The police are at the beginning of the continuum, and while providing an obvious access point to both the criminal justice and social care systems, cannot be seen nor act in isolation. The policy and legislative framework of the police illustrates the multi layered context and social arena in which they operate. One further implication for the police, and significantly their interactions with the mentally ill, can be seen in the Crime and Disorder Act 1998. The Act deals with several criminal justice issues, including juvenile offending and community safety. Essentially, it stresses the development of community partnerships. While appearing to echo and sustain the practice of community and problem orientated policing, the Act places a duty on the police and the local authority to instigate and lead schemes aimed at dealing with localised problems of crime and disorder.

The Act, in operation since July 1998, considers a variety of elements, including community forums and the inclusion of services such as the police, social services and education in the formulation of community approaches. It was envisaged that responsibilities would be placed on the police to take the initiative within these partnerships and to develop answers to the problems identified, regardless of those involved in the process. As Newman [1998] states,

"The inspired, can do, practical experience based approach does provide an early resolution to many of the incidents which police officers attend....."

[Newman, 1998, page 23]

Jenkins [1999a] provides some indications as to the difficulties in designing and implementing the community safety approach, highlighting the issue of participation and partnership. One could surmise that, within the remit of these forums, issues pertinent to the mentally ill in that community would be addressed. The Crime and Disorder Act 1998 and the community policing stance could deal with the wider issues relevant within communities, including the mentally ill. It may also fundamentally alter the response provided by the police.

These developments have to be seen in light of the various public demonstrations of anxiety towards the mentally ill. As community care was developing and governing the treatment of the mentally ill, events impacted on police actions. With the publication of the 'Clunis Report' [Ritchie et al, 1994] there was a growing awareness of the interface between the criminal justice system and other social agencies in managing the mentally ill in the community. This significant report highlighted the failings of the system as a whole in dealing with the mentally ill who pose a significant risk to themselves and/or the public. For the Metropolitan police particularly, their involvement in the care and treatment of Christopher Clunis highlighted the deficiencies in inter agency working. While the police were not as severely criticised as others, they devised procedures to prepare their officers to deal with the mentally ill. The impact on the police may be seen as two fold: firstly, the development of organisational systems, such as greater inter agency co-ordination manifesting itself in processes aimed specifically at the mentally ill, and secondly, the individualised response of officers.

2.15 Care and Control Dilemmas

Central to this discussion of the police role are the two potentially contrasting aspects of care and control [Stephens and Becker, 1994]. Prominence has been placed on the controlling dimension at the expense of a thorough analysis of the care that is involved in police actions. The occupational culture of the police and its emphasis on the police identity highlights an insufficient acknowledgement of care as an attribute of policing. Yet,

"despite the subculture..of the lower ranks that still stresses 'real' police work in control terms, there is undeniably a significant care aspect in the work of the police."

[Stephens et al, 1994, page 214-215]

What becomes clear from the literature is the lack of a coherent distinction between care and control. Rather than there being a dichotomy between the two concepts, there is a spectrum of police activities that includes both elements. The police are seen as agents of both care and control. The principle model for this combination is the implementation of section 136. The blurred boundary between care and control continues to illustrate the interactional environment and potential dilemmas that the police encounter in their dealings with the mentally ill.

The experiences of the police in dealing with this particular group are multi faceted and require reflection on possible solutions and approaches. However, these formal responses are primarily concerned with those the police originally perceive as having committed an offence, even a minor infringement. There is a fine line between being formally recognised as a 'mentally ill offender', being dealt with through the various methods outlined and being seen as in need of treatment and not the responsibility of the criminal justice agencies.

As Staite et al [1994] acknowledge, the establishment of a link between mental illness and criminality is fraught with difficulties. Questions around the influence of the illness on their actions and their ability to understand their behaviour needs to be established, and in the short term may often remain unresolved. It is this dilemma the police face. In order to deal with the situation officers use such generalised categories as 'breaching the peace' as a method of negotiating their situation and resolving the plight of the mentally ill [Bowler et al, 2001]. It is the formal recognition of mental illness that allows for the next stage of the process. It is here that the organisational response is activated, which may include following inter agency formal procedures for section 136 or the use of diversion schemes. As with the response to youth crime, the familiar scenario of variation and deviation emerges with the experiences of the mentally ill and the police being dependent on locality, timing and resources.

While the measures such as diversion at the point of arrest are commendable, they do not address the causes behind these increases, the suitability of police involvement and the perceptions of individual officers. Nor does it take into account the everyday encounters of the police with the mentally ill, those that do not necessarily result in an arrest or section, but do contribute to the police extending their 'welfare' role. As Evans [1974, in Bean et al 1991] argues, without this role the wider community would lose "one of its major social services" [page 2]. Many of the mentally ill would be deprived of a gateway to other services. The recent concern surrounds the reliance of others on the police to fulfil this gatekeeper role. This, in turn, influences the police role and police priorities.

As Bean et al [1991] note, police involvement with the mentally ill is not new. The growing concern today is the frequency of this involvement, the level at which it is occurring, and the manner in which the mentally ill are dealt with. From a police perspective an additional area of concern must be the serious type of mental illness they are required to assess and deal with [Bean, 2001]. As the shortfalls of community care and mental health policy become apparent, the seriously mentally ill are now increasingly likely to be incorporated into the community.

The police, when dealing with the mentally ill, are required to examine a variety of issues which compounds an individual's mental illness. These issues concern not only their practice as individual officers, but also the wider organisational issues. How the police subsequently resolve these dilemmas, in the face of demands on their time, their expertise and their resources, is dependent on their perception of the identified 'problems'. Equally, there is a responsibility on the police organisation to provide an answer to the demands made by this particular social problem.

2.16 The Police Role

While the formal approach provides one context, the police generally use their existing powers and resources to manage the mentally ill in the community. Questions concerning the function of the police remain, regardless of the proliferation of formal procedures and schemes. The main question centres on the following; what is the role of the police and how do they manage the mentally ill.

The police as a social institution possess certain attributes and generate assumptions surrounding both their function and practice. These have changed in response to social, political and historical developments. The combination of these factors has led to an often-changing image of the police service. The police service fulfils several functions

in society. These responsibilities cover a range of social aspects, including protection, welfare and community support. There is a mismatch between the public desire and acknowledgement for the police to be visible and 'on the beat' and the need for increased efficiency and cost effectiveness in police performance [Hick, 1997a]. The dichotomy between these two elements means that the police face difficulties in prioritising and fulfilling their various roles and tasks.

The police are no longer certain about the role they fulfil [Hicks 1997b]. The response of the police to this uncertainty has been to implement organisational changes, such as the emphasis on crime prevention and the implementation of organisational systems at regional and national level [Morgan and Newman, 1998]. While they have displayed increased levels of professionalism, the main argument surrounding the central role of the police in society has not been resolved. The apparent lack of common morals or values in post-modern society has placed the police in a difficult position. The police remain an integral part of the construction of social behaviour and norms.

By recognising certain behaviours and actions as unlawful, they reaffirm these acts as deviant and illegal, thereby maintaining the acceptable standard of social order. As McLaughlin and Muncie [1996] argue, the police have considerable power and influence, effecting the content of their work and determining their own development. The changing focus of practice has ensured that the police continue to face a variety of demands on their skills, resources and time. This reliance has contributed to the apparent defusing and complex nature of police work and the difficulty in defining the boundaries of the police roles and tasks.

2.17 The Key Roles of the Police

With the changing objectives of the police and the emphasis on efficiency and effectiveness, it may prove difficult to assess the diverse roles of the police and the functions they currently fulfil in society. For officers in particular, the dual expectations of order maintenance and law enforcement incorporate possible conflicting responsibilities and approaches [Muncie and McLaughlin, 1996].

De Lint [2003] questions the role of modern policing. In doing so, he recognises the core perspectives on police work that have developed over the past forty years. To summarise the main findings, it is suggested that the police are seen in the following guises:

- As 'coercive authorities' [De Lint, 2003, page 380], where the predominant issue is the application of force in the maintenance of order. Here the police are defined by their function. Encompassed within this perspective is the recognition that police authority, morality and working personality all contribute to the 'framing' of coercive action. This equally recognises certain attributes as core, such as self-reliance, individual interpretation and the reactive skills of the officer. A critique of the approach is its limitation in defining the police role. The police do more than merely use force.
- As 'knowledge workers' [De Lint, 2003, page 383], where the complexity of the situation is taken into account. The essential elements within this description include the information processes, which allows the police to detect and present cases. Via the communication systems and schedules of the organisation, the officer acquires information and subsequently knowledge that is used as a 'means to an end' in achieving the police goals. This is within the context of a society dominated by risk. As De Lint [2003] argues, questions need to be raised including, 'for whom' this role is fulfilled. The police here 'penetrate the viability of targeted populations, gather certain kinds of information, and use it in the aid of political authority' [De Lint, 2003, page 386].

De lint [2003] argues that these two definitions or descriptions need to be re-examined and offers a further analysis, which will be discussed later.

The debate continues as to the role or roles of the police in modern society. Waddington [1993] provides one interpretation of police work. He identifies, in particular, the 'order maintenance role'. It is a role that the public feel most comfortable with. Order maintenance, or 'peace keeping', however, requires different interpretative levels. This involves not only an interpretation of the law, but also the labelling and assignment of blame and a determination of the appropriate level of conduct. This leads to the issue of the initiation of police action; whether it is 'citizen invoked' or whether the police have initiated the action themselves. As Waddington [1993] states, these distinctions effectively differentiate the law abiding from the lawbreakers. It is an area by which the police are judged and one that is difficult to evaluate. It also is an area where the police have the least control.

Certain issues transpire with regard to the order maintenance role, most notably the extent and type of order imposed. The usual perception of the police as controlling crime is unrealistic. Policing methods and presence do not necessarily influence crime levels. The factors behind an criminal act are various. While the police develop certain methods [such as the increased use of technology], the control of extralegal factors raises issues concerning the extent of police influence. Possible variables such as social deprivation, unemployment, the moral and social climate and the standing of social norms may all contribute to the enactment of crime. The police, as one of numerous social agencies, are equally effected by these demographic and cultural factors.

Key roles emerge when detailing the actual work and practice of the police. The police fulfil seemingly contradictory roles; as order manager, as law enforcer and as public servant. Each carries specific connotations, such as the enforcement of the law, the

maintenance of social order and the provision of, a 'humanitarian and community service' [Pugh, 1986, page 2]. There are connecting features in these roles. For,

"In fulfilling this public service role, law enforcement itself is seldom used but its potential use is always present. The police officer's special access to law enforcement gives him a large advantage over other professionals....who may also attempt this public servant role."

[Pugh, 1986, page 3]

One problem that emerges from these multiple roles can be seen in the expectations of individual officers. The various internal personas provide dilemmas and conflict. The police require mechanisms in order to identify and form solutions, both internally and externally. What is of specific relevance for the police officer is the notion of the 'internalised self', which manifests itself in the tendency for the individual to include part of themselves within the particular role they fulfil, incorporating a definitive manner or style [Pugh, 1986]. The outcomes for the police involve the concept of policing which interplays with their role and in the performance of tasks. Their choices and priorities are dependent on their perception and perspective of policing.

In situations which emphasise the law enforcement aspects with its clear defined boundaries, the notion of policing and police work is straightforward. When situations arise that require numerous roles, conflict emerges [such as domestic violence and mental illness]. In cases which appear simple and defined, the police officer has a clear concept of the action to take [Pugh, 1986]. In more ambiguous occurrences, additional factors are prevalent. These include the ideology and principles of the police officer, his or her concept of policing and the details of each individual case.

The notion of the police fulfilling a crime-fighting role is pertinent to any assessment of the image of the police. In reality statistics suggest that the police spend a relatively limited amount of time on crime related issues. Figures from the Association of Chief Police Officers claim that the numbers of calls related to crime matters amount to eighteen per cent of the work of the police [in Muncie et al, 1996, page 53]. Subsequently, this type of work takes a disproportionate amount of time to complete, approximately thirty per cent of the officer's time. Thomas [1994] additionally notes this reality of policing, reiterating that up to seventy per cent of police work is concentrated on non-crime matters. Police practice today suggests that, regardless of the their inclination to be mainly crime fighters, the majority of their work centres on the wide ranging service role.

It is in this capacity that the police can be seen as working with the mentally ill.

However, it is this service function which is often viewed negatively by the police themselves. As Waddington [1993] comments,

"The problem lies in the police attitude to such demands for assistance. Officers - most of whom regard their role as that of 'crime fighters' - often regard such calls as 'rubbish' on which they are reluctant to spend time and effort."

[Waddington, 1993, page 6]

Attached to this service dimension is the realisation of the 'caring' aspect of policing [Becker and Stephens, 1994]. The distinction between the service ideology and the more authoritative presentation of the police is apparent within their dealings with the mentally ill. It is difficult to distinguish between the caring actions of the police towards the mentally ill and the more controlling aspects established within the police role, which call for the maintenance of order and stability. The apparent contradiction in roles further hinders the police in finding a clear definition of their practice.

What De Lint [2003] suggests, in light of the various proposals, is that we see the police as 'access brokers'. This argument recognises the power and conflict involved in police

action. The police here obtain and grant access to the few that they identify as 'troublesome'. De Lint [2003] defines this as the ability to

"...deploy the special instruments of law, force and information to broker access to troublesome individuals and problem populations for an administrative authority."

[De Lint, 2003, page 380]

The major contention for De Lint is to clarify 'for whom' the police operate, particularly in democratic authorities. Whilst in essence the police act for political stability, this term should recognise that the 'political' involves not only the wider state apparatus, but also commercial interests and local communities. There are several key themes in fulfilling this role, placed along a range of actions and preoccupations:

- The 'immediate objective' or the acquiring or restricting of access to people and places. This is a key role for the police.
- The 'special instrument' the use of powers, force and the technological advances available to the police.
- The 'intermediate justification' the wider notions of law enforcement, order maintenance, crime prevention and knowledge.
- The 'ultimate objective' of the police, which is to 'maintain the political, economic, institutional and/or moral authority' [De Lint, 2003, page 389]

What appears here is an attempt to encompass the aspects of the police role that other commentators have isolated, such as the use of force, law enforcement, crime prevention and the maintenance of the status quo. De Lint [2003] has argued for a definition of the police role that incorporates the changes that have occurred in policing, recognising the various levels at which the police operate.

Overall, the tasks of the police are 'means orientated' rather than 'goal orientated' [Waddington, 1993, page 175]. Both the level of resources and the consent of the public influence the role of the police. The police, to some extent, are acting with the

permission of wider society. Further examination of this seems to point to the police fulfilling tasks and roles because 'that is what they have always done' [Waddington, 1993, page 175]. The roles of the police are, therefore, partly determined by their history, expectations of their actions and the justification of their presence. Whether they are maintaining the social order or fighting crime, behind their role is the dependence on a working culture to help the police define what they do and why they do it.

2.18 The Canteen Culture of the Police

Within any examination of the police force, consideration has to be given to their culture. Any examination of the police culture attempts to explain those who enter the police and the differentiation between officers and those they serve. The culture provides the police with protection, confirming their shared beliefs and norms. It assists officers in their own interpretation of appropriate action. By incorporating various aspects, the culture helps to legitimise the police and the actions they take.

Within the police culture there are defining characteristics. As Horn et al [1997] and Reiner [1992b] note, the style and predominance of the police culture may alter, depending on the police force in particular and individual units within these forces. However, there are shared features common to all forces, which are not dependent on regional or operational features. Among these general aspects, there occurs an adherence to unity, recognition of rank, and an attitude that distinguishes the police from groups outside the organisation. This reinforces their 'outsider' persona.

Their 'belief system' sees police work as a challenge, with uncertainty in the situations faced and the possible dangers encountered. Reiner [1992b] sees a link between this perception and the police view of their 'indispensability'. The police consider the

consequences to law and order if their presence was no longer maintained. This demonstrates the importance to the police of maintaining the status quo. The police also develop a sense of pessimism. Officers often become cynical when they consider the moral ineptitude that characterises certain sections of society [Reiner, 1992b]. Equally, while officers believe in action and danger, in reality police work deals with more basic and ordinary circumstances [Reiner, 1992b].

Central to this is the idea of what constitutes a 'police officer'. Pugh [1986] conceptualises the notion of the 'good officer' and assimilates the positive qualities and behaviours. The development of a 'good cop' within the police culture is dependent on the existence of several distinct, but interrelated qualities; a moral awareness, a recognition of the circumstances behind action and the development of mechanisms in order to achieve a resolution. Four types of officers can be identified and although 'good' or 'bad' characteristics are not attributed to their presentations, they do indicate distinctions between individuals and roles [Reiner, 1992b]. Included within these categories are the 'bobby' or ordinary officer in a mainly peace keeping capacity, the 'uniform carrier', who maintains a more pessimistic attitude, the 'new centurion' as crime orientated officer and the 'professional', pursuing policing as a career. While common themes emerge, it may be that a significant amount of organisational style will dictate the role incorporated by the majority of its officers. A combination of these themes and presumptions influences action.

There are difficulties in isolating the notion of the 'good police officer' and distinguishing it from the overall direction, role and concept of policing [Pugh,1986]. The roles and characteristics identified recognise the skills and abilities required in the fulfilment of police work. They involve the ability to react quickly, to initiate

judgements, formulate appropriate and inventive responses to numerous events, manage stressful situations and essentially exhibit a 'common sense' approach. Surrounding these ideals is the competence of that individual. This manifests in the ability to 'do the job', whilst displaying common sense. The strength of the notion of 'common sense' is a determining factor in distinguishing the 'good officer' in the culture of the police. However, it can be argued that to distinguish the characteristics and abilities required to perform the job from the role itself is difficult. In some respects the attributes determine the extent and nature of the role.

2.19 'Danger' in the Police Culture

Within the police culture there is a wider recognition of the dangers and violence involved in police work [Muncie and McLaughlin, 1996]. It is an essential part of that culture. Danger arises out of the unknown and the sense of unpredictability, out of the differing meaning of authority and power. This again contrasts with the reality, where the occurrence of violence may not reach the levels assumed [Bayley et al, 1984]. However, it is its central position in the ethos of the police that enables officers to distinguish themselves and their work from the rest of society. It defines the police and their response to tasks [Skolnick, 1975]. It is natural for officers to place violence and danger as central features of their work. This notion of danger is so encompassed within the police role that it forms both an overt and covert framework in which to judge social interactions and the other participants in those encounters. In association with criminality, past violence and assumed further violence are key factors in determining action by the police [Klinger, 1996b].

Additionally enshrined in their working persona is the need for predictable behaviour and outcome. The preference for predictability in the daily experience of the officer is

obvious. It acts as a guide to the methods of management and action. The necessity of regularity and predictability includes the types and characteristics of those they encounter. Similarly, the most overriding fear components include beliefs around actual violence and its potential. The predominant concerns mirror those indicated by Alder [1996], such as the 'unknown' and the unmanageable aspects of their involvement with the mentally ill.

The police concept of force is linked to the presumption of danger [Hunt, 1985]. The determination of force involves a process of allocating notions of 'normality' to every day encounters. Normal force is indicated within a continuum of definitions, comprising legality, its use and the acceptable employment of force. The centrality of force, actual and as a symbolic tool, to the police role and authority underlies their behaviour and actions towards individuals. Force is often used as a measure to re-instigate order, and ultimately control, when authority is questioned. Force can also be explained in the more abstract sense, with the belief in moral reasoning [Hunt, 1985]. This justifies its use on the grounds of morality, with its enactment occurring against those the police consider in need of punishment.

2.20 Police Culture and Masculinity

The cultural background of the police is shrouded in connecting themes and beliefs. What is also evident is the presence of the 'cult of masculinity' [Horn and Hollin, 1997]. The emphasis on shared aims additionally stresses the importance of authority and courage [Skolnick, 1968]. These features not only are the basis for the police 'in group', but exclude other members within the service, such as women, who are not seen as possessing these attributes. Underpinning these particular components of culture is

the military like structure of the police, which reinforces the numerous attitudes [Walker, in Stephens et al, 1994].

There are issues around gender and the police force. The structure identified by Walker [in Stephens et al, 1994] reinforces the notions of masculinity, with an emphasis on perceived male characteristics such as courage, force and authority. These factors affect not only female officers, but also women seeking help from the police. There has been a proactive approach to raise gender issues within police forces, with specific programmes aimed at increasing the career progression of female officers, introducing women forums within police forces [e.g. Bedfordshire Police] and increasing female representatives throughout the various divisions and across all ranks. This general 'Gender Agenda' has sought to facilitate improvements in the working environment of the police service and enhance the inclusion of female police officers at every level.

For those women aiming to contact the police for help and assistance, some key features are highlighted [Women's National Commission, Thames Valley Police, Todd Consulting and University of Surrey, 2002]. It has to be recognised that women are often involved with the police as victims of crime, rather than as suspects or witnesses. The issues highlighted for women involve perceptions and practical considerations. From the perspective of women, they often felt that the police often viewed them as 'low status', feeling that they were patronised. Accessing the police was seen as difficult. More 'specialised' areas, such as domestic violence and sexual violence were characterised by lack of resources. Overall, the women within this particular study highlighted common features of women's experiences of police forces generally; marginalized and not listened to [Women's National Commission, Thames Valley Police, Todd Consulting and University of Surrey, 2002].

What is apparent form this report is that certain key areas of police work need to be improved to ensure that women receive an effective service. One comment to come out of the report is the need to change the 'police culture'. Suggestions to facilitate this change include gender and cultural awareness training and the inclusion of more senior women officers throughout police forces. The 'Gender Agenda' recognises elements of these issues.

When looking at the culture of the police there is also an emphasis on experience. This aspect of culture translates into a method of informal instruction. Culture operates as a guide on how to act and respond. It provides an introduction into the peer group and the informal hierarchy. The newer police officer is told of the various folklores and traditions of their particular force or unit. It is part of the socialisation and initiation of the officer, contributing to the assimilation of culture [Muncie and McLaughlin, 1996]. This, in turn, develops into a police sensibility on how to be an officer [Shearing et al, 1991]. The more experienced will provide a 'common sense discourse on crime' [Muncie and McLaughlin, 1996]. The moralising on crime, detailing the purpose of the police, aids the officer in distinguishing between 'real' police work and the more mundane aspects that do not necessarily result in action being taken.

The police culture is by no means a unique occurrence. Other established professions have identifiable traits and notions that combine to produce their own culture. For these professions a collective sense of identity permits the more difficult and demanding aspects of their work to be acknowledged by their peer group. It provides support and a sense of belonging. There are advantages in the existence of a police professional culture. However, the dominance of a culture may lead to conflicts and difficulties, especially for the management of organisations and the enactment of powers.

2.21 Police Authority, Discretion and Action

While discussion has centred on the presence of the police culture and the expectations of officers, we need to return to the question of what the police actually do. This concentrates on the power of the police and their methods of working. The police culture forms an important and vital part of the overall practice of the police. Two issues, however, remain; the powers of the police and their reliance on discretion as a fundamental tool.

2.22 The Powers of the Police and the Basis for Discretion

The powers of the police centre on the detection of criminal offences, the establishment and maintenance of peace and the prevention, where possible, of offences being committed. This is enshrined in the oath taken by new officers on joining the police [in Fitzgerald and Muncie, 1983]. The police have the power to stop and search people, to investigate alleged crimes, to arrest identified individuals and to interrogate suspects.

The difficulty in deciphering the police powers arises from the diverse circumstances surrounding their foundation. The definition of these powers is unclear. One point to remember, however, is that there is no legal duty on the police to carry out any of the highlighted duties or powers;

"..if a house is burgled, the householder has no right to demand the police investigate.. A person who is attacked has no right to demand the police prosecute the attacker. The police have no specific legal duty or responsibility to help individual citizens."

[Fitzgerald and Muncie, 1983, page 46]

In basic terms, therefore, the police are performing a service for the community by implementing their powers.

The powers and functions of the police are not always found in legislation [Feldman, 1993]. It is difficult to formulate a complete list of the powers of the police and to provide a complete outline of police activity; each force can dictate how they operate, and the issue of discretion or the 'doctrine of constabulary independence' predominates, with individual officers interpreting the law in their own manner [McConville et al, 1991, page 2]. This basis is often found not in legal powers, but within the informal practice of the police. The suspicions, knowledge and experience of each operational officer determines the execution of these powers. While senior officers have responsibilities to supervise the lower ranks, police practice is carried out by the middle and lower ranks, away from the sight and control of senior police officers [McConville et al, 1991].

The ability to decide and determine own styles of working, responding to situations as they occur and managing incidents in a manner that coincides with professional and personal expectations, has led to the deployment of discretion as a tool. The use of discretion within the work environment is in answer to the numerous and varied circumstances that arise in police scenarios. The police are frequently the subjects of analysis regarding the use of discretion [Bittner, 1967a, 1967b, Goldstein, H, 1963, Goldstein, J, 1960, LaFave, 1962, Reed, 1980, Reiss, Jr, 1984, Rumbart and Bittner, 1977, Waddington, 1999]. The definition of discretion implies that the officer exhibits choice in actions and behaviour. This relies on their interpretation of events and an anticipation of possible outcomes. Reed [1980] provides a clear definition of discretion when he states that

"Discretion is the freedom to do what one chooses to do and to use one's personal judgement in reaching decisions."

[Reed, 1980, page 54]

Choice within police work plays an important role [Shearing et al, 1991]. The extent of police discretion is not necessarily a known feature and its use may be over estimated [Bayley et al, 1984]. In order to achieve their aims, the predominant perception of their environment, how they interact with it and the exchange between the various parties involved becomes both relevant and important. While the police are not the only agency to establish informal methods and practices, this is an essential part of their work. There is also an overriding assumption that,

"Police activity always takes place in a context of action that could have been, but was not taken."

[Shearing and Ericson, 1991, page 487]

The vital feature that an officer appreciates is the ability to quickly understand a situation and to subsequently follow the course of action which they perceive as the most appropriate. Shearing and Ericson term this 'the craft of policing' [1991, page 487]. However, this does not necessarily involve the conscious use of rules. It is experience that assists in the interpretation of events and the process of decisions. The complex world of the police demands skills that protect not only the officer and their colleagues, but also their community. Other issues influence the actions of the police [Chan, 1996]. By using Bourdieu's concepts of the field and habitus, notions of rules and beliefs [Shearing et al, 1991] and Sackman's notion of cultural knowledge in organisations, Chan [1996] provides a framework for police action, their culture and the possible influences on police discretion.

There are certain elements of Chan's [1996] study that illustrate the background in which the police operate, notably in three particular areas. Each differs in the emphasis taken, with

- culture being seen as shared knowledge within an organisation, which provides a semblance of reality for those involved. The emphasis is on the 'group' functioning in place of individual expectations and behaviours. This is further defined as incorporating four elements; 'dictionary knowledge' where the aim is to provide definitions of situations, 'directory knowledge' providing descriptions of how practice should occur, 'recipe knowledge', which details what should and should not be done in certain circumstances and 'axiomatic knowledge', which maintains the assumption that this is the way things are done in the organisation [Sackmann].
- the notion of construction, where police officers are actively involved in the formulation of practice, rules and beliefs. Officers develop a knowledge which informs them of which actions are appropriate in differing situations [Shearing and Ericson, 1991]
- the relationship between the concepts of field and habitus influencing the police, and their behaviour [Bourdieu]. When referring to habitus in this context, Chan [1996] appears to recognise the internal structure of decision making of the officers, their consideration of options and alternatives. It is a process that is 'ingrained' in individual officers. It serves as a frame of reference for action, with the social field being the wider political, economic and social setting of action.

There is a similarity between Bourdieu's concept of the habitus and Sackman's development of cultural knowledge. There is a connection between these two concepts, with Sackman's idea of cultural knowledge providing more detail to the notion of the habitus. Discretion can be seen in light of the various approaches to understanding police action.

This argument recognises that the informality of discretion and its culture should be used in conjunction with the more formal rules. The relationship between the two guides police practice in reality. While Chan [1996] notes the role of managerial strategy and implementation, it is this area of informal processes that has consequences for the use of police discretion. The arena they operate in on a day-to-day basis and the sensibility they use are the main responsibilities for the officers involved.

However, various issues still remain concerning the overall purpose of discretion. In police practice, looking for the unusual becomes a determinant factor of police action [Hester and Elgin, 1992]. It has been acknowledged that the police environment is often unpredictable, undefined and complex. Several factors permit the practice of discretion. The myriad of influences, personal and professional, situational and sociological, legal and organisational contribute to the existence of discretion within the every day practices of the police. Equally, various external factors such as the demeanour of individuals, the context of the police action and the location of incidents influence police behaviour. [Smith and Visher, 1981].

2.23 Discretion in Practice

Discretion can be seen as pervading the vast area of police work. For example, in the process of identifying suspects the use of discretion is particularly evident. It involves a process of singling out potential suspects by means of compiling a list of characteristics that serve to confirm the police officer's view of suspicious behaviour. The theoretical basis of defining a suspect population incorporates those who are criminal and those whose behaviour is an indication of criminal intent. Such distinctions are not straightforward.

In reality, the process is complex. There are key indicators in the process of suspect identification that not only involves the self-identity of 'criminals', but also the interpretation of the police and wider society. Classification and categorisation singles out the suspects and determines the future course of action. This explains why some are released without further action, others cautioned and a significant number are formally charged. This arises not from any predominant legally defined rules, principles or powers, but from the police perspective of actions and behaviour.

When examining the powers of the police, an important distinction emerges. The police do not necessarily have to arrest an individual, but they may do so [McConville et al, 1991] Additionally, they do not have to stop and search people in the community, but they can if they choose to. It is this distinction between having these powers and using them which is at the core of police work. The decision-making processes of the police explains why and when they use their powers. Rather than be aware of the legal powers that are in existence prior to an incident, and subsequently assess the appropriateness of these powers, the officers behave in such a manner that amplifies their discretionary capacities. The argument can be taken further; the dilemma for the police is not predominantly who to arrest but what for. It is in recognising the appropriate powers and legal rules that appears to provide the difficulty, as opposed to the identification of suspects.

The police operate 'on the street' in their particular community. In line with this premise, the group that has a tendency to inhabit this environment more frequently will have contact with the police on a regular basis, such as young working class males [McConville et al, 1991]. Therefore, the suspect population comprises of members that are more easily visible to the police. This group, however, are members of the section of society which also has the most frequent contact with the police as victims of crime. There is a correlation between being a suspect or offender, and being a victim of criminal acts. For, as Waddington [1993] states,

"..the very people for whom the police have least regard [are those] who need police services most."

[Waddington, 1993, page 9]

The police also determine the normality of an area so that an officer can tell when an unusual event is occurring. Additionally, using suspicion as a basis, officers adopt the

'incongruity procedure', where the occurrence of 'out of place' incidents assists in their decision-making and practice [Sacks, in Hester et al, 1992]. It becomes an integral part of the practice of the police officer and frequently develops into what is termed 'instinct'. These areas are important. They aid in determining the client group and ultimately the functioning of the police force generally. Without a suspect population, regardless of the individual characteristics and demographic factors, the role of the police would be limited or non-existent. This does not diminish the actuality of crime, however, but acknowledges that their selection of suspects is socially and politically constructed. How the police use their discretion is important.

In simplifying the process, two significant factors that determine the use of discretion are identified. Primarily, one concern involves the nature of police involvement itself. Whether it is a law enforcement or order maintenance matter influences the deployment of discretion [Wilson, 1968, Waddington, 1993]. The rationale and the reasons for discretion are due to the ambiguity of the law and the need for some form of legal and moral norm. As a result of this clouding of roles and purpose, the police have to devise a system of working. Within this is the need to be flexible [Goldstein, 1963]. The lack of a coherent and definitive legal framework allows for the application of flexibility. The vague nature of police work means that, while discretion is not necessarily an overt policy, in order for the police to accomplish their many roles it is a necessity

With discretion the officer exerts considerable influence and power within the community. This is particularly relevant when looking at the community policing approach. The proactive style incorporated within the notion of community policing may result in the 'arbitrary and capricious use of police powers' [Vinzant et al, 1994, page 190]. What appears to be a significant factor in the act of discretion is the

community itself. The involvement of the community, its assumptions, perceptions and beliefs influences the extent of police discretion. This approach realises that the police cannot be entirely separated from the belief or value system of the environment in which they serve. The police value system, specifically around the area of crime and punishment, can be illustrated by the work of Fielding et al [1991]. Police officers generally have a 'conservative' view of crime. They believe that any problem behaviour in a community assimilates from a few individuals. They use this concept as a measure by which to judge the behaviour of the rest of the community. Essentially, the police in order to fulfil their tasks have to perceive their own 'normality' [Fielding et al, 1991].

These concepts help to shape the belief system of the officer and contribute to the construction of discretionary choices and actions. The problems of the community are placed within the remit of the officer to resolve. This questions the exact role of the police. One has to ask whether officers are social workers, community leaders, or a referral agency for other services [Vinzant et al, 1994]. This police practice recognises that the officer needs to have a sense of empowerment and responsibility to make choices within their specific community [Mastroski et al, 1995].

Central to this argument is power and the use of force. The core practice of policing involves the control of individuals and an unequal distribution of power [Smith and Visher, 1981]. When an officer uses their decision-making ability in assessing the choices available to them, there is also a presumption that they can determine the level and extent of coercion required to deal with circumstances. However, when implementing a more proactive community approach, the police require a variety of skills to ensure the completion of their tasks. Using a theoretical basis of power to explain the position and practice of officers in community policing is insufficient

[Vinzant et al, 1994]. Discretion involves not only deciding on the use and type of force, but also the deployment of negotiation, communication and co-operation.

Goldstein [1990] defines this decision making as a logical process, involving the police in problem realisation, assessment and solution. This systematic behaviour of the police officer is seen as rational, especially when dealing with a particular community problem. However, this model of discretion as decision-making places significant responsibility on the police. Within this context they are seen as social commentators, analysts and reformers, defining and managing a mass of social dilemmas and problems, in addition to fulfilling the traditional tasks expected of the police.

2.24 The Police and the Mentally Ill: The Need for Discretion

It is apparent that when the police are required to make decisions that do not evoke legal precedents, their use of discretion becomes both essential and necessary. This is particularly the case when examining the police interactions with the mentally ill. The apparent lack of a consistent approach has consequences for both the police and the mentally ill. Within these situations the police negotiate not only their own safety and environment, but also a variety of considerations that do not appear to have a place within the police remit. However, the police cannot discontinue their involvement with the mentally ill. Daft [1980] notes that the police are the most consistent mental health service available to the mentally ill within the community. Smith [1990] remarks that the police are the most valued of the services for the mentally ill, providing essential assistance and never failing in their response to help. The comments by both Daft [1980] and Smith [1990] highlight the continuing necessity of the police to provide aid to the mentally ill.

2.25 The Police Role and the Mentally Ill

Contributions into the research on the police and the mentally ill can be seen emerging from both North America [Borum et al, 1998, Coie et al, 1975, Finn et al, 1989, Green, 1997, Holley et al, 1988, Lamb et al, 2002, Mathews Jr, 1970, Meehan, 1995, Menzies, 1987, Patch, 1999, Stein et al, 1985, Teplin et al, 1992] and Britain [Bean et al, 1991, Berkeley, 1995, Bowler et al, 2001a, 2001b, Bradley Taylor, 1996a, 1996b, Duff 1997, Gandhi et al, 2001, Rogers et al, Roughton, 1994, Smith, 1990]. They all acknowledge the role of the police as gatekeepers for the forensic services, the dilemmas faced and the implications for police practice and decision-making. As with other areas of practice, the assumptions of the police contribute to their decision-making process, aiding in the selection of suspects and the level of involvement. The police attitude towards the mentally ill plays an important function in how they identify problems and decide on the necessary course of action.

The problem for the police in defining their role in their interactions with the mentally ill is apparent. The police have contradictory roles, providing both care and control. They need to use a variety of techniques, including persuasion [Meehan, 1995]. The police are the first resource approached by those in need of medical help. Remarking on the central role of the gatekeeper as a link between the individual and the mental health services, Coie et al [1975] note that the police are often used the most widely as a source of assistance, and predominantly in cases where the individual is viewed as difficult. The police consequently act the role of 'psychiatrists in blue' [Menzies, 1987]. The theme of the police as ad hoc agents of the mental health system is common in a significant number of studies into the interactions between the police and the mentally ill. Bowler et al [2001b], Green [1997] and Teplin et al [1992] note the emergence of the police as 'frontline mental health workers' and 'street corner psychiatrists'. Equally,

Lamb et al [2002] remark that such involvement is enshrined in the requirement for the police to protect the community and act as a 'parens patria' for those deemed as in need of special help, such as the mentally ill.

Two key beliefs are shared by the police; that the mentally ill are sick, and therefore in need of medical assistance, and that they are potentially violent, with the police attempting to exit the situation as quickly as possible. Mental illness, for the police, is conceptualised in either the risk of violent behaviour or perceived abnormal actions. This, in turn, influences the practice of the police towards this group. The perceptions of police officers show similarities with the public and other professional groups [Lester Pickett, 1978, Matthews, Jr, 1970]. One of the main conditions for the police to decide to act is the potentially violent behaviour of the mentally ill [Matthews Jr, 1970]. Those the police encounter who display signs of mental illness are more likely to be seen as a future source of violence [Menzies, 1987]. The police appear not to clearly define criminality and mental illness, with the obvious combination of the two notions emerging in the referrals and police reports. This predominant feature of the assimilation of criminality and mental illness translates into the influence and discretion of the police in conveying this message to other professionals within the forensic system.

When they make judgements on the mental state and potential violence of the mentally ill, the police include a moral assessment. This moralisation of mental illness has implications. This adds further consideration to the police role as community psychiatrists. The result of these assumptions is that

"Mental illness manifest[s] itself to police as disrespect, recalcitrance, moral defect, and silence, instead of the more subtle indicators of pathology...."

[Menzies, 1987, page 446]

The police role in defining and labelling individuals mentally ill is important. These basic, subjective indicators are used to determine the presence of mental illness and the appropriateness of this label. While the police are good judges of mental illness, their moral appreciation of the individual effects how the interaction proceeds.

One consequence of the usual police experience in dealing with and managing perceived disorder is to increase their awareness and tolerance of extreme behaviour. Their concepts of normality and deviance are wider than those of other professions. The particular case of personality disorder illustrates the dilemmas in defining behaviour. It is evident that the police have increasing awareness of the issue of personality disorder [Bowler et al, 2001b, Gandhi et al, 2001]. The presence of personality disorder in the community causes significant concern. As Gandhi et al [2001] state,

"...contact with the police was potential proxy for 'community concern' because even if the incident was minor, they would normally be interpreted with some anxiety by immediate residents and other people in the vicinity"

[Gandhi et al, 2001, page 95]

Consequently, it is cases of extreme disorder or behaviour which come to the attention of the police. Again, this has to be within the current climate that accentuates risk.

A significant proportion of research in the field notes the important part the police play in initiating or re-establishing the psychiatric careers of the mentally ill. Green [1997] remarks on the position of the police at the beginning of the 'carceral continuum' [Foucault, 1977] and their role in protection and care. This carries with it the identified problems associated with care and control. Equally, various issues contribute to an officer's decision to arrest or to establish and implement other disposals. Three variables are identified which influence action; the offence, the experience of the officer

and an awareness of the individual's previous behaviour. These factors appeared to be key determinants in the use of discretion and the subsequent actions of the police.

2.26 The Police Approach to the Mentally Ill

The police approach to the mentally ill encompasses a variety of dynamics including time, place, familiarity and personal experience. The act of discretion is further displayed within the police choice of the occupants within each group. The police implement discretion not only by deciding what practice to undertake, but also in the membership of the mentally ill sub groups. One consequence for the mentally ill involves their entry into the criminal and mental health systems. The influence of the police to determine these factors is an area highlighted by Duff [1997]. Rather than the decisions being taken by the prosecutors in diversion schemes, it is the police who influence and determine, via their reports, the recipients of diversion. What the police state in the reports, the cues they provide for the prosecutors and equally what they do not say, provide a basis on which decisions are made.

The approaches available to the officers can be placed within three categories; no action, formal sanctions such as arrest and the informal disposals. There is a consequence of this ad hoc system, which involves the use of more informal measures, especially as generally the police are the least likely to commit to hospital. When distinguishing between the normal roles of the gatekeeper disciplines, the police as 'guardians of the social order' have a certain level of autonomy and discretion in deciding the various disposals to implement [Coie et al, 1975]. The duality of the role of the police in their management of the mentally ill, as an access point for both the criminal justice and mental health systems, requires negotiation. This occurs both on an individual, situational level and also at the interchange of the two systems. A frequent

difficulty for police centres on obtaining treatment and admission to hospital for individuals [Green, 1997, Patch, 1999, Teplin et al, 1992].

The police experience a dilemma as to what action to take and where to place the mentally ill. Officers assume they will have to manage these situations and individuals again and again. They attempt to access medical treatment for individuals who are often released back into the community without the necessary assistance. What Green's [1997] study highlights is the short-term nature of many of the police solutions and the inability to address the fundamental mental health problems of the individual. The most that the police can achieve is a temporary cessation of troublesome behaviour. While the police appear to have numerous discretionary options available to them and a certain amount of control as to what action to take, the outcomes for the police are far from productive.

Each case faced by the police is individual and how they respond is dependent on a number of factors. In some examples, there appears to be a distinction between officers of the approach to take with the mentally ill. Those officers with more experience tend to adopt a position of 'no action' when encountering a mentally ill person who may have committed an offence [Green, 1997]. This is in contradiction to those police officers with more limited experience who favour arrest as opposed to more informal sanctions. The experienced officers 'ignore' the mentally ill, taking no action, whereas the officers who had served fewer years in the police tend to interact with the mentally ill, with the result that they either resort to arrest or implement more informal methods.

The reasons for such distinct behaviour patterns include an awareness by experienced officers that certain options available to them for managing the mentally ill are limited and inefficient. Generally, the police develop their style and method of working through

a process of shared information and education. The established officers transfer their knowledge of the 'system' onto the less aware members of the force. By the mechanism of self-learning, the officers learn to evaluate the various options available to them.

Green [1997] argues that through this the police realise the possibility of using an informal approach against the more formal disposals, where there is a feeling that they are in a 'no win' situation. There is also an issue concerning the motivation for officers to interact and deal with the mentally ill. There is an assumption that the more experienced officers no longer have an inclination to engage with the mentally ill, especially if there exists a sense of frustration as to the courses of action they can undertake and questions around their effectiveness.

These studies illustrate the transitory solutions of the police and the apparent failure of the system in managing the mentally ill in the long term. Arrest is often used as a means of obtaining assistance for the mentally ill [Teplin et al, 1992]. While hospitalisation is one option available to the police, and would suggest the most effective method of obtaining treatment, Teplin et al [1992] found that the police are discouraged by the structure and system in place. The bureaucracy and the strident criteria for admission curtail the police in actively pursuing hospitalisation. The solution for the police is to determine and assess other options. Subsequently, arrest is an option through which the police could negotiate and control the situation, especially in three key areas.

- when the preferred solution was initially hospitalisation and the officer perceived difficulties in admission.
- when the mentally ill individual is visible within the community and is causing distress to other citizens.
- when the police perception sees the individual as a continued problem. The likelihood of this course of action would increase if the police felt that they have to subsequently return to the incident.

These actions are the result of 'necessity' rather than design.

Predominantly, it is informal measures that characterise police behaviour and practice towards the mentally ill. Over seventy percent of incidents and situations concerning the mentally ill are dealt with informally [Teplin et al, 1992]. They are more likely to be treated by the police with informal measures if they occupy the following groups; the 'neighbourhood characters', the 'troublemakers' and the 'quiet crazies' [Patch, 1999, Teplin et al, 1992].

One factor that influences police decisions is the degree to which they and their behaviour patterns are predictable. Those considered 'neighbourhood characters' are considered predictable with the police acquiring knowledge and familiarity. They are tolerated more than other sections of the mentally ill. With this group the police are fulfilling their role of street corner psychiatrists, assisting the mentally ill in their maintenance within the community [Teplin et al, 1992]. This contrasts with their perceptions of the 'troublemakers'. Here the police routine is disrupted. While there may be more appropriate formal measures, the police view them negatively. The officers consider the informal approach as 'the resolution of choice because troublemakers are thought to be too difficult to handle via either arrest or hospitalisation' [Teplin et al, 1992,page 153]. For the final group, their invisibility allows the employment of these informal sanctions. The police view this group of 'quiet crazies' as less serious and less likely to be in need of either arrest or admission to hospital.

For the police, the problems of managing the mentally ill in the community centre on finding the quickest and easiest solution for themselves and the mentally ill. This involves using their existing skills and powers to resolve a demanding situation. The

current climate of mental health has placed further pressure on the police to manage situations involving the mentally ill. As Bradley Taylor [1996] states,

"One inevitable result of the care in the community programme has been to bring police into ever-increasing operational contact with mentally ill people. The disturbed man who shouts and gesticulates....in the hospital ward will be rapidly calmed down by trained nursing staff. The same incident in the shopping precinct will frequently lead to the attendance of the police whose primary concern is to stop the disturbance...."

[Bradley Taylor, 1996, page 300]

2.27 Police Dilemmas

The police experience of the mentally ill suggests that they occupy a fundamental role in delivering both control and assistance. It is one that creates a variety of dilemmas. It illustrates the necessity of discretion in managing the mentally ill. In one sense, the police reaction to the mentally ill is the same as that displayed in their interactions with numerous groups. Their attitude and concept of their environment, in addition to the structural and situational factors highlighted by theories such as that outlined by Bourdieu, impact upon their interactions with the mentally ill and other sections of society.

As illustrated, the police utilise skills, knowledge and an awareness of practical considerations to manage the mentally ill in the community. In the context 'keeping the peace' and 'psychiatric first aid', they institute a variety of mechanisms, with the aim of quick and effective resolution. The complex situation of the mentally ill in the community, for the individual, for care agencies and the police, requires the development of management mechanisms. Discretion provides the police with a series of options, in addition to a framework in which to understand their actions, behaviour and perceptions.

The dilemmas faced by the police in dealing with the mentally ill occur on a variety of levels. There are organisational demands aimed at ensuring a certain level of order and public security. There is an element of care involved in assisting the mentally ill, especially when in crisis. There are professional demands that dictate a certain level of inter agency working. The social features of the mentally ill, additionally, places requirements on the police. This questions their ability to deal with the diversity of multiple issues. One also has to draw attention to a possible function of the police within these social encounters and the issue of public expectations. The police removal of the mentally ill from the public view is based on notions of their behaviour and actions. It forces the police to act as gatekeepers of the social concepts of normality.

There are several characteristics presented to the police in their involvement with the mentally ill [Bean et al,1991]. While the police do not directly deal with issues of homelessness or drug use, their influence in bringing the mentally ill to the attention of the police has to be taken into account. The social circumstances of the mentally ill contribute both to their mental illness and an increased likelihood of police and criminal justice involvement. The predominance of a disadvantaged socioeconomic position effects their general situation and their propensity to become involved in the criminal justice system [Boast and Chesterman, 1995].

The question that these findings initially raises involves looking at what function is being served by incorporating the police in the management of the mentally ill. When examining the police as an organisation and their various roles and practices, there is a need to provide a basic analysis as to why the police are seen as a source of containment and help, bearing in mind the general view and perceptions of the mentally ill. It is partly the control function of the police and of the wider criminal justice system [Boast

and Chesterman, 1995]. Society places expectations on the police to deal and manage the mentally ill in the community [Bean, et al, 1991]. The fear of the mentally ill and the apprehension that surrounds their visibility increases the public demand for action. The police involvement reinforces this fear, providing confirmation of the dangerousness of the mentally ill by incorporating them into the criminal justice system. This process of legitimising fear accelerates levels of anxiety. This, in turn, incorporates the police in further management and containment in an attempt to reduce this sense of panic. This cycle of apprehension and control will continue as long as the unsubstantiated fear of the mentally ill is not addressed.

2.28 Conclusion

It is evident from this review of the developments in mental health care and the police management of the mentally ill that important issues remain. The research studies looking at the police have emphasised aspects of their behaviour, actions and organisation. Research has included the following areas: police culture [Chan, 1996a, Shearing and Ericson, 1991], the use of discretion [Reed, 1980, Reiss, Jr, 1984], the police ability to identify suspects, and the subsequent process of action [McConville et al, 1991, Sklonick, 1975, Sykes et al, 1975], their relationship with numerous sections in society [Berkley, 1995, Meehan, 1995, NACRO, 1989], the police personality [Dixon et al, 1989], the police attitude [Horn and Hollin, 1997, Wilt and Bannon, 1976], the distinction between the various hierarchical levels within the police and their impact on practice [Grimshaw et al, 1987, Reiner, 1992a], and their powers and the situational factors which interact with the police sensibility to produce action [Smith and Visher, 1981, Wortley, 1997]. This is in addition to those studies focusing on the police encounters with the mentally ill, such as work by Borum et al [1998], Bowler et al [2001a, 2001b]. Bradley Taylor [1996a, 1996b], Duff [1997], Gandhi et al [2001]

Lamb et al [2002], Meehan [1995], Menzies [1987] and Teplin et al [1992]. They all acknowledge this function. One of the fundamental dilemmas in examining the police and their practice has been deciding the nature of their role. It is this dilemma that raises issues for the police role with the mentally ill.

What appears relevant today is a re-examination of the police involvement with the mentally ill. While some discussion has occurred on a broad spectrum concerning certain issues, there still remains a need to look at the police in light of the changes in recent mental health policy and particularly the experiences of officers in the UK. The research and the literature has also provided a wide discussion on such issues as the nature of policing the hierarchical struggles [Grimshaw et al, 1987, Reiner, 1992b] and the modern role of the police [De Lint, 2003]. There are also possible benefits from examining the police perceptions of their involvement, in addition to the practicalities of managing this group.

The literature and research indicate that the context of mental illness, the social policy response to its presence and the issue of the police management of the mentally ill are inevitability interlinked. The changing climate of mental health care and the developments and priorities in policing raise questions around what we expect from the police force and the resulting management of the mentally ill. The research study attempts to look at the police response and the issues that concern a group of officers in this changing area. How officers perceive their role and their concerns around involvement with the mentally ill have relevance for ideas on modern policing.

Chapter Three: Methodology and Methods

3.1 The Research Problem

Mental illness and society's response to its presence is fraught with contradictions and difficulties. For the police, their professional responsibility ensures that they have to develop mechanisms for coping and dealing with diverse predicaments. The focus of this study looks at the response of the police to current issues of the mentally ill in the community. Several questions occur regarding the police and their experience of dealing with the mentally ill in the community. The main areas and questions for this study centre on the following:

- How has the emergence of a new direction in mental health policy influenced the role and perceptions of the police?
- How have issues around risk and the dilemmas around care and control influenced the police?
- How has the policy response of the police proposed a change in the police management of the mentally ill?
- What approach do the police officers expect to take towards the mentally ill, which additionally recognises their own perceptions and definition of roles.
- What role/s are identified by the police when managing the mentally ill
- What factors do they consider in responding to the mentally ill in their community

The aim of the study was to examine the issues relevant to the police in their management of the mentally ill in the community. The research focused on how officers perceived such involvement, their approach to the mentally ill and the various dilemmas faced. It also examined the alternative solutions that, from the point of view of the police, would benefit them as officers in the community. The research was conducted on a limited group of officers [operational and managerial] over a specific time period,

between 1996-1998. What is presented, therefore, is a snapshot of the police management of the mentally ill in the community.

3.2 Methodology

In order to examine these questions, a qualitative approach was considered necessary. Qualitative research allows for a detailed examination of the social world of individuals. It aims to develop insight to the their practice and reality. It suggests a deeper process than merely ascertaining the number of encounters and frequency of events. It implies that there is an instinctive and observable approach involved in the formation of a social world. With the possibilities of insights to be gained into police practice and their perceptions, this approach has been favoured for this research.

The argument for using a qualitative approach such as participant observation and interviews was to increase understanding of the structure of a particular social world. Within the study of the social interactions, to focus on statistics and facts, rather than the meanings, perceptions and interpretations, limits the picture that can be gained. The advantage of using a qualitative approach was to establish a hypothesis of how actors in that world operate. The aim by using this approach was to provide a view of what is occurring for the police at a particular time with reference to the social area of mental illness.

The use of qualitative research, particularly in examining the police, is not to establish the basic facts of their encounters, such as the frequency and number of contacts with the mentally ill. The use of official statistics suggests more about the organisation than the actual social encounters. Quantitative analysis, when based on official statistics, may prove unreliable in view of the fact that 'statistics' per se are social constructs [Ritzer, 1992]. They can be used or misused to portray a particular 'social fact' as is

required by those who construct the figures. In the police-mentally ill process, the lack of substantial official figures on the type and level of involvement illustrates the difficulties in categorising these incidents. It also highlights the police belief that implies that this area is not within their 'usual' domain.

Therefore, whilst an awareness and use of statistics has relevance for forming an overall picture of police encounters with the mentally ill, the descriptive element encompassed within a more qualitative standpoint allows for a further understanding of the perception and meaning of police officers. From the argument presented by Becker [in Jessor et al, 1996] the purpose of this type of research is not to define and determine the presence of certain phenomena, but to remark on the connections between incidents, events and practice.

3.3 Bourdieu's Notions of Habitus and Field

Bourdieu offers a framework to understand police behaviour. He examines the ideas around social practice and what people do in their everyday experience. Bourdieu examines the details of everyday life and how individuals construct action. It also involves analysing the behaviour of these individuals in light of their experiences. Two connected features are highlighted; the individual 'habitus' and the wider 'social field'. The usefulness of these notions to the police experience of mental illness is various. The area of habitus and the police identity assumes that a professional habitus would be formed due to role socialisation and the use of existing dispositions.

The suggestions made by Bourdieu's work may offer some explanations for the actions of the police with the mentally ill. Questions arise concerning the theoretical usefulness of such terms and the possible interference one has with the other, seeing any macromicro link solely in terms of the transferring of the social norms developed with a

cultural habitus to individualised dispositions [Bohman, 1991]. While such criticisms have to be at the forefront of any analysis of Bourdieu and its relevance for any examination of the police, the explanation offered still has relevance for developing a possible framework for understanding.

Bourdieu's foundation for the ideas of habitus and field involves practical rather than theoretical concerns. The concept of habitus was utilised

"... to get out from under the philosophy of consciousness without doing away with the agent,"

[Bourdieu, 1985, page 14]

To simplify Bourdieu's arguments for the habitus and the field may be to ignore the finer details of his theoretical standpoint. However, it serves to form a basis for understanding and one which may be applied to the experience of the police.

By habitus, Bourdieu is referring to the internalised structure, by which people perceive, understand and judge their world. Essentially, it is a mental process and one which guides the practice of the individual. Such notions additionally recognise that for the habitus to develop and exist, there has to be a form of social interaction. This interaction is not only between individuals but also their environment. Rather than act as a preordained rulebook, the habitus provides an outline of options and alternatives.

Bouveresse [in Shusterman, 1999] argues that instead of visualising 'rules', Bourdieu talks of dispositions and strategies within the habitus. A disposition, in broad terms, refers to a propensity or inclination [Jenkins, 1992, page 76]. However, it also applies to a state of being and the outcome of actions. It is essentially a form of structure that is well established within the cognitive processes of the individual. [Bouveresse in Shusterman, 1999].

There is a danger that a belief in systems leads to generalisations. Bourdieu appears to distinguish between different forms of habitus, particularly between an individualised schema and a collective construct. The cultural and historical issues put forward by this argument for a collective habitus implies a sense of objectivity. He sees it as existing 'outside' of the individual. This basis in history is the 'foundation of the habitus' [Jenkins, 1992, page 80]. It also provides a wider reference for the establishment of collective reality, and subsequently, practices. Notably, when examining schemas and methods of action, habitus can be described as the

"...result of long term occupation of a position within the social world."

[Ritzer, 1992, page 578]

The link between habitus and practice is a manifestation of the complexity of the social world; habitus is created by practice, which in turn generates the social world itself.

Conversely, it is also assumed that within the habitus is the notion of predictability. It behaves as a 'basis of a forecast....., despite the fact that 'it is not based on any rule or explicit law' [Bourdieu in Bouveresse, 1999, page 60]. It appears that this conflicting basis of habitus, as both a predictable set of dispositions that result in practice and as a options of available alternatives, raises concerns over the theoretical arguments for its existence and practical application. Discerning how individuals act amongst this myriad of considerations and assumptions can become confusing.

It can appear that individuals resort to their habitus purely as a method of acting, which is primarily situated within their unconscious, rather than adopt a careful consideration of events [Jenkins, 1992]. Bourdieu, however, acknowledges that the decision making process allows for limited conscious awareness of practice or actions. Reference is made to a state of 'less consciousness', a state in between, which accounts for deliberate

consideration within a framework of the subconscious foundation of dispositions.

Subsequently, it is the habitus which provides the basis for the decisions of the individuals, instead of a divorce between rational decision-making and the formative foundation of their dispositions.

Difficulties arise, however, in determining not only the nature of practice, but also the translation of habitus to practice. Jenkins [1992] raises concerns with regard to the application of habitus. Bourdieu's notion of practice is linked to his notions of habitus and field. How one's dispositions transform into practice is difficult to determine. Equally for the police 'habitus', it is difficult to distinguish between the police culture and individual practices. However, the use of these concepts to provide a possible framework for understanding behaviour recognises that, in being a police officer, there are distinct characteristics and practices. They are the result of experience, socialisation and the social environment.

The transfer of subjective understanding into actions in an objective sense, which Bourdieu applies, suggests a form of determinism and even a limited ability of the individual to change his world. Therefore,

"...the habitus is the source of 'objective' practices, but is itself a set of 'subjective' generative principles produced by the objective patterns of social life. Such a model is either another version of determination...or a sophisticated form of functionalism."

[Jenkins, 1992, page 82]

There are several considerations involved in practice; its time framework, where 'time is both a constraint and a resource for social interaction' [Jenkins, 1992, page 69], its place in space, the notion of 'practice logic' and the sense of events happening. Here it appears that Bourdieu is using several metaphors for action, and specifically with

reference to the nature of the 'game' and 'game playing'. The individual not only shapes his own social world, but is also an intrinsic part of the particular event.

Understanding the social world becomes part of the process of practice.

This particular stance recognises that individuals, while having aims and directions, also are positioned within their own sense of the social world. It is the marrying of the subjectivity of the actor and their habitus with the objective reality of the field that Bourdieu argues for. He suggests developing a view of the actor's own experience, rather than adopting theoretical models in which to place it [Jenkins, 1992].

What distinguishes Bourdieu from more interactionist approaches is his focus on 'agency'. Ritzer [1992] summarises Bourdieu's stance

"...Bourdieu focuses on the dialectical relationship between objective structures and subjective phenomena....To side-step the objectivist-subjectivist dilemma, Bourdieu focuses on practices, which he sees as the outcome of the dialectic relationship between structure and agency. Practices are not objectively determined, nor are they the product of free will."

[Ritzer, 1992, page 577]

This necessity to avoid the division between the ideas of subjectivity and objectivity results in Bourdieu's belief in using practice as a method of observing the individual within their own contexts and social field.

3.4 Bourdieu and the Social Field

Regarding the notion of a 'field', Bourdieu interpretation relies on the work of others, such as Weber, and the transfer of the concept of a religious field to other areas of social activity. By elaborating on the notion of 'field', Bourdieu is recognising the similar traits that exist within the social world.

The notion of a field is inevitably linked to the habitus. Habitus cannot exist without a social field. It operates within this field and is responsive to different situations and circumstances. The combination of the various fields results in a societal model, or 'social space'. By field, we are referring to a 'arena of struggle', with each field displaying its own 'logic' [Ritzer, 1992, page 579-580].

In this sense, the field is an external and objective setting in which individuals apply their habitus. Subsequently, it is where practice takes place. Within the field are social positions, which suggest positions of power between agencies, institutions and individuals. The idea of struggle implies that each position is competing with others for resources. The 'power games' envisaged within the field comprise not only economic and political resources, but also social, cultural and intellectual.

While it has been argued that fields vary, both in logic and across time, certain features pertain to a common description of a field. The commonality of fields, demonstrates that

"each has its dominate and dominated, its struggles for usurpation or exclusion, its mechanisms of reproduction... But every one of these characteristics takes on a specific, irreducible, form in each field."

[Bourdieu, 1989, in Jenkins, page 86]

Certain aspects of the field are identified, including the relationship between the individual habitus and the structures of the field, and importantly, the possibilities that exist in this arena [Jenkins, 1992]. According to Dreyfus et al [in Shusterman, 1999],

"Human beings are socialised into this system of dispositions that enables them to produce on the appropriate occasion skilful social activity that embodies, sustains and reproduce the social field that in turn governs this very activity."

[Dreyfus et al, in Shusterman 1999, page 86]

3.5 Bourdieu and the Police Experience of Mental Illness

While Bourdieu provides a wide theoretical stance in terms of the practice of society, questions remain as to its relevance for the police, and particularly in terms of their encounters with the mentally ill. On a fundamental level, the police have a certain habitus, which may be outlined as 'culture'. This suggests a certain 'way of being', which includes a set of dispositions that aids the police officer in their work. It provides the individual with a 'tried and tested' response to various situations encountered in every day practice. The police, in their interactions with the mentally ill, develop a sense of routine or regularity in their choices as a result of their habitus. This occurs when acknowledging that situations are often unpredictable. The police also have a 'collective habitus' which is the result of shared experiences and common situations.

Equally, the police 'habitus' reacts to the situation faced. The police have to respond to a variety of incidents quickly and effectively. They have to rely on a several skills and techniques that help them manage particular situations. It also aims to inform the police of the features of the incident. It provides a 'mental backup' which enables them to decide what action to take. It can also be argued that there are certain features of the police habitus which come into play when faced with a social problem. These include methods to manage the conflict arising from caring and controlling the mentally ill. The police 'habitus' facilitates the police in negotiation of their environment. It achieves this by providing them with 'ready made' answers to the problem of managing the mentally ill in the community. Bourdieu implies that 'habitus' provides individuals with an inclination to act in a certain way. The use of habitus' in this research suggests that the police have a set of responses for dealing with the mentally ill in the community which are engrained. These include developing strategies aimed at the minimal disruption and

which manage the mentally ill effectively and quickly. This aids officers when managing social situations and negotiating their social environment.

The police, additionally, operate within a distinct social field, with its own definition or logic, which influences police action and which, in turn, is influenced by that action.

There is an apparent divergence between the supposed objectivity of the police world in the form of guided behaviour via their organisational demands, and the subjectivity of their response to differing situations.

In one sense, the social field of the police is a struggle between the maintenance of their notion of order and policing. Equally it also involves the individual dispositions that inform their practice. These may be in conflict with the wider aims of the social field or police organisation. Their social world involves a variety of alternatives, of learnt responses which are incorporated within their habitus. Underlying this specific context is the importance of recognising the predominant structure that enshrines the actions of the police. The police field in question is not merely a setting for power struggles, but also involves an interpretative process. This field incorporates the external factors that influence the habitus of the individual officer in their response to the mentally ill, such as the policy framework and their organisational response.

The arguments for using this approach with regard to the interactions of the police with the mentally ill was to find out the prerequisites of police work, and to 'see' where possible what it means to be a police officer when, potentially, their perceived role as law enforcers is extended. It is an attempt to understand the perspective of the police, and how they perceive their involvement with the mentally ill. It aims to view their world, if not from 'their shoes', at least from their stance. This is within the context of organisational and legislative pressures and changes, or using Bourdieu's term, the

social field of the police officer. This may have an impact on police behaviour and action.

3.6 Methods

The methods which I have attempted were determined by the affinity between the perspective and the methodology; a preference for a qualitative approach. At the same time the methods adopted were guided by practical constraints. Permission and access to respondents, the practical restrictions of the setting, such as the problems in directly observing the police, and the referral and choice of individuals for the interviews all influenced the research process.

3.7 Methods of data collection

From participant observation to statistical analysis of police actions, attempts have been made to examine and explain police behaviour and action [e.g. Patch, 1999, Teplin et al, 1992]. A more pragmatic argument continues with the insistence that to understand social actors and subjects, we need to know what they 'do' [Ritzer, 1992]. Incorporated within this is the changing or dynamic nature of this reality or alternatively the existence of reflexivity. The links between these notions and the work of Bourdieu in the area of practice and observation are apparent. It is necessary to note the complimentary approaches and the fundamental importance of understanding the 'practice' or actions of the individuals to understand their behaviour.

Two main methods were considered for this research. They were semi structured interviews and observation of police practice. When choosing this qualitative approach, it was anticipated that some level of participant observation of the police with the mentally ill could take place. It was considered useful to examine how the police managed an incident involving an individual with mental illness. This would have

provided additional information as to what the police 'actually' did when faced with a particular social problem. This observation, in conjunction, with interviews may have provided more insight to the dilemmas faced by the police.

However, the difficulty when researching such institutions as the police is the acceptance of the researcher into a direct participant observation role. To directly observe the police involves overcoming obstacles, including the difficulties in providing a safe environment for the researcher in the field and the police acceptance of observation of their practice. It was anticipated that I could directly observe the police practice, including such as aspects as their relationship with the mentally ill individual and other professionals.

However, in discussions with police managers, it was felt that this would prove difficult to organise. Concerns centred on ensuring that I would be safe whilst in the community and, from the police perspective, that I would not negatively effect the practice and the work of the police. As no resolution to these difficulties was found, the method employed involved the use of interviews and an examination of relevant documentation. The methods of data collection included:

- interviews with operational officers.
- interviews with managerial police officers of the rank of inspector and above, who were either responsible for mental health policy of the force or had a particular interest.
- examination of the police procedures, policies and guidelines for dealing with the mentally ill in the community. The documents included section 136 procedures, policies on diversion, joint procedures on mental health issues, training packages and research documents.

The interviews with both sets of officers were conducted with the use of a semistructured questionnaire (see appendix). The semi-structured questionnaire for operational officers was categorised in such a way to reflect the process of the police interaction with the mentally ill. As Dey [1993] acknowledges such procedure is 'forward looking', anticipating the outcome of the research, as opposed to a retrospective look at the data. In doing so, it is anticipated that the social features and factors within the police encounter with the mentally ill will be included.

In developing the relevant areas for research, several processes were instigated to ensure the inclusion of significant areas. These included the obtaining of information from an overview of previous research into the police encounters generally, and with specific reference to the police and mental illness, and from informal discussions with managerial officers. Thus the questionnaire covered the following issues:

- The context and initiation of the encounter and how the interviewee came into contact with the mentally ill
- The ability of the police to identify mental illness and their mechanisms of determining the presence of mental illness
- The factors prioritised and the reasoning behind such action. The main focus was on what the police do when faced with an individual with mental illness and the factors considered. This concerned their methods of dealing with the mentally ill and the skills employed.
- Their perceptions and examination of the dilemmas faced by them as police officers. These included care and control issues. Within this possible solutions as to how the police, if given the choice, would manage the mentally ill the community were also addressed.

For managerial officers, the questions focused more on the policy and the development of a organisational response to the mentally ill. The semi-structured questionnaire used in these police interviews examined the following areas:

- The police involvement with the mentally ill. This included questions around implications of mental health policy on the police and any estimates as to the numbers of mentally ill who come into contact with the police.
- The policy of the force for dealing with the mentally ill and the development of schemes, including Home Office directives and locally organised schemes. This section also recognised the changes to policy that have occurred.
- The practical implementation of policy and the perceived effectiveness. This includes expectations of the police managers and the transition of policy into practice.

By interviewing both operational and managerial officers, it was hoped that an organisational and operational view would emerge. In recognising the role of both every day practice and the policy governing that practice, a wider picture of the management of mental illness by the police may be achieved.

For the examination of the various documentation, several issues were relevant. These included:

- Timing of the document when were they produced i.e. in relation to key policy changes in the field
- The source of the document where they completed on an organisational level or an indication of individual interest or impetus.
- The extent of guidelines and procedures- practical implementation of the guidelines

Such methods attempted to ascertain the extent and type of police involvement with the mentally ill in the community. However, as with any methods utilised, there are a number of issues that need to be addressed.

3.8 Practical and Ethical Issues in the Methods Used

The assumptions in interpreting any interaction between the subject and researcher are various. Included in this appraisal should be an awareness that some information is limited, wrong meanings or interpretations may be given, or that the subject may

change their responses and their meanings, depending on circumstances. As Bourdieu comments in his discussion of methodology, there exists a range of data gathering techniques. Each method will, therefore, be reflected within the research findings. This is in addition to the dual role expected from the researcher as both insider and outsider.

Questions of reliability and validity arise, particularly when acknowledging the varying consistency of the individual and groups' accounts and meanings. An underlying consideration, however, centres on the issue of guiding or directing the subjects into areas, instead of allowing them to provide their own accounts. The importance in 'unanticipating' responses is important [Becker in Jessor et al, 1996]. Assuming the answers prior to the given account limit the information obtained. It may also overlook certain aspects that are important in understanding the process. For,

"... the observation which requires less inference and fewer assumptions is more likely to be accurate, although the accuracy so produced might not be worth bothering with."

[Becker, in Jessor et al, 1996, page 64]

The researcher has to develop an understanding and awareness of their own practice and the impact on their research. The researcher in the process can be described as being a 'receiver' for the viewpoint of the actor in recounting their social environment. The regulation of assumptions requires the impartiality of the researcher and the desire to 'prove' a social encounter to be ignored. By recounting events, the subjects could describe how they shape and determine their social world.

The issue of interpretation is important. When recalling events and formalising the process, subjects, including the police, are not necessarily providing a 'true' account [Shearing et al, 1991, Waddington, 1999]. Rather the notion of figurative language assumes that their accounts, be it in mythical repeated terms or as illustration, provides

insight into the police worldview; what they exclude becomes as important as what is included. The 'truth' that emerges is based on the internalised or subjective approach to appropriate action and placed within an externalised context of understanding the social world of police operation.

The dilemmas of subjectivity and objectivity have an effect on understanding the answers given, and differentiating between the two. Bourdieu argues that it is difficult to make a distinction between the two. They both describe the social world as seen from distinct viewpoints. The difficulties arise when asking respondents about their subjective meaning before objectifying their world, their field and habitus. The resulting emphasis on official accounts subsequently can lead to an account with limited foundation in practice. Within this analysis, such a response may tend to,

".. describe the state of affairs which ought to happen because the nature of the occasion inspires them to explain [or justify] their behaviour, in addition to [or instead of] describing it. The accounts which they produce are thus 'official accounts'."

[Jenkins, 1992, page 53]

Within the realm of the police, therefore, there may exist a tendency to further embellish accounts with reasons for such actions. With Bourdieu's discovery of the recourse to official answers being a recurring limitation, it should be recognised that the examination of the police world also involves understanding the basis of their practice.

The examination of police documents as policy and as guidance also raises certain issues. How answers are framed within the interview arena can be equally applied to the study of documentation. Why and how such procedures and guidelines are formed is as important as what they say. Documents may become distorted to reflect the needs and requirements of the organisation [and society] involved. As illustrated within the

examination of the organisational context, the use of formal guidelines can be seen as an attempt to control the actions of officers by curtailing the use of discretion. The underlying impetus for the development of policy and procedure has to be acknowledged, both positively and negatively, in order to understand its place within the context of the police interaction with the mentally ill.

There is also a risk in using semi-structured questionnaires. This concerns the guiding of the subjects into the required responses. The element of direction cannot be ignored, even in the open-ended semi structured interviews used. In distinguishing the areas to be examined, there is a possibility of ignoring other social determinants that may influence the behaviour of the police. It was hoped that officers referring to their experience, collectively and individually, would highlight their main experiences, meanings and perceptions of the encounter. The specificity of questions was limited to the key areas of the interaction process, and maintained throughout the interview process. The key features highlighted within the findings where confirmed and reiterated throughout the interviews by the officers themselves. The development of this particular process also recognised the time limitations of access to the officers for the purposes of interviews. While a full discussion took place in all the groups, an awareness of the reality of officers in their workplace meant that they were restricted in how much time they had available.

Equally relevant within this approach are the questions posed by the observer and the subsequent interpretation of the subjects. Methodological issues are posed within this approach, notably the reliance of the observer on the subjects' understanding of their practice. The generalised nature of the discussion and interpretation may hide the more

detailed actions of practice, with the questions asked reflecting this standardised and rationalised view.

What is apparent is that such a process is impinged with bias, subjectivity and dilemmas that affect the research. As Horn et al [1997] point out, the police culture and the emphasis on masculinity within the police force has implications for 'outsiders', both in an organisational sense and in terms of gender. Additional factors also contribute to the process of research and influence how that research is subsequently presented.

3.9 The Role of the Researcher

The role of the researcher within the overall process is obviously important as it influences the responses received and the findings presented. The social practice of individuals cannot be fully understood until the practice of the observer or researcher has been scrutinised. The need to know the researcher, their background and their practice are prerequisites to understand and evaluate the research process.

3.10 Background

My employment in the Personality Disorder Service at Ashworth Hospital at the time of the research obviously influenced the research process on several levels. As a mental health worker, and particularly in a forensic setting, several personal factors influenced not only the research topic but also the process. While it provided me with some insight into the issues of mental illness and the potential difficulties for the police, there was also an awareness of the potential for bias to occur, both in the selection of questions and the interpretation of findings.

It is important to recognise the dilemmas posed for the police and how they develop any methods of management. Why the police behave as they do and how they maintain their 'status quo' in light of the difficulties presented became important considerations. It was these aspects that further instigated my interest into the role of the police when involved with the mentally ill.

There remain additional ethical dilemmas in the process. As outlined previously, the observation of social actors within a particular environment is open to interpretation. The communication procedure involved is equally open to differing meaning, as is the interview process itself. Through my familiarity of the mental health and criminal justice systems, and previous experience of working in the criminal justice system, the interpretative process was made easier. However, I acknowledge the danger in making assumptions [as reiterated by Becker previously, in Jessor et al, 1996] as to the process and outcomes witnessed. To be familiar with the setting may imply a bias as to the anticipated responses of both the subjects and the process observed. It choosing these methods it additionally places emphasis on the possible bias that may result. The limitation of examining police practice by interview rather than direct observation acknowledges the reliance on asking the 'right' questions. Additionally, examining and detailing responses are open to interpretation. Making conjectures on police practice does not necessarily equate to police practice in reality.

While Bourdieu calls for participant objectification and a research process that 'steps into the shoes' of the subjects, such positions in reality prove difficult to formulate and maintain. Several limitations become apparent, notably in terms of the 'outside' observer being accepted into the world of the subject. Additionally, personal and professional considerations deter from full participant observation and interaction. The position of the researcher in response to the subject, especially if examining structures

that entail a different culture, albeit an organisational one, and the apparent distancing between the two effectively may establish a boundary.

There are obvious factors and features of the research process that influence the collection and collation of results. In addition to requiring police permission for access to their officers, the differences in interview processes raises issues. The police canteen culture often distinguishes officers from not only outsiders, but others in the police service. Much has been said with regard to the culture of the police [Chan, 1996, Horn et al, 1997, Reiner, 1992b, Waddington, 1999]. Often it is seen as a subverting force, corrupting the flow of policy into practice by overtly applying its own rules and agendas [Waddington, 1999]. Therefore, any research into the police has to take into account the individual and collective police culture of officers. It aids in how they not only perceive their role, but also act within that role. This was particularly pertinent in conducting interviews both on an individual and group basis and the resulting information gained.

3.11 Gender Issues

When discussing the role of researchers and their impact on the research process, issues of gender have relevance. This was particularly the case in terms of my role when examining the practice of the police. Gurney [1985] illustrates the concept of isolation in the case of the female researcher within a male dominated setting and the various roles encompassed within such situations. Noting the lack of reference to the specific experience of the female researcher, possible issues that may occur are signified. The 'devaluing' of the female researcher, the classification and establishment of subservient roles and the expectation of a non-threatening demeanour all contribute to the subjects'

perceptions involved in the research process [Gurney, 1985]. The possible effect on the research findings, due to gender and attitudes, is difficult to ascertain.

The methods to alleviate any bias include the avoidance of personal involvement and the maintenance of professional boundaries [Easterday et al, 1977]. Encompassed within this is the consistency of researchers in the application of their research and the issue of objectivity in relation to the status of the observer. There needs to be an acknowledgement of these issues within other demographic groups and the experiences of other researchers. The difficulties experienced by female researchers may be features common to all and exist in the working environment of a variety of settings [Gurney, 1985].

These aspects generally are particularly evident when examining an organisation such as the police. The nature of the police as a closed institution, with an established culture and set of working beliefs provides certain research dilemmas and potential difficulties. Horn [1996] highlights the need for the researcher to develop negotiation skills when working with the police in particular. In understanding the hierarchical nature of the police and the potential for difficulties as a result of top-level consent for research, the researcher in these situations has to acknowledge the prerequisite of establishing trust and rapport with the subjects. The need for 'informal access' often arises despite the acquisition of formal consent from the police managers [Horn 1996]. The belief that officers generally are suspicious affects the level of participation in the research process, coupled with identification made by officers of a researcher link with higher management. The achievement of this informal consent by the research subjects takes various forms.

The acceptance of the researcher within the police organisation is dependent on several strategies. The adoption of a non-threatening role, some semblance of credibility and a recognition of the contribution to be made within the research field by the police.

Referring again to my professional background, it is evident that this assisted in gaining access to the officers. The perception of officers of my role and their curiosity around Ashworth allowed for the development of a level of rapport. As the interviews occurred at the time of the Fallon Inquiry into incidents in the Personality Disorder Service, interest was further heightened. Additionally, areas of concern around the issue of personality disorder and officers' experience and understanding of the concept enabled an exchange of information. It may also have signalled to officers a level of awareness of the issues the police face in their encounters.

During the course of the interviews, similarities between Ashworth and its public image and the related experiences of the police in maintaining a positive image in light of criticism often levelled at the service became apparent. Comparisons can also be made with regard to culture. These cultural similarities included the shared belief in maintaining one's safety till the end of the shift and the use of 'black humour' as a coping mechanism. Therefore, a recognition developed of the similarities between the two working environments, the assumptions attached to each, and the effect on the workforce of these cultural factors.

As noted earlier, gender issues may also have contributed. I was assigned a non threatening role at the onset, as acknowledged by others researching male dominated settings. Horn [1996] notes, with regard to her own research, that the attachment of this role involves demonstrating support for the police, empathising with regard to their experiences in dealing with the mentally ill, and placing oneself at the lower end of the

hierarchy. These factors helped to establish a level of understanding and rapport, which allowed for the some exploration of key areas within the police-mentally ill interaction process.

The area of mental illness is equally open to interpretation. Whether one considers mental illness to be a social construct or not, individuals have their own perception of the nature and extent of mental illness. It is difficult to qualify how others perceive events and interactions.

3.12 The Setting

The prime setting for the interviews with managerial and operational officers was the police station. While a preliminary examination into the formal schemes involving the police and other agencies was useful, it became apparent that the police environment itself was more important. This was primarily in terms of trying to understand the practice of the police themselves, rather than their interactions with other professionals and agencies.

3.13 The Police Forces and the Sample of Managerial Officers

In choosing the police forces to examine several considerations occurred. The type and location, access and availability to officers all were factors involved in the research process. The restrictions in researching a social agency such as the police, its closed nature, and the practicalities in direct observation of interactions all effect the type of participants, the numbers involved and progress of the research.

At the start of the research process, six police forces were approached. The aim was to provide a diversity of forces, from both urban and rural locations, with their differing experiences of encounters with the mentally ill. Contact was made via a letter outlining

the rationale for the research and its aims. This was followed by phone contact and arranged meetings. All forces were provided with copies of the questionnaires.

To gain access to a group of officers involved formally applying for permission to police headquarters, outlining the aim of the research and the process to be undertaken. This process was in itself time consuming. From this process, two groups of officers were identified: those managerial officers responsible for policy and groups of operational officers based at specific stations.

Five police forces permitted interviews to take place with six managerial officers.

In obtaining access to the managerial or administrative officers, it was hoped that a view of the wider response of the police perception of mental illness would be provided.

The interviews with two hierarchically distinct groups of police officers was, consequently, an attempt to distinguish between the 'official answers' or 'official response' and the reality as the police in their every day practice viewed.

Six officers of the rank of Inspector and one Chief Inspector were interviewed to ascertain the official position of the police towards the mentally ill in the community. The officers involved had participated directly in the response of the force to the problem of the mentally ill in the community, through the development of schemes with other agencies or through instigating research and training in their particular forces. In other words, the managerial officers interviewed all were assigned responsibility for the force's guidelines, procedures and systems for dealing with the mentally ill. The combination of the two groups of operational and administrative or managerial officers can be found in the final chapter. However, as previous research has testified, the divergence between what 'ought' to be done and what activity takes place in practice is notable [Green, 1997, Teplin 1994].

Throughout the process, I continued to be reliant on the willingness of the police to let me into their environment, to such an extent that how the participants and groups were selected was at the discretion of the police managers. The dependence on the police themselves to choose and refer individuals meant that the police could influence the research by selecting those they considered suitable. This did not prove as problematic as first anticipated as the police allowed me a certain level of freedom, especially in terms of the questions posed, with no restrictions on the type of areas to be covered.

3.14 Sample of Operational Officers

While all these forces offered help in the form of documentation and general information, only two allowed direct access to operational officers. Both these forces were in the Northwest area of England. There was no link between the forces, other than geographical location. Each operated independently, with their own localised procedures and policies.

The access and selection of these officers involved negotiation with the police. The sample proved to be largely determined by the police force themselves. Attempts were made to distinguish interviewees via experience, role and rank. Restrictions were inevitable, such was the reliance on the police to co-operate in the research. However, in practice the issue of police selection only occurred in one station, and may have been more to do with the availability of staff than a covert effort to present a positive police image.

Although I was allowed access to the staff at this station, I was guided to some extent towards officers who had relevant experience of the mentally ill, and who were also available. Therefore, the group of officers selected was as a result of police managers' interpretation and their individual availability. However, it proved to be a cross section

of individuals, ranging from a probationer with six months experience to a CID officer with thirteen years. These interviews of six officers at this station took place in March 1998.

In the second station, however, access was allowed for all the operational officers at the station. All the five shifts located at this station participated. Out of a potential number of forty officers, access was obtained to thirty-two. The interviews took place throughout June 1998, with each different shift being made available for interview. Of those eight officers who were not interviewed, the reasons given included training and annual leave. In total and at both stations, thirty-eight officers were interviewed.

Officers were interviewed in groups and individually. This was a result of police organisation and the availability of officers. It was anticipated that all accounts would be recorded and later transcribed. However, this was not possible. As a result, the scripts of the account were written in the presence of the officers. While a tape recorder would have aided the process, diminishing the risk of losing information, this proved difficult on both practical and ethical grounds.

The practical concern centred on the reluctance of the police to allow the use of a tape recorder. Ethically, it may also generate suspicion and lessen the development of rapport, and by implication the likelihood of establishing the police understanding of their practice. However, it has to be equally acknowledged that writing responses in the presence of officers also has limitations and problems, including the development of trust and the reversal of role for police officers. While it is they who are use to obtaining information, they were being assigned the role of interviewee in having their responses recorded. In some respects this may have limited their openness to providing

accounts of their experience, and may have involved an initial wariness to the research itself. However, through issues highlighted by Easterday [1977], Gurney [1985] and the techniques stated by Horn [1996], I attempted to develop and maintain rapport with the officers through the perceived common experiences and through listening to their experiences.

The reliance, therefore, was on the writing of the accounts while conducting the interviews, and immediately after. This involved using the interview schedule as a guideline and detailing the replies in relevant sections. It also involved constant clarification by officers to ensure that the response recorded was accurate to them. However, it does not diminish the disadvantages that relying on an interview process as opposed to directly observing the police interaction with the mentally ill poses. It accentuated my ability as an observer and researcher to interpret the main features of the police accounts, and to avoid distorting the remarks of the officers interviewed, placing my bias, subjectivity and understanding of the situation above that of the officer. There is also a limitation in actually recording the responses, a recurring danger that 'something will be missed'. While the method used allowed the development of insight into certain aspects of the police world, it could not compensate for prolonged access and observation of the practice and actions of the police. The research methods therefore had to be designed to account for the actualities of the arena and the participants under view.

3.15 Data Analysis

The data collected via the group and individual interviews with all officers was examined and compared. The interviews with managerial officers was transcribed after the interview and all relevant documentation from each force was examined. The

documents from each force were compared, to highlight any differences or similarities. After each interview with individual operational officers and groups, the interview was also transcribed and themes identified from the statements of officers. Any similarities and differences were noted. These were there compared with previous interviews to ascertain any collective themes and topics. This comparative approach allowed an examination of clear themes to emerge. The response of one group of officers, therefore, aided in informing the next group. This process was maintained as the research progressed. Several themes were identified as a result of this process, and while certain limitations are evident due to the method of data collection, it ensured some level of consistency.

There are obvious difficulties in using both individual and group interviews. The data that was obtained displayed some differences between individual and group responses and experiences. The individual officers had more time to formulate their answers and were not aided or restricted by their colleagues. Within the group interviews, it was evident from the transcripts that certain experienced individuals contributed more than other officers.

Analysis of the relevant documentation from five forces was undertaken. Documents included the following:

- Monitoring forms for section 136
- Joint working and procedure guidelines
- Training packages
- Research documents carried out by several police forces

This analysis looked at what the policies stated and how they intended to be implemented. This should be seen in conjunction with the information obtained from the interviews with managerial officers. The documents identify the organisational

response to the mentally ill and the application of schemes that aims to serve the interests of those involved. It also highlighted how different police organisations deal and manage the mentally ill in the community. The findings from the documents are included in chapter six.

Interpreting written documentation for relevance and meaning carries the same obstacles as the interpretation of oral accounts [Hodder, in Denzin et al, 1994]. These include some of the issues previously highlighted such as the timing of the production of documents. For this study, this examination and analysis of documents was attempted in order not only to expand on the assumption of the recourse to 'official answers' and to understand the 'official' response, but also to acknowledge the deviation between the procedures as outlined in documents and the practice of the individuals interviewed. One example of this deviation concerns the development of schemes and policy guidelines on the use of diversion projects, which aims to place the mentally ill in the health system. There appeared to be a reluctance by operational officers to regularly use such schemes. Officers considered the various practicalities involved in accessing the schemes, but tended to rely on their own methods of policing to manage the mentally ill. This variation highlights the importance of examining the process from both the organisational viewpoint and the experience of officers.

3.16 Conclusion

Previous work has highlighted the need to see the police as fulfilling a variety of roles, and importantly, providing assistance for the mentally ill. This research aims to reaffirm such beliefs in the role of the police in society. This study, while recognising the work of previous studies, uses as its scope the current reactions of the police to a particular social problem, its impact on their individual practice, and the wider considerations of

police service. It aims to provide a depiction of relationships and practices. In doing so it does not make claims to provide all the answers to the police interpretation of the mental illness and their encounters with this particular group. The complexity of the police world and the expectations both on and of the police combine to produce a study whose claim can only be to offer a limited perspective into the encounters, experiences and interactions of specific police forces.

The focus was concentrated both in terms of the subject area and the subjects interviewed. The research deals with a certain group of officers who are subject to individual beliefs and experiences. This may prove difficult to translate to other police forces. However, the research does draw attention to one particular dilemma and raises questions that may be further examined and hopefully prove relevant to the police world of mental illness and social encounters.

This thesis aimed to look at the several components of the police reaction to the mentally ill, by examining the view of the police, their priorities and the solutions to the dilemmas posed. It has to be acknowledged that some actions take place within an organisational setting, containing a form of social structure. The research, therefore, recognises this by looking at both the policy and practice of the police.

It is important to reiterate that this study is primarily concerned with police perceptions at a given time. Future developments may influence police practice on a wider level. The findings suggest particular dilemmas faced by police in dealing and managing the mentally ill in the community within the context of modern policing and a fluid mental health policy.

Numerous themes emerge from these interviews with officers. The main themes include the following:

- The informal management of the mentally ill, the use of discretion and the problems of solving a social problem within a policing framework.
- The issues around the care and control of the mentally ill and the dilemmas surrounding their vulnerability and the police perception of the risk of violence.
- The policy response by the police forces and the variation in practice, which attempts to influence the behaviour of officers.

It could be argued that these themes influence and determine the development of the police 'habitus'. The themes identified by this group of police officers play a crucial role in their encounters with the mentally ill. They are seen as the main factors influencing police behaviour and action towards the mentally ill. As Jenkins [1992] acknowledges, the practice of individuals is located in the habitus. Their ability to make decisions is result of their individual dispositions. Any choice made and action decided is the result of these core characteristics, which have been developed over time. At a basic level, 'the habitus disposes actors to do certain things' [Jenkins, 1992, page 76]. The habitus of the police facilitates their practice towards the mentally ill by identifying the key options available. This is achieved via experience rather than through learning or teaching.

Part I

POLICE PERCEPTIONS OF THEIR ENCOUNTERS WITH THE MENTALLY ILL IN THE COMMUNITY

JENNIFER SHAW

2004

PhD

Chapter Four: Findings - The Risk Agenda and The Police

4.1 Introduction

When examining the police perceptions of the mentally ill in the community and their role, collective themes emerged for officers dealing with the mentally ill in the community. The nature of the police involvement with the mentally ill and the frequent contact with a group of mentally ill individuals meant that, for the groups of officers interviewed, similar scenarios appeared. The issues that were raised and their apparent existence in the police working framework emerged from the interviews conducted. While there are obvious limitations as to the extent and scope of the research [which will be discussed later], the findings provide an indication of the views of these officers and options available when dealing with the mentally ill.

The modern police officer has to not only appear professional but also achieve this in a climate of risk assessment and management. This is in conflict with the obvious vulnerability of the mentally ill. The notion of risk for the officer involved various components. These included violence and unpredictability, the use of force and the officers' perception of the mentally ill. This involved their understanding of mental illness and the characteristics they identified in the mentally ill in their community. For the police officer, there were obvious concerns and reactions to the dilemmas posed.

4.2 The Sample and Setting:

In total, thirty-six operational officers were interviewed. The operational officers came from two police stations in the Northwest. Within the sample group of operational officers, there appeared to be a variety of experiences.

Police constables were the predominant group throughout the interviews. Of

these officers, the majority were male. Most had several years experience as officers, with a significant number gaining most of their experience within the same force.

There are implications apparent within the dynamics of each group of officers, particularly for those from station B. This was evident when there were more experienced officers present. The more experienced officers, and particularly the sergeants when present, guided the discussion. Here they appeared to reiterated their own experiences. The officers with less experience mainly took the lead from their more knowledgeable colleagues

However, these findings overall attempt to provide some insight into the experiences of these officers with the mentally ill. The main findings from the operational officers are contained in this and the following chapter. The findings are divided into various sections, which highlight the main issues for the officers in the management of the mentally ill in the community.

The main concerns for officers in the community incorporate several areas, including the main source of contact and the police assumption of the behaviour of the individual. Predominant throughout is an awareness of the issue of risk.

4.3 The Identification of Mental Illness

In the every day encounters with the public, the police had to devise a system which served as an indication as to the character, behaviour and intentions of the individuals they faced. They relied on their experience, the wide-ranging notion of police skills, and knowledge of the area they policed. When that individual was mentally ill, the officer had to develop further their understanding of behaviour,

signs and cues to determine what was wrong. It was debatable as to whether this was sufficient, especially when considering the actions of the officer after the initial 'diagnosis' and the problems posed for the police.

The officers interviewed categorised and detected mental illness using a variety of methods and signs. This was linked to their own individual experience of mental illness. Equally, it was dependent on individual situations. Several of the officers in Station A referred to the development of instinct in informing them of an individual's mental state. These officers, in particular, had considerable experience, as custody sergeants and as a community beat officer. This may have influenced their ability to determine mental illness. This was in contrast to another officer in the station, who has been in the police for eighteen months at the time of interviews. He outlined his approach to identifying mental illness as involving examining more internal features, such as the individual's ability to deal with stress and the presence of depression. He acknowledged that if someone appeared to be distressed he would ask further questions and attempt to find out more about that individual. It is significant that this officer had less experience of mental illness than some of his colleagues.

The police in both stations acknowledged that they were not trained to detect mental illness. Their primary determination of mental illness involved no more than sensing that 'something was not quite right' about an individual. The following were responses and factors the police considered when attempting an assessment of mental illness.

Officers generally identified the following as cues for mental illness.

Table One: Police Signs of Mental Illness

1.	Unkempt behaviour/ not looking after themselves/ how someone looks
2.	Eccentric behaviour
3.	Demeanour/ 'wild eye' look
4.	Daft requests
5.	Not responding when questioned by the police
6.	Someone who is stressed
7.	Mood swings/ irrational behaviour/ rocking back and to
8.	Conversation/ what someone says /talking gibberish
9.	How they react in conversation

The arbitrary and wide-ranging comments around their method of perceiving mental illness indicated that the police rely on physical signs of mental illness. The content of conversation and the manner presented by the individual were cited by all officers as key indicators for deciding on the mental state of an individual. The police, although unable to quantify their ability, felt that these cues were accurate. This was in conjunction with the existing confirmation of mental illness from relatives or other professionals, within both the citizen and professional calls for assistance.

The police notion of 'something not being right' about the individual, and with the initial indication of mental illness, provided them with sufficient cues and indicators to precede to act. The reliance on existing methods of judgement in given situations was translated into the specific encounters with the mentally ill. The police, rather than develop a wide range of new skills and methods, incorporated the 'tried and tested' skills of policing, even in diverse situations. Being a police officer assumed a core

role that remained, and adapted to, changing social encounters.

4.4. The 'Police Regulars' - The Recurring Problem of Policing the Mentally III

One of the main issues at the start of the research was establishing the level of police contact with the mentally ill. It became obvious throughout the groups of officers and individuals interviewed that there were a group of mentally ill individuals who came into contact with the officers on a regular basis. The police officers in Station B estimated that there were thirteen individuals in the community who had regular contact with the police and who were mentally ill. Whilst this, for some, did not occur on a daily basis, the contact was considered significant. This was the case for the police constables generally. For officers in station A, this figure was hard to ascertain, with figures ranging from five to ten. One officer stated that he believed that the officers had contact with at least one mentally ill individual a week. This lack of statistical evidence confirms the mainly anecdotal evidence of many of these police encounters with the mentally ill.

For the custody sergeant at station A, their specific role recognised the context of managing the mentally ill. He outlined his role as mainly making assessments on arrest. He made the distinction between his usual clientele and the mentally ill.

"If a burglar is aggressive, this is because he has been caught, if they are mentally ill, it is due to their illness. Things have to be put into context and balanced with what they have been arrested for."

[Custody Sergeant, Station A]

There are various examples of the type of incident involving the mentally ill. The custody sergeant gave several examples from his own personal experiences, such as the call to a local hotel where a middle aged woman claimed to be the owner. She was released, moved onto another area and then was detained. The sergeant saw her mainly

as a 'nuisance'. A more serious incident involved a young woman at the local train station, who heard voices telling her to sacrifice her child. The sergeant described her as both being physically and verbally aggressive. She was later sectioned for seventy two hours. These examples of encounters with the mentally ill can be seen as occupying extremes described either as amusing or nuisance incidents, involving their local eccentrics and the more serious cases such as attempted suicide.

The cyclical nature of many of the police encounters with the mentally ill was a common feature.

"The regulars – I know them – I know what to do with them – its worse with someone I don't know."

[Police Constable, Station B]

The officers often saw the same people month after month. These police regulars were described by one officer in the following way,

"The eccentrics- you just accept them, you tolerate them – then there's the other mentally ill, and the violent. A lot are seen as personality disordered. Even the eccentrics though have the potential to be dangerous – you have to always be aware of this."

[Police Constable, Station B]

In addition to dividing the group of mentally ill into groups, the police recognised the frequent meetings with the mentally ill and the potential of violence, issues which will be discussed further. Officers in Station B distinguished between three definitive groups, the eccentrics, the 'other' mentally ill and the violent. The 'other' mentally ill comprised a group of individuals who were seen as to be distressed but did not share all the characteristics of the other two groups [e.g. 'odd' and humorous behaviour]

There was also an awareness of personality disorder throughout a number of the officers in Station B. This situation provided the officers with additional dilemmas.

They understood personality disorder to be a mental illness. The individual was seen

as disturbed and in need of help. For them, however, the diagnosis was given after the individual was taken to the local hospital.

"A lot of them are classified as personality disordered by psychiatrists. The reason for this is so they can be released with no service or hospital required. It saves resources. It's a standing joke amongst officers the number of people seen as personality disordered. It's used by psychiatrists to let them go."

[Police Sergeant, Station B]

The officers felt that reasons for use of this diagnosis had more to do the ability of the local hospitals to admit individuals. There was a perceived lack of resources available to manage this particular group. It was unclear from the interviews whether other factors such as alcohol and substance use influenced decisions, rather than only the presence of personality disorder.

It was the combination of the potential of violence and repeated contact which characterised the police-mentally ill interaction for all the officers. Despite this, the police often perceived these people as in need of help, but that they had either fallen through the social and health care services' net, or who had persistent problems, which facilitated their visibility to the criminal justice agencies. Officers maintained that the system was not dealing with them, making it more likely that they would come into contact with the police [Police Constable, Station B]. The risk issues concerned both the mentally ill as potential perpetrators of violence and the risk of harm to themselves.

"The mentally ill shouldn't be put into these situations in the first place. I feel sorry for them, they get picked on. They are vulnerable. I usually end up leaving them – I know what the system is like."

[Police Constable, Station B]

Officers saw the system as facilitating their contact with the mentally ill. The criminal justice process involved the mentally ill several times, with officers recognising the

occurrence of repeated episodes of criminality within this population. This contact continued until the desired recognition of the underlying mental health problem ha occurred. Officers argued that only after several encounters with the police was their mental illness recognised and dealt with. They stated that their aim within these situations was to react to and establish their safety and that of the individual. They would then try and use the system to achieve a result. The problem identified by these officers was the time consuming progression from the first meeting to getting through the system. These themes of time, effectiveness and bureaucracy were central throughout the police experience of the mentally ill. The police, therefore, faced similar problems regularly, in various differing contexts, with a group of individuals who 'just keep coming back'.

The officers interviewed all outlined these features. The persistent presence of a core group of mentally ill people within the police environment had consequences for their actions towards the mentally ill and the perception of them as a specific group.

Recognition of mental illness had associated characteristics and assumptions which the police utilised in their every day encounters.

To distinguish between types of call, the police often developed their own classification system. This outlined the types of calls, incidents and responses to a variety of situations. This system of differentiating their involvement was a useful mechanism in which to place the role and response of the police in their dealings with the mentally ill. In this particular study, the emphasis appeared to be on the order maintenance or peacekeeping role with an additional focus on citizen evoked police involvement.

4.5 The Reactive Nature of Police Involvement

Officers in both stations tended to agree that they often were called to incidents involving the mentally ill, rather than adopting a proactive approach. They recognised that with the all-encompassing feature of the role they could be called for any situation; such is the nature of being a police officer. In these situations, the police initially relied on an outsider interpretation of the event. This police involvement was confirmed by the frequent demand for police assistance as a result of citizen interpretation and decision-making.

Officers had little control over the type of incidents they attended. They believed they were the first response for many, rather than being seen as an agency of last resort.

Officers felt that people tended to call them for a variety of reasons, including trivial non-policing issues. A frequent utterance when discussing the nature of police work was the declaration that 'when something goes wrong, the police are called'.

Within the area of the police and the mentally ill, this feature was clearly evident to officers. As one officer stated,

"We get called when people don't know what to do, any time – neighbours and relatives don't know what to do with the mentally ill. We're supposed to know."

[Police Sergeant, Station B]

The most common request for police assistance resulted from relatives or neighbours of the mentally ill. The police felt that these individuals turn to them as an agency when they perceived the mentally ill as reaching a 'crisis point'. The timing of these encounters and demands reflected the unique position of the police in their twenty-four hour capacity. As several officers noted, the frequency of contact often occurred in the

evenings and at weekends, after the services normally associated with the mentally ill had closed.

The use of the police as a twenty-four hour service for numerous dilemmas was evident to officers. This included their inability to refuse help. This had clear implications for the police in this study. It was the authority and capability of the police that resulted in officers assuming an all-encompassing social and legal role. The police is this study confirmed these observations. They accepted that this situation will continued, as

"We shouldn't be involved but we have no choice, we're got a duty to protect, but is a police officer the right person to sit with the mentally ill for 6 hours. There are other people who need police help."

[Police Sergeant, Station B]

This highlights not only the dilemma of becoming involved in a social problem, but the practicalities for the police officer in the demand for action. As part of their role as social agents, expectations of action remained paramount, regardless of the time taken.

Judging the crisis level of individuals and situations was a difficult aspect of police work. It had its basis in the subjective awareness and judgements of individual officers. The Detective Constable interviewed, who had regular contact with the mentally ill in his previous role as a custody sergeant, stated his belief that the mentally ill were not necessarily at a crisis level when in contact with the police. He preferred to see them as distressed, often acting in a disturbed manner as a cry for help. The role of the officer in this situation was stated in the following manner,

"You have to figure out where they're coming from, put yourself in their place. Some officers haven't so much patience as others, they escalate things. You have to try and understand them."

[Detective Constable, Station A]

The holistic approach advocated, however, often proved time consuming. It was seen as a luxury when faced with the demands of the immediate situation.

On the official action and options available within these encounters, officers noted that they rarely were involved in sections, and particularly section 136. As one officer reflected, he had no personal experience of sectioning any individual, or in his words 'I've never had one that's been sectioned' [Police Constable, Station B]. He believed that this was due to the reluctance of doctors to section, predominantly due to a lack of resources. There was a shortage of beds for acute emergency admission within the local area, with the local accident and emergency hospital being used by the police as a facility to place the mentally ill. This facility was not secure, with the result being that officers had to stay with the individual until the situation had been resolved. There had been a local psychiatric unit where officers could take the mentally ill. This unit, however, had closed. It would appear that this local situation influenced the use of sections.

In conjunction with the views of other officers, it can be seen that sectioning individuals via the mental health legislation was an infrequent, if not rare, occurrence. Additionally, officers noted the rarity in which formal means of detaining and dealing with the mentally ill was activated. Why this happens may have been the result of a variety of factors, including bureaucracy and the time consuming nature of formal encounters, features which will be addressed later.

Overall, officers saw that the context in which they operated involved incidents that were primarily 'citizen evoked'. These involved a variety of contextual and situational factors. There was also a requirement for a variety of responses. The citizen evoked requests meet with a different response. This was particularly evident when compared

with those instigated by the police themselves. The police recognised that they had no choice in entering encounters with the mentally ill.

4.6 Violence within the Context of Unpredictability

Whilst officers recognised the difficulties and vulnerability of the mentally ill, they were also aware of the issue of violent behaviour. Throughout the interviews in both stations, officers referred to any contact with the mentally ill as being surrounded by assumptions of violence. It was natural for the officers assessing these areas to place violence and danger as central features of their work. This notion of danger remains a central aspect of the police role that it forms both an overt and covert cognitive framework in which to judge social interactions. In association with criminality, past violence and assumed further violence were key factors in determining action by the police.

The nature of violence and unpredictability in police encounters carried certain connotations when linked to the issue of mental illness. Violence was cited by the officers interviewed as a feature of their involvement with the mentally ill. Violent incidents were seen as typical.

"Violence is involved in a lot of cases- you don't know what you're facing."

[Police Constable, Station B]

"What we think of first is our own safety – violent incidents are typical – they [the mentally ill] are very unpredictable."

[Police Constable, Station B]

"We go round in circles – they are barking mad but they're aggressive – do we say they are mentally ill or not?"

[Police Constable, Station B]

These comments confirmed the police belief that their work typically escalates into violent situations. Inherent was the additional belief that police work per se was a

dangerous occupation, with violence a common feature. The commonality of violence as a distinguishing mark of police work was confirmed by the police officer's awareness of the potential for violence. This was linked to the police consciousness of the dangers of their 'beat' or area. For those who transcend the 'normal' range of police suspects and clients, additional emphasis was placed on their behaviour and the possibility of violence. Naturally incorporated within this particular classification were the mentally ill.

Of the mentally ill identified by the officers in Station B, one individual was known to be violent. This statement by one group of officers appeared to contradict the earlier assertion of the commonality of violence. The impact of these experiences for officers meant that the generalisation and incorporation of violence was a central feature in their understanding of the mentally ill. The predominance of stereotypical assumptions of mental illness and the unpredictable aspects of their behaviour translated, for police officers, into a preoccupation with the potential for violence.

Additional components for officers also included a duality of beliefs around the existence of actual violence and the feeling of its potentiality. For the officers in this study, the predominant concerns were factors such as the 'unknown'. This was seen as an unmanageable aspect of their involvement with the mentally ill. Its appropriateness as a description of the mentally ill for the officers was apparent when establishing their picture of the mentally ill. There was a need for the police to use this image when assessing and determining action. The result was a picture of the 'symbolic mentally ill'. This was a general description, which included the physical signs of mental illness and the collective beliefs of the officers interviewed. Central to this was the belief that the mentally ill had the ability to inflict violence and pose a threat.

The police learnt to distinguish who had the possibility to inflict disorder and violence through a process of cues, both verbal and non-verbal. Involved in this process was an awareness of the occurrence of 'out of the ordinary' encounters and 'odd' behaviours. In essence, it was those situations that signified unpredictability, in behaviour and in outcome. As one officer argued,

"There's potential for things to happen – it is a cause for concern. There is no system for assessment, it's the police who have to deal with criminality or mental illness and we need to intervene before something happens."

[Police Constable, Station B]

Enshrined in the working persona of the police officer was the need for predictable behaviour and outcome. The preference for predictability in the daily experience of the officer was another important feature highlighted. This idea of predictable situations and behaviour provided them with a guide. It enabled officers to identify methods of management. The need for predictability was extended to include the types and characteristics of those they encounter.

Equally relevant was the police role in potential situations,

"The mentally ill are unpredictable – we are there just in case. The diagnosis doesn't help – they're not sane, but we have to deal with the situation. You try and understand their mental illness, but they can inflict damage and violence."

[Police Sergeant, Station B]

The unpredictability of the behaviour as indicated by the police definition of mental illness and their experience of involvement with this group were prime considerations for all police officers. One example was given to highlight the reasons for establishing the link between violence and mental illness.

"One man – a 21 year old – had a history of violence. It once took eight officers to look after him. He had absconded from the local hostel after he had broken another patient's arm. He was basically scared. We had to talk to him for about one and half hours to calm the situation down by talking and understanding. They [mentally ill] are often distressed, and not necessarily at a crisis point."

[Police Constable, Station B]

This case also highlights the dilemma often faced by the police. While this individual was violent, he was also seen as vulnerable and distressed by the officers, who subsequently identified and used their skills to manage the situation without the use of force.

Whilst the risk of violence needs to be acknowledged, other influences are relevant and require examination. They involve individual perceptions and the impression of others towards the police role. The dilemmas for officers included the dichotomy evident when introducing notions of care, control or management in the interactions between the police and the mentally ill. Role ambiguity, therefore, became evident. This role confusion translated into many factors for the police officers, but notably it appears to them that

"We don't know what to do for the best – the mentally ill are vulnerable and they need help. It's easy to take them to hospital, but the problem is what happens after. We end up with them back."

[Police Constable, Station B]

We have to ask what do we expect the police to do in light of their concerns for the potential of violence and their knowledge of the criminal justice and mental health systems. Their roles and their tasks often became confusing.

One officer encapsulated this fear or wariness of mental illness when he stated that,

"I wouldn't like to work on a psychiatric ward. It was an eye opener. I'm more protected on the street than on a psychiatric ward."

[Police Constable, Station B]

It was important to consider when understanding these police beliefs the possibility of the police being called and present for the very reason they cite as a major concern; they were likely to be involved when an individual or professional held the view that violence would result. This argument meant that the chances of officers encountering violence were inevitable.

There was an element of discrimination by some of the officers in their categorisation of the mentally ill. There was an important distinction between those the police knew and those where little information was obtained. Besides distinguishing between the eccentrics, the violent and the more quiet individuals, the police acknowledged their treatment of these groups differed according to how well they knew the individual. The more they encountered them, the more informal the police response would be.

However, the police continued to see the mentally ill within two extremes; those who were meek and quiet and those that had a reputation for violence.

Yet, while they might view the local characters as eccentrics and harmless, the officers were aware of the potential for violence. It was the unpredictability of this group that has to be placed in the context of police involvement. This was a group that officers in Station B felt had the potential to be violent. One case given as justification for this view concerned an individual who was displaying a knife and gun, which resulted in the armed response unit of the force being called. They later found that the items were plastic. The association of violence and mental illness meant that the police entered

such encounters with caution. They anticipated that the usual signals that they perceive as indicators for trouble were not apparent in people with mental illness.

However, the more knowledge they developed concerning individuals, the more likely they were to treat them informally. One police constable [Station B] went further and stated that for those individuals he had knowledge of, and previous encounters with, he was aware of the course of action to take. For him the dilemmas arose with those he did not know. The amplification process of initial encounters diminished with more police involvement. Included within this analysis by officers was the establishment of predictability. This led to a lessening of the sanctions imposed, in terms of formality and severity. Officers stated that it also diminished the use of alternative actions, which included the use of force.

The police talk of these individuals known to officers contained a mixture of humour, concern, frustration and some intolerance. The familiarity of some and the unfamiliarity of certain aspects of the group altered the methods employed to manage the situation. The officers worked within their own definition of what constituted mental illness, making assumptions regarding the frequency and type of events. In the initial police awareness of mental illness, the police were at the beginning a course of events and actions that presented officers not only with administrative problems, but also social dilemmas.

4. 7 The 'Immediate Situation' and the Safety of Self

The police assessment of violence and risk and their knowledge of certain individuals all contribute to the decision making process of officers. The police belief in dangerousness and violence translated further into an awareness of the importance of maintaining their safety in any encounter. The rationale for such procedure recognised

the police belief in the unpredictability of situations. It can be conceptualised within the following statement, 'we don't know what we are walking into' [Police Constable, Station B]. Assessing the situation was considered a key factor. This was primarily due to the constant and realistic belief that they were 'going in cold'.

Surrounding this premise was a combination of practical factors and considerations.

• the amount of information received

This involved

- the police knowledge and familiarity of the individual
- the social composition of the situational.

It was the 'immediate situation' that officers had to deal with. This involved an understanding of the potential risks.

This issue was a constant feature throughout this research and one that highlights the police reaction to the mentally ill and the social context of the situation. The basic principles of policing were highlighted within this discussion of priorities. The officers in both stations maintained that the protection of life and property were core to the definition of the police role and function. As with other situations the police faced, their safety was considered paramount. As they noted, how can the public be helped when the officer her/himself is in danger.

All of the officers recognised that each incident had to be assessed on an individual basis. Additionally,

"Our assessment of a situation is based purely on what we see or what we believed happened. We tend to know people but not the background. We need the background to know what we are dealing with."

[Police Constable, Station B]

When faced with competing influences and factors, the officer had to assess which of these demands took precedence. In this sense they considered that they were containing the encounter within an present timeframe. This was in order to continue to assess the possible alternative scenarios that may develop.

While core areas were highlighted and themes generalised, officers expressed an overt understanding of the individuality of cases. Decisions were made in specific areas.

These included the judgements around cases and the police assumption of seriousness, methods of managing an incident and its relationship to other cases faced by officers.

Within this process was the insistence of the possible role of the first case as a precursor for future action.

The police decision-making process in assessing situations involved not only the issue of violence and risk, but also the anticipated outcome of the encounter. The practical expectation of how the police resolved the situation involved knowledge of the possible alternatives available. The type and amount of services available to manage and treat a particular group impacted on the decisions of the officers and the action taken. The social dynamics of the police with the medical and social care sector additionally influenced their consideration of priorities and behaviours.

"We deal with other agencies' need for assessments. It isn't the police job to assess for mental illness. But then no attention is paid to our assessment. Psychiatrists don't listen."

[Police Constable, Station B]

What the police encountered was a demand for their presence in the face of conflicting organisational and agency needs. In their overall assessment of the situation and the resulting decisions concerning action, there was a requirement that judgements were made upon the resources available and the likelihood of accessing these resources.

These issues regarding the context of police operations will be discussed later.

However, it is sufficient to state that in the immediate situational assessment, the variety of factors to consider in a given incident incorporate the wider question of the final destination of the individual.

Generally, the officers interviewed all recognised the need to establish the parameter of their safety prior to commencing the interactive process. They also acknowledged the need to follow some form of guidelines. Often these were informal in nature. This system and style of working incorporated individual and group methods of dealing with the mentally ill. Such a process involved not only developing certain techniques for management, but also a reliance on established police practice. This included a controversial feature, notably the use of force.

4.8 The Use of Force and the Mentally III

The police concept of force, as highlighted by Hunt [1985], involved a process of allocating notions of 'normality' to police practice in their every day encounters. The definition of force implied a coercive and physical restriction. The aims of force include the containment and management of a difficult situation. Normal force was indicated within a continuum of definitions, including legality. The centrality of force, actual and as a symbolic tool, for the police role and authority influenced their behaviour and actions towards individuals. The officers in Station B felt that the ability to use force provided them with a 'mental back up' when entering situations.

The power and authority of the police in their use of force posed many questions.

However, one has to remember the police definitions of force and the possibility of differences in emphasis when compared with concepts outside of the police. These were arbitrary terms and open to subjective interpretation, dependent on officers'

individual perceptions and the police services' collective mentality. The officers who formed the basis for this research demonstrated their own understanding of the issue of force.

The situational element involved in the use of force was recognisable within the context of police encounters generally and their interaction with the mentally ill specifically. Primarily, it was the need for force that arose out of the demands for control of a situation. Officers seen the use of force as a measure to re-instigate order and ultimately control. As with many powers of the police, the interpretation required meant that it was individual officers who determined the meaning of force, its appropriateness and its application. There were difficulties and potential conflicts for officers when deciding upon and using force.

The added dilemmas for this group of officers when dealing with the mentally ill were seen as the following:

- the collective assumption of the vulnerability of this group
- the presence of outsiders who may condemn the police for their management methods
- the public image of the police in these situations.

For the police officers themselves, the demand for the immediate containment of potentially violent individuals can limit the options available to them. The consideration of force became a significant priority.

The issue and determination of force was raised by several officers in these interviews. It centred mainly on the assessment of force and its appropriate use. It was considered a factor to assess alongside the maintenance of safe parameters. For these officers, force took on numerous guises. However, one option became a focus for examination.

As a feature of modern policing, the use of CS gas was seen as an effective tool of police practice. The extension of this to include its use on the mentally ill was recognised as a viable possibility by officers in this research. They saw CS gas generally as a welcome addition to their methods of managing potentially difficult circumstances.

Two groups of officers in Station B made specific reference to the use of CS gas on the mentally ill. When discussing its use on this particular group of individuals, officers acknowledged its usage, especially in incidents involving violent mentally ill individuals. The distinction between the different types of mentally ill individuals was placed within the police equation in determining the use of CS gas. While some required assistance through conversation and de-escalation, the police felt that others' behaviour, actual or perceived, sanctioned the use of force.

The apparent proactive use of force, rather than its use as a defensive tool, signalled a departure in policing methods generally [Corry, 1999]. With specific reference to the mentally ill, it suggested a differing meaning attached to these encounters. This placed emphasis on control in the likelihood of violence or risk to self. In the police meanings associated with mental illness, officers were developing methods to establish control, prior to further management decisions. The use of force was seen as a primary step in the officers' individual organization of the situation. However, one needs to question how this use of force influenced the interaction process of the encounter and the possible outcomes of the situation.

To officers the potential use of force represented an additional method of dealing with the mentally ill. They argued that it allowed them to take control of the situation and consequently assess the next stage of the encounter. The foreseen violence, real or anticipated, was dealt with by this method. However, such action carries implications.

One dilemma for the police concerned the legal implications of its use. The officers argued that difficulties would occur if CS gas was used on a mentally ill person and they were not subsequently arrested. The physiological consequences of CS gas on the mentally ill and the interaction with medication were not predominant issues for the police officers. For these officers, the situation and the risk of violence meant that such issues were not addressed. However, the medical consequences of the use of CS gas may not have been known to officers at the time of interview.

The appropriateness of this action has been questioned nationally since these interviews [Kliman, 1998]. The outcome of this recognition was the call for the a set of guidelines for officers in the operation of the technique. However, the Home Office perceived the issue to be primarily one concerning operational matters and one predominantly centred on self-defence. The additional issue of the effect of CS gas on those taking medication has resulted in questions over its appropriateness and the potential risk to the mentally ill [Police Review, July 1999].

The use of force on the mentally ill was an area of both concern and debate, legally and morally. The extent and application of force by the police, and their use of CS gas, altered the contextual factors within their involvement with the mentally ill. The association of mental illness, risk and violence may be too strong that the interpretation placed by officers on the encounter provided them with justification for its use. However, one should note that reference to its use was specifically made by two groups of officers only, and while it may be seen as a established practice, others may seek out initial alternatives to the use of force.

4.8 The Welfare Role of the Police

It was evident that the police had to deal with both aspects of care and control when dealing the mentally ill in the community. Throughout the interviews the police expressed concern regarding not only the vulnerability of the mentally ill, but also the inadequacies of the system. While the issue of risk has increased the necessity of the police to take into account such factors as the possibility of violence, the officers also had to deal with individuals who were distressed and unwell. For the police this conflict produced a variety of dilemmas.

"At times, you end up acting like an advocate – looking after their interests. But that isn't our role."

[Police Constable, Station B]

Officers perceived some individuals as frequently in need of assistance. One officer highlighted one specific case. This highlighted, for him, typical situations facing officers.

"One lad who self harms and has a history of mental illness often calls the police, but its often attention seeking. These people have slipped through the net."

[Detective Constable, Station A]

Common to the criticisms of the mental health system, officers also questioned the existence of community. Demonstrating sympathy for the plight of the mentally ill individual, they commented on the vulnerability of the group and the lack of understanding of mental illness within the community itself. The police continued to occupy a difficult role in their role of ensuring welfare for the mentally ill.

Officers were aware of an increase in the numbers of mentally ill in the community.

They considered this to be the result of community care policy. Others, however, were more ambivalent as to the increase in numbers. This may have been the result of the informality of the process and the lack of statistical evidence. Explanations as to

why there was perceived increase often lead to the conclusion that there was a correlation between police involvement and deinstitutionalisation. This was also seen as the consequence of a corresponding lack of community resources.

The notion of community care was treated with both dissatisfaction and disbelief.

Comments ranged from questioning the existence of any sense of community to the change in society that has resulted in a loss of a 'personal sense of identity'. Overall, the situation of the mentally ill in the community was seen as one of deprivation.

"The mentally ill shouldn't be put into places like this. The community is a jungle. There's no such thing as community."

[Police Sergeant, Station B]

"The community is picking on the mentally ill. It isn't fair to the mentally ill, they can't look after themselves, they are very vulnerable, and are persecuted by the youths. They give trust too easily, like the elderly"

[Police Constable, Station B]

"The problem is the poor placement of the mentally ill. They are persecuted. The better areas don't want them. People don't care and don't want the mentally ill in the community. Where do you put them?"

[Police Constable, Station B]

The officers seen the mentally ill as being 'targeted' by the community. They commented that the mentally ill were often placed in the most deprived areas. They were predominantly seen as the victims of crime themselves, rather than the perpetrators. The vulnerability of the group meant that officers questioned the problems facing the individuals, arguing that the mentally ill should not have to face these situations in the first instance. This included living in a community that does not want them. From this perspective, the main problem for the officers involved managing

the mentally ill within a climate of risk, whilst also seeing them as a disadvantaged group.

The social problems of mental illness, and its association with violence and unpredictability, present a confused and complex problem to officers. Their awareness of the social context of such encounters presents officers with a dichotomy of responses, criminality versus sympathy.

It is the management of the situation and the difficulties for both themselves as police officers and the mentally ill individual which concerns officers. The causation notion that community care was primarily responsible for these predicaments provided officers with a theory as to the possible need for their involvement with the mentally ill. It offered a cause and effect explanation, when in fact many situations and circumstances were often far more complex and multi layered.

4.9 Conclusion

There are several issues relevant for the police in their contact with the mentally ill.

Officers rely on their own methods to detect and assess mental illness. They cannot refuse to help individuals. They also have to deal with a variety of professionals and 'outsiders', whilst maintaining the police image. A major area for the police, however, was managing the conflict that arises when confronted with risk and vulnerability. The police have to operate within a climate of fear towards mental illness. They also have limited options available for dealing with the situation.

The methods of management, discussed in the next chapter, offered a variety of strategies for the police. These can be seen as predominantly informal. These police

techniques provided the officers with a way of managing the complexities and difficulties evident when encountering the mentally ill.

Chapter Five: Findings - The Informal Management of the Mentally Ill

5.1 Introduction

At the start of this thesis a question was raised concerning the influence of the current mental health policy on police contact with the mentally ill. While figures are hard to ascertain, what appeared evident was an increased awareness of the mentally ill in the community. The next issues posed for the police were how they approached and managed this group.

The police considered various approaches and styles when they became involved with the mentally ill. These were considered alongside an acknowledgement of the problems of managing both care and control. What characterised the approach of the police was the more informal management of the mentally ill within their communities. This involved the use of established police practices, existing sanctions and measures and effective interpersonal skills. While there were similarities in the approach taken, one exception concerned the experiences of the community beat officer in station A.

5.2 The Exception to the Rule - The Experiences of the Community Beat Officer

The reactive nature of policing and the diversity of situations encountered was a point that occurred throughout the groups of officers interviewed, but with one exception. The role of the Community Beat Officer highlighted a more proactive style of policing. This approach had implications regarding the move towards community partnerships and the shift in the philosophical underpinning of policing within his local force.

The community beat officer had been in post for three and a half years, after gaining over twenty years as an officer elsewhere in the force. This officer saw the context of his work as being focused more on the dilemmas experienced by the community. These mainly related to the concerns of the local residents to criminal matters. This was the starting point for his involvement in the local community.

The central role of the Community Beat Officer has been highlighted by Mastrofski et al [1995] and Vinzant and Crothers [1994]. There appears to be a reliance on the discretion of the officer. This approach also recognises the potential for the officer to direct and inform the community of solutions to identified problems. Negotiation, consideration and consultation replace the emphasis on immediate police action.

The background to the notion of community policing emphasised, in turn, the potential for a deviation in the type of interaction and experience for the community officer in their encounters with the mentally ill. The different emphasis in the work of the community beat sergeant involved integration with the community, attending local meetings and being the focal point for concerns. This further translated into defining mental health problems within this remit. He saw this issue as one of the various dilemmas evident within the community. For this officer, the presence of mentally ill individuals was not a predominant concern. He felt that this was due to the specific nature and feature of his role. The officer argued that

"People are not necessarily at crisis point, some have nervous breakdowns but how do you cope with that – I can't really get involved"

[Community Beat Sergeant, Station A]

The community beat sergeant recognised that there were limitations to the help he could provide. He tended to respond to particular problems identified by the community, rather than deal on a day-to-day basis with the unpredictability of police work. The most likely way to become involved with the mentally ill would be if the community identified a specific problem and defined the individual as mentally ill. One example was cited by the

officer of a woman who was experiencing apparent mental health problems, culminating in her refusal to let anyone enter her house. In this particular case, the course of action taken by the police, in the guise of the community beat officer, was seen as a proactive approach. The officer contacted other agencies, including social services and local doctors, in addition to family members. Here, the community beat sergeant made the referrals and resolved the situation. His direct involvement with the mentally ill was, therefore, in response to the demands of the community. He argued that this took both time and a thorough knowledge of the local community.

The officer cited the advantages of this type of policing. These included a thorough knowledge of the area and its residents and the time and ability to build up a picture of the situation. He did not see the various factors in isolation. For those officers who predominantly work 'on the beat', they did not have the luxury of developing a system whereby often time consuming solutions can be sought. It was apparent that they dealt with situations as they arose, drawing on learned skills through experience, a common feature of police work. They, therefore, had to rely on other factors and other roles in their involvement with the mentally ill.

For the community beat officer, the existence of the mentally ill in the community was one of a series of dilemmas and problems, and one where he utilised his knowledge of the area and the residents to deal with the situation.

5.3 The Use of the Police Methods of Judgement

The experience of police officers helps to develop their skills. Through this experience they develop a form of routine. This involves knowing what they need to do once attending an incident. The police skills identified included the ability to respond quickly, to assess the situation effectively and to act accordingly.

This process was a common features of police work in relation to the majority of groups the police encountered. For their involvement with the mentally ill, officers felt that the usual rules that helped them decide on what action to take were not necessarily obvious. Additional cues and factors had to be considered. The police relied on external features to seek confirmation of the mental state of the individual. This in turn influenced the decision making process they then employed.

The usual police training provided officers with a basis for assessing the 'defensive signs of attack' and the likely triggers for violence [Police Sergeant, Station B]. Officers felt that the mentally ill 'jump' this process, thereby obscuring any assessment of the situation. As a consequence, the police believed that they needed to be aware of the 'strength' of the mentally ill. By this term officers equated the 'strength' of the mentally ill to their physical power. Officers also linked this their experience of the unpredictability of the mentally ill.

The frequent reference to this unpredictability influenced the police understanding of mental illness. Understanding the full extent and type of mental illness experienced by the individual, while seen as beneficial for the police, were not the factors that were used. Officers had an awareness of the 'out of the ordinary' event. What they then relied on were two factors identified as core policing skills: common sense and intuition.

5.4 The Issues of Common Sense and Intuition

It appeared that in understanding the mental state of an individual the police combined numerous aspects:

- their own experience of police-public relationships
- their awareness of the dynamics of a situation
- the construction of assumptions and assessments.

When describing their method and approach in assessing mental illness, officers placed their reaction within a context of common sense and intuition.

It was a frequent comment from the officers in station A that they could judge if someone was mentally ill by instinct or through common sense. The issue of common sense featured in their subsequent approach to situations. Officers argued that the process of police work enabled them to determine circumstances quickly and to assess when something was not quite right. For one officer, situations were placed in context by using an internalised scale to determine what was occurring, whilst also considering the grounds for possible arrest [Custody Sergeant, Station A]. He argued that events could be determined on a scale of one to five. These provided him with a quick and effective way of judging situations. Within this, he compared other criminal incidents with the behaviour of the mentally ill. He argued that the mentally ill behaved due to their illness, rather than any criminal intent.

The skills and requirements of the police in their every day encounters were various and often included contradictory elements, recognising the conflict between care and control. In referring to their method of policing as the use of common sense, they identified a wide range of possible styles and approaches. The use of common sense arose in the majority of situations faced by the officers. The officers could not determine what they faced; the availability of common sense and intuition as a basis for judgement became useful. It provided the foundation for action, as the first step in the decision making process,

"We're got a common sense approach – we get no training in mental illness."

[Police Constable, Station B]

"You learn on the job- I picked things up as I go through. I see police work as doing the shift and surviving, then going home"

[Custody Sergeant, Station A]

"You're got to think on your feet. Training in mental illness?— next time we're called to an incident - you learn on the job"

[Police Constable, Station B,]

Officers saw this reliance on common sense as enabling them to assess situations and make decisions quickly. This also illustrated the importance of 'surviving' the job and using the skills developed to aid this process [Detective Constable, Station A]. This was connected to the notion of learning through experience, the ability to judge the aims of the encounter and to examine a variety of solutions.

When assessing the presence of mental illness and judging the behaviour of the individual, the officers faced potential difficulties in making sense of the situation. As some officers pointed out, a contradiction for them was the extent that mental illness contributed to the actions of the individual. This was particularly evident when the individual was breaking the law. If they were violent and mentally ill, the officers tended to see their mental illness as causing this behaviour. These mentally ill individuals, therefore, were viewed differently. They were not classified the same as those who committed criminal acts and, therefore, were not categorised as criminal. Rather, officers adopted a different labelling approach, stigmatising their perceived strange behaviour as mad rather than bad. The aim for officers was to place them within the remit of another social system [i.e. within the realms of mental health care]. The police intuition informed them that the place for the mentally ill was not in the criminal justice system.

The use of common sense and the police skill of intuition was summarised as a heightened awareness of events and situations. This was evident not only in terms of

their immediate response, but also in the subsequent methods used, particularly with regard to those unknown to the police. Officers confirmed the earlier finding of their different treatment of those already known to officers, providing previous reference to the individual's behaviour and the previous police response. The notion of familiarity was relevant when encountering a recurring situation.

5.5 The Police as Professional Persuaders

There were a variety of approaches available to the police. One that became useful for these officers was the their role as 'professional persuaders'. There was one aspect that the police identified that was effective in the management of the mentally,

"We get called upon to persuade people to go back to hospital—if someone else did this job, they would detain them."

[Custody Sergeant, Station B]

The officers saw the main use of this approach as two fold; first dealing with those who have absconded and the immediate situation that was created. Secondly, it was used as their overall approach to the mentally ill, persuading them to comply through the use of informal rather than formal measures. In the specific area of the absconded patient/resident, officers identified several key factors.

While some officers saw this as a relatively simple procedure of returning the individual to the hospital/hostel, another group elaborated on this particular approach. As they commented, the police were often required to advise and influence the individual in deciding to return to hospital. The officers perceived this as a recurring and frequent problem, with the wider issue being that while they assisted in 'putting people away', they later found that they had returned to the community.

The use of their skills as 'professional persuaders' recognised the fact that the police often faced the same situation time after time. They, therefore, developed techniques in managing and resolving the situation. For these officers, this repetition was a frustrating element of their work. They argued that they were dealing predominantly with short-term solutions to often long term problems.

The benefits of 'persuading' people to comply, rather than enacting formal measures, were numerous for the police. They viewed this in terms of both bureaucracy and outcomes associated with dealing with the mentally ill. In adopting an approach of non-coercive compliance, they aimed to resolve the situation with the minimum disruption. By enacting the role of 'professional persuaders', the officers saw that they were limiting the workload faced in situations of non compliance and the subsequent dangers they envisaged as a result of the use of coercion.

5.6 'Others Around' - The Observational Effect

The social composition of the police encounter with the mentally ill did not necessarily encompass only officers and the mentally ill themselves. Often the presence of others impacted on how the police interacted and were perceived. The officers demonstrated an awareness of their presence upon the mentally ill. However, outsider involvement within the social process, even limited to a form of observation, effected the police decision-making procedure. Themes that emerged cover the variety of incidents the police experienced; in the hospital, at the scene of an incident, and at the police station. Initially, it was the reaction displayed by the mentally ill upon the arrival of the police that they highlighted.

Officers recognised their impact on the mentally ill. Two distinctions emerged, those who appeared to welcome the police, and situations where the attendance of officers

escalated the situation. Several officers noted how in certain incidences they had the potential to make matters worse. An officer in station A similarly saw that their presence could increase the tendency of a violent confrontation. Conversely, other officers recognised the positive reaction often displayed, especially by relatives, to their presence. They describe their arrival as providing the observers with a sense of relief, as they 'know the police will do something'.

Officers also recognised that at times both types of situations exist; the presence of the police uniform as a signal of action in the form of help and the possibility of a adverse reaction on the part of the mentally ill to the same police uniform. The police uniform was the most visible component that suggested a role designation, with certain assumptions being made by outsiders as to the characteristics of the incumbent. The social significance of the uniform, in conveying a message and intention, effected both the observer and the wearer of the uniform. Primarily, it communicated a sense of legitimacy and an expectation that the wearer of the uniform would not only manage the situation in hand, but also do so in a manner that encompasses the principles of the organisation represented. For the officers this involved the ability to use both coercive and non threatening methods of containment.

The officers felt that the situation was complicated by the understanding of the mentally ill of the significance and the meaning of the uniform and the presence of the police. Equally relevant in such cases was the police understanding of the ability of the mentally ill to comprehend the situation and the consequences of actions. This additionally applied to the understanding of outsiders to the police involvement with the mentally ill and the appropriateness of such encounters.

The adverse reaction to their presence and their authority involved a particular expectation and reaction from the police officers involved. The chances of the mentally ill going 'ballistic' or 'mad' resulted in officers having to deal with people and the situation cautiously. The officers were concerned of entering situations 'cold' and not knowing what to expect. Not knowing the initial reaction or the likely responses presented difficulties in determining the manner and subsequent stance to take.

The impact of the police on others within the encounter mirrored similar aspects of the initial reaction of the mentally ill themselves. Officers from both stations generally expressed an opinion on the impact of their involvement with the mentally ill and their 'image' as police officers and law enforcers. The key factors in their description of the external features of their interaction with the mentally ill can be distinguished as follows.

The first feature highlighted by the police was that when called to an incident, those people observing the event appeared to be surprised by the presence of the police. The officers noted that it was their involvement with the mentally ill which surprised the observers. As a consequence of this, their presence was then questioned, which officers maintained then had an impact on the police image. This subsequently curtailed the officers in any action considered, such as the use of force.

Additionally, officers described the potential for seeing their practice and actions as an example of police brutality. These concerns of officers included being aware of the opinion of the outsider and observer to police behaviour in a hospital department. This was particularly if they were restraining individuals. They asked,

"How does it look in casualty when two officers are struggling with a patient."

[Police Constable, Station B]

The officers acknowledged the discomfort felt generally in this type of action. The issues highlighted included the possible perception of outsiders to this behaviour, especially when surrounded by individuals experiencing different and mainly physical ailments. The problem for the police was two fold; the manifestation and maintenance of an outward display of control and the management of a mentally ill individual in a socio-medical care setting. The issue of image and the picture of police behaviour to outsiders had further repercussions in different settings.

Surrounding this context, with its various meanings, influences and behaviours was a preoccupation with maintaining a positive police image. Whether in the community or at the police station, officers noted the need to be conscious of the perception of others and the interpretation placed upon police behaviour. The police felt that certain expectations existed. These included that the belief that officers would not only deal with the situation via action, as opposed to non-action, but that this action was beyond criticism [Detective Constable, Station A]. This officer argued that their approach and manner had to be the same for all individuals. This included the demonstration of respect, regardless of their opinion of those they were dealing with. Ultimately, officers were aware that "you have to treat everyone with the same level of respect" [Police Constable, Station B,].

The apparent contradictory nature of police action in treating individuals the same, while assessing situations individually was translated into the skill base of officers themselves. Officers interviewed reiterated their approach towards the mentally ill and the predominance of common sense as a useful tool. Bayley and Bittner [1984] noted the notion of 'presence' as a feature of policing which incorporated not only the skills and schemas involved in being a police officer, but also the outward appearance of authority.

These notions can be linked to the demonstration of respect towards the public and the need for externalised calm in maintaining the situation.

A group that often played a pivotal role in the police-mentally ill encounter, if mainly on the boundaries, were the family. The family frequently and coincidentally initiated police contact in the first instance. When examining the interaction and the influence of the family in such surroundings, the officers felt that the family generally took a 'narrower' perspective when compared with the police. They assimilated information in the present and questioned the police decisions and action. The officers, however, believed that they had to take a wider view of the situation, incorporating their information and understanding from past experience.

The distinction between the view of the police and those of the observers, including the relatives, was an influential factor in the immediate interactive situation. Equally relevant was the distinction between those the police engage with generally, in terms of class and status. Officers distinguished between certain sections of society and their response to the police. This, in turn, effected the police response to particular groups. These officers noted that the most of the local community were familiar with the police. They were aware of the police presence, and as one officer commented,

"The barriers have been broken down. ...the area is well policed, they know us, but in the suburb areas they are not use to us, they treat us differently".

[Police Constable, Station B]

Equally, the police in making this distinction, stated that they reacted in a different manner, when compared with the encounters in their more familiar area of practice.

Officers felt not only restricted or effected by those observing the social encounter, but also those supervising, albeit from a distance, their practice. An additional expectation

that impacted on the behaviour of the officers was the anticipated reaction of the custody sergeant to the decision of the officer in individual situations. The officers in Station B acknowledged the possible effect of the custody sergeant on their decision to bring a mentally ill individual to the station.

Previous experience had indicated to the officers that the negativity demonstrated by the sergeant resulted in the removal of the individual from the station and the start of the usual formal procedure [involving the use of the local hospital]. This had implications for officers in terms of bureaucracy and the outcome of the incident.

"When we arrest someone, the custody sergeant tells us to take them to hospital. The doctor then says they are not mentally ill and doesn't section. The problem for us is what to do then. Often we don't charge as its not in the public interest and the case would be dropped by the CPS [Crown Prosecution Service]."

[Police Constable, Station B]

The role of their supervisor/superior in their interpretation of the situation restricted these officers and their ability to choose. This was in light of organisational demands and the perspective of the custody sergeant overriding the need of the officer to end the encounter.

The role of the outsider to the process involving the police and the mentally ill, whether a member of the community or police managers, played an important part in the dynamics of the social situation. When considering options and determining what choice to make, the police officers considered not only public perception of their action, but the views and opinions of their supervisor. This process, however, was complex and often involved a myriad of considerations and alternatives.

5. 6' What to Do with the Mentally Ill' - The Final Destination?

After examining the initial reaction and the various considerations and influences that occur, the final aim for the police was the resolution of the situation and the end of the encounter. The central concern for the police was deciding an option which took the least time and presented them with the least trouble. The decision making process included numerous practical considerations, such as the powers to use that would ultimately bring the matter to an end. However, the preoccupation with reaching an outcome transcended the appropriateness of some of the decisions taken.

Encompassed within this was the use of the breach of the peace as a common legal device. It was a response to the particular circumstances within the police-mentally ill encounters.

"We have to decide what to do –often it's just a sideways move – we deal with people, either by getting them to calm down, then leave them or use breach of the peace, bring them to the station and then use the mental health act."

[Police Constable, Station B]

Using a legal sanction on a social problem demonstrated the difficulties for the officers in managing the mentally ill. Officers rationalised the use of breach of the peace as one possible method of coping with the mentally ill. A reason given for its use was that it avoided the use of mental health sections. The power was a means to an end. It was seen as a way of avoiding more problematic choices, rather than as a solution to the particular situation encountered.

This reluctance to section was connected with common aspects associated with the officers' involvement with the mentally ill. These included:

- the time taken to enact a section
- the bureaucracy involved in the process
- often difficult relationships with other professionals.

All officers saw the use of hospital as a last resort. Several other stages were considered first. If the police skill as professional persuaders was not effective, the influence of relatives in the process was minimal, or ultimately, the situation was deemed serious with no viable alternative, then transportation to hospital occurred. For this to become a method of managing the mentally ill, several alternatives had to be dismissed or viewed as impractical by the police. All officers had reservations as to the use of hospitalisation. These were in terms of the likely rejection of individuals due to the lack of resources, the added dilemma of bureaucracy with the average time for officers remaining in hospital being four hours. There was also added dilemmas regarding the general expectations of the police when dealing with other professionals.

For officers, one of their main problems in the use of the hospital was the sharing of information between the police and the hospital staff and the resulting potential for conflict.

"The main problem is lack of communication. The hospital staff only see us as bringing people in and leaving them. They don't think we can make decisions, or listen to our opinions."

[Custody Sergeant, Station B]

This lack of multi agency co-operation frustrated the police. They felt that they were required to manage a difficult situation, only to be limited by the actions of other professionals. The limited understanding of the roles of staff involved in the mental health process compounded the situation. This was translated into unwillingness on the

part of officers to become involved in the process, with the added dilemma being the failure of other professionals to acknowledge their wider ability and role.

As a result, there was a reliance on the more informal means of management. Where possible, the use of informal measures was seen as a more effective method of dealing with the mentally ill, via the persuasion to comply or the involvement of significant others to manage the individual. Other methods took precedence, using the skills of the police to draw the situation to a close,

"You have to be as daft as them —it's a way of claming the situation down. Acting like them is the police way of bringing people down."

[Custody sergeant, Station A]

"With behaviour that's not normal, we go and look, but if they're harmless, not a danger, we just leave them" [Police Constable, Station B]

"Often the mentally ill need someone to talk to. If I've got the time then I will speak to them. I fob them off, though, if there's no time"

[Police Constable, Station A]

"It's a sticky plaster reaction. You have to look after your back. It's easier to do breach of the peace."

[Police Constable, Station B]

The informal versus the formal approach to the mentally ill has been well documented [Green ,1997 and Teplin et al ,1992]. Little appears to have changed. These officers, as criminal justice agents operating in the social context of mental illness, considered non-legal issues and informal methods as key determinants for action.

In their practice, officers felt that such issues as the timing of the incident in the shift were given prominence. If at the end of a shift, the action taken will take a different form with the early and quickest intervention being pursued. An example of this action was the police role of 'professional persuaders', when they attempted to talk and encourage

individuals into voluntary admission. Equally, the style of policing, while within police perception of potential violence, exhibited less informal characteristics.

The manner in which the police interacted with the mentally ill was characterised by the use of humour. This was seen as a management method and as a de-escalation tool.

Again, knowledge of the individual and the situation influenced the type of action taken.

Referring frequently to 'humouring the mentally ill', the officers saw this as a skill that not only calmed the individual down by acknowledging their interpretation of events and action, but also acted as a method of containment and control prior to the next stage of action.

In terms of action and the solutions available, there was little the police could do for the mentally ill,

"It can take a full shift [to deal with the mentally ill].
You feel like you're done nothing, just hanging around.
It can be stressful – someone might kick off"

[Police Constable, Station B]

"If someone is mentally ill, you have in mind that you'll be stuck for hours 'just in case'. I wouldn't mind if I knew that something would happen, but it doesn't. I'm obliged to stay"

[Police Constable, Station B]

The frustration of officers was evident. They had to rely on their skills as 'jack of all trades' being partly social workers and partly law enforcers. While there was some awareness of the need for care in addition to control, the prime concern in the situation was for the officer to take charge. How this was achieved depended on the following:

- timing of the incident
- space [i.e. in the public domain with observers to the situation]
- the demands of the police organisation

- the knowledge of the officer of the community and of the individual
- the expectation of violence.

Whether it was informal or formal in terms of style, for the officers a quick solution was the main concern. They recognised their persona as a 'taxi service'. While they accepted this, it did not detract from the realisation that the actions they took often were a 'sideways move'. This involved either attempting to calm people down in order to leave the situation, or using a breach of the peace before transferring them to the station.

Overall, the solutions employed by the police were short term. This police expectation of their role centred on them as 'first aiders'. They supplied a 'sticking plaster' approach, knowing that the mentally ill would return to the community to again become involved with the police. Hence the continuous cycle for the police-mentally ill encounters. It was the combination of the police awareness of risk in their interactions generally, their need to care for the mentally ill, their expectation of unpredictability and the socio-medical framework of mental illness that brought forward the various issues for the police as social agents.

5.8 Police Dilemmas

Having recognised the transitory and repeating nature of their encounters, the officers exhibited an awareness of the dilemmas posed within these specific interactions. The strong emphasis placed on the inadequacies of the system by all officers pointed to the continued use of the police as a safety net for the mentally ill. However, further dilemmas and perceptions arose which illustrated the police concept of working with a group in need of special treatment. The implications of dealing with this group were both social and practical, suggesting various resource implications.

The perceptions of the police to their management of the mentally ill can be divided into several sections and key themes:

- The practicality of interacting with and managing this group [identifying mental illness and managing the various risks involved].
- The police as social commentators, providing insight into the causation of the situations they attend.
- Their decision making conundrum and the presence of 'first aid policing'.
- The question of the appropriateness of involvement.

These areas indicated the frustration perceived by the officers in this particular aspect of their work. The central premise behind these assumptions can be defined in one simple question, often stated by the officers themselves, 'what can they do?'

5.9 Practical Dilemmas to Complex Problems

Policing the mentally ill appeared to involve a limited array of practical options.

Analysing the notion of discretion implied that the police had a list of possible alternatives. In reality, this list was restricted to what was practical and what was legal. In theory the diversity of police work and the availability of powers to officers suggested that options were numerous. In reality, the individual was curtailed by both the legality and appropriateness of the proposed solutions. Besides the use of breach of the peace, the officers believed that there was little formally that they could do. In terms of their discretionary power, the issue arose regarding the justification of their actions.

The complexity of the social encounter was further demonstrated by the police ability in assessing the mental state of the individual, or 'working them out'. What the police did, therefore, was to question whether they could actually state that someone was mentally ill. Some officers felt that it was appropriate to limit the use of this label on an individual.

By doing this they recognised that they were ending their involvement and delaying the entry of the individual into the system.

As stated earlier, the police adopted a general, yet seemingly useful, method of diagnosis.

The consequences for the police centred on events after the assessment had been made.

The limited communication between the various agencies involved compounded the situation faced by the police. As one officer commented,

"The most we can do is find some reason to take them to the station or the hospital and deal with them that way." [Police Constable, Station B]

The development of skills to manage the mentally ill within the community was a result of 'doing the job'. The training available to officers was limited to introductory sessions concerning the basic characteristics of certain mental illness and the use of police powers. Similarly for officers in Station A, their training was considered basic and rudimentary. What they learnt about mental illness and the individuals in their community occurred whilst attending an incident. This incorporated their ability to 'think on their feet'. This learning on the job was often associated with police work generally. The officers felt that it can be extended to include their interactions with the mentally ill.

5.10 The Decision Making Conundrum - The 'First Aid' Approach to Policing

What summarised the police decision making conundrum was the belief in a 'Catch 22' situation - whatever way they turned, they faced dilemmas, difficulties and the consequences of their actions. If the police did nothing, then the mentally ill 'go on to do what they intended'. If individuals then injured themselves or others, the potential for a major incident increased. As a result, the officers faced prolonged involvement. Dealing with the 'here and now' and determining the actions that would result in swift

termination was a significant proportion of police work. The officers, however, did acknowledge that

"We need to find a 'quick fix' Often it's not in the best interests of the individuals."

[Police Constable, Station B].

To arrest or not to arrest was a major dilemma for the officers. The informal methods highlighted previously offered some solution. However, the initial reaction of what needed to be done, as opposed to what could be, centred on the use of existing police powers. The specific intervention for the police was one which emphasised the need for treatment, rather than the imposition of legal sanctions. To arrest then became a legal solution to a socio-medical problem.

Their actions had not only to be justified, but also accounted for. When questioned as to what they should do, several appeared reluctant to argue the case for police involvement with the mentally ill. Officers did not believe that they should assess individuals. They felt restricted by their powers, referring specifically to those incidents that took place in public. When the arena was moved to that of the private house, breach of the peace was the favoured response. However, there was a contradiction between the experiences and perceptions of officers in the various groups, with others seeing the dilemmas of dealing with the mentally ill in private as a major issue. The use [or non-use] of powers, however, brought further legal difficulties for the police. The police often found that they were 'bending the law to get the job done' [Police Constable, Station B].

The transitory nature of these encounters suggested that this approach was effective in dealing with often crisis situations. However, it also brought frustration. For several officers in Station A, once the situation had been dealt with, they often did not know the long term outcome, unless the individual became involved with the police again. The

limited information available to the police as to the fate of individuals was one aspect of police work that suggested that complete resolution was restricted.

To state that these dilemmas were restricted to the behaviour of the mentally ill and the police response would be to fail to recognise the need for immediate action when dealing with other groups in society [e.g., juveniles, domestic violence incidents, etc]. The issues that carry significance for their encounters with this particular group included their management of issues that outwardly required more specialised knowledge, dealing with a group of individuals who were perceived as both vulnerable and violent, and the recurring emphasis on extralegal factors. As they argued, no amount of training could assist the police in their encounters with the mentally ill,

"What would training do – the system is crap, community care doesn't work and the hospital is overrun. Our duty should end at the hospital but there's no secure unit, so we have to stay."

[Police Constable, Station B]

5.11 The Suggested Solutions of the Police

The common solutions put forward reflected the similar problems experienced by the officers. The suggestions covered practical and resource issues, including more and better facilities. These proposals from officers included the introduction of early intervention measures and more secure facilities.

The central requirements for police officers in their involvement with the mentally ill can be seen as containing two policy features; the facilities for placement of the mentally ill and more sharing of information. One officer recognised the type of information needed.

"What we need is a register. There are about thirty or forty places in the community, but we don't know who lives there. We need to know. This area is popular for learning disability and mental illness homes."

[Police Sergeant, Station B]

These recommendations appeared to solve the dilemmas of placement and information gathering. They aimed to reduce the unpredictability in interactions and outcomes. What the police were arguing for was the establishment of a 'safety net for officers'. They went on to state that this would offer clear guidelines, limit the use of their discretionary powers and provide boundaries for police practice.

The solutions offered can be seen as being beneficial for the police working within the area of mental illness. Officers called for the following:

- the introduction of assessment centres
- a place to admit the mentally ill in order to provide them with treatment
- strategies aimed at releasing officers to fulfil their more 'law enforcing role' in the community
- the introduction of medication monitoring.

Officers perceived that these areas would influence police action. This is particularly the case regarding issue of medication compliance. Officers believed that once individuals failed to take medication, the likelihood of police contact with the mentally ill increased.

The issue of increased powers also occurred. This included the ability to commit those deemed as violent. Officers also called for a revision of the Mental Health Act. These policy suggestions have become more relevant since the introduction of the White Paper in December 2000, and its proposals for the reform of the 1983 Mental Health Act. In the meantime, what the police argued for was the effective use of existing legislation and powers.

Diversion schemes were highlighted as a possible solution to the problems faced by officers. The introduction of these schemes varied from force to force. There was also evidence that they varied within forces. For some officers, the introduction of a diversion

scheme offered a reference point for advice and information. However, for the group of officers who were involved in a local diversion scheme, the bureaucracy and procedures outlined meant that they all were reluctant to use these schemes [Station A]. For these officers, the schemes were seen in the same light as the formal measures. This questions the distinctions between the organisational response of the service and the individual actions and requirements of the officers [an outline of the force's policy is provided in chapter six].

The solutions of the police highlighted the problems of working with a system that contains expectations of resources and of management. This demonstrates the differences between the organisational demands and the practicalities for individual officers. The solutions additionally indicated the idealisation of increased resources and facilities. However, this should not detract from the features of the police involvement with the mentally ill. Regardless of resources, police-mentally ill interactions and encounters will continue to be a feature of police work. The situational characteristics of that first initial contact will remain. This is in conjunction with the desire of the police for quick and effective outcomes.

5.12 Conclusion

The interactions and the management of the mentally ill highlighted by these officers involved the negotiation of unpredictability. This was on several levels; in situations, interactions, reactions and expectations. How they managed this was via the reliance on and use of the established means of policing. This was in addition to an awareness of the specific dynamics of each situation. It was a heightened sense of awareness that signified the police-mentally ill situation, from the first call for assistance, the perception of 'going in cold' through to the decision making process. The factors involved in the complexity

of encounters were incorporated along a spectrum of experience, past and present, and an expectation of future dilemmas arising from this type of involvement.

The key themes when discussing the police involvement with the mentally ill were common to policing generally. These included the issue of violence and unpredictability, the establishment of authority through the use of force and the need for priorities contained with the demand of the immediate situation, with the added dilemma of bureaucracy. However, for the specific involvement with the mentally ill, other extra considerations carry meaning for the officer, including the issues around care and control, the level of sympathy displayed, the ensuing frustration in the restriction of options and the appropriateness of their involvement. This is within the context of an agency that cannot refuse such involvement.

However, the police operate within a wider organisational context. As officers developed methods of managing the mentally ill, the police force have also examined their response to this issue. Questions arise as to the extent the practice of officers and the expectations of the police force converge.

Chapter Six: Findings - The Service Response to the Mentally Ill

6.1 Introduction

Police officers face the dilemmas posed by the mentally ill in the community on a variety of levels, including the managing the immediate situation and deciding on the action to take. The police organisation is also required, as a public service, to develop a strategy to deal with particular problems and groups. The strategy and the development of schemes aims to ensure a consistent approach across the police divisions and departments.

As with other identified groups within the community, the response to the management of the mentally ill in the community has involved both national and local initiatives.

The police management, with the objectives laid out by government agencies and with the expectations of the Chief Constables, aims to transfer these guidelines and directives downwards to operational police. One additional consequence of this is an attempt at controlling the actions of the officer in the community.

Wide deviations occur. This is due to a variety of factors, including local resources and individual practice. Involved in this process are the individual personalities connected with the development of schemes. This means that the organisational guidelines and response experienced by officers are dependent, to some extent, on where they are stationed. The nature of police decision-making on an organisational level has implications for the actions of the police officers within the community. Involved in this process of examining the organisation's demand for consistency are the police managers' expectations of their officers.

This chapter deals with the responses of five police forces to the mentally ill in the community. What is encompassed within this is the hierarchical structure of the police

and their system of filtering guidelines and objectives to operational officers. The outcome of such schemes and policy initiatives, however, may take a different development when considering the power and autonomy of the officer on the beat.

6.2 Specific Schemes and Specific Roles - The Official Response

During the course of this research, several police forces where approached and personnel interviewed who were either designated with the responsibility of dealing with the mentally ill or were involved with schemes aimed at managing this group. The police forces concerned covered several parts of the country and illustrated the diverse nature of the schemes and their specific answer to the mentally ill in the community.

The police forces examined included the following;

- a police force area in inner city in the South of England [police force 1]
- a police force in the Midlands area that covered a wide region [police force 2]
- two forces in the North West of England [police force 3 and police force 5]
- a force in a more rural area in the south of England [police force 4].

Two of these forces also allowed direct access to the operational police officers interviewed in this research [police forces 3 and 5].

The interviews were conducted with six managerial officers. An examination of relevant documentation also occurred. The managerial officers came from a variety of police forces and were involved at various levels in the formulation and implementation of policy for managing mental illness. Five of the officers were at the level of Inspector, with one Chief Inspector participating in the interviews.

These police forces offered a differing level of service for the mentally ill. Any disparity can be seen on a continuum, with one force providing no specialist procedure

or service, while several forces attempted a more co-ordinated approach. Overall, these forces operated a variety of schemes at the various stages of the criminal justice process. Within the development of each response, one has to remember the political climate influencing the police approach to the management of the mentally ill. Each scheme and procedure contributes to the individual officers' practice towards mentally ill individuals. It highlights the level of responsibility placed on police and other agencies in providing an solutions to the specific dilemmas manifested within society.

This police force faced several issues relating to the care and control of the mentally ill. Generally, the policy and practice of this force can be pinpointed to the publication and recommendations of the Clunis Report in 1994 [Ritchie et al, 1994]. This police force received harsh criticism for their management of the mentally ill and the lack of co-

6.3. Police Force 1 - The Formation of a Specialist Officer and a Formal Scheme

within the force of the numbers of mentally ill individuals involved with their officers.

ordination of policy. This is in conjunction with an increase in a general awareness

The high profile given to the issues raised within the inquiry report provided a mechanism for agencies, such as the police, to re-evaluate their actions towards, and interactions with, the mentally ill. For one station, in particular, the interactions they perceived and the contact they had with the mentally ill occurred on a daily basis. However, no official figures were available to confirm this belief. It was maintained that the police contact continued to be informal in nature and could not be formally and systematically monitored and analysed.

The Inspector interviewed at this police station outlined the major areas of development in the management of the mentally ill. This included his role as the Mental Health Liaison officer, with responsibility for implementing the policy of the force at a local level. Throughout the interview several practical and pragmatic issues were highlighted.

The practical answers to incidents involving the mentally ill recognised by the Inspector included a formalised procedure for section 136, the use of the powers of arrest, and their involvement as protection for other professionals when sectioning individuals under the Mental Health Act. The Inspector estimated that they were involved in between three and four section 136 referrals per week. For this station and force specifically and its significant ethnic minority population, there was a concern regarding section 136 and the sizeable number of ethnic minorities who were detained under the section. The possible reasons considered by the Inspector for this occurrence included the difficulties many experienced in accessing psychiatric services and care. However, the role of discrimination may have contributed to the significant numbers of ethnic minorities involved.

6.4 The Organisational Response

From the point of view of the Inspector, the organisational answer to the management of the mentally ill involved three main approaches:

- the employment of specialist mental health liaison officers in each police division.
- the joint formalised procedure for section 136.
- the training of the operational officers force wide.

These themes covered a wide range of issues for the police, they involved an awareness of the main features of both the criminal justice and mental health systems, notably multi agency working. Additionally, a recognised individual, co-ordinating mental

illness policy was a feature which the Inspector seen as aiding the implementation of a consistent approach to the mentally ill.

Similarly, shared guidelines were established to increase accountability, permitting each agency an understanding of the expectations and responsibility of the other professionals involved. The issue of training highlighted in the government circulars and various reports recognised that the police in particular were no more expert in determining the mental state of individuals than other non specialist groups. These combined factors, on an ideological level at least, attempted to deal with the main organisational dilemmas faced by the police. Benefits identified from this approach included the development of skills and professional boundaries.

This police force installed mental health liaison officers at each of its 62 divisions. The Inspector cited the advantages in naming and identifying an individual officer.

Primarily, he saw this approach as narrowing the possibility of a disorganised, unstructured and informal management strategy. With organising the majority of the policy and practice through one senior member, the operational officers had a clear line of hierarchy and command concerning one area of practice. It also became apparent to those involved in the process that this development assisted in the practice of interagency working, by providing an identifiable person to provide a contact point.

The post of mental health liaison officer was one that was fulfilled by officers of the rank of Inspector. It was essentially viewed as a part time post. This occurrence was additionally seen as being mainly determined by resource issues. However, as the Inspector acknowledged, it would be a worrying factor if there developed a need for a full time mental health liaison within each police division.

In March 1995, the police force with the local psychiatric services documented arrangements for good practice with regard to the use of section 136, the place of safety procedure. It highlighted the sequence of events leading from the initial contact through to the assessment of patients. In conjunction with the social services department, the local procedure outlined the responsibility and level involvement of the various agencies. For the police, and their predominant place at the beginning of the procedure, the local policy provided officers with points of communication and outlined the steps to be taken.

The procedure essentially removed or curtailed the discretion of the officer, after the preliminary determination of mental illness. It ensured that officers adhered to a set of standard instructions as to what action to take, including the stage at which psychiatrists and approved social workers became involved. The procedure additionally implemented a form of joint monitoring, using a police form [Form 434] outlining the key factors in the occurrence of section 136. The monitoring form required the police officer to outline the circumstances of the detention, including the behaviour of the individual and the intended place of safety.

As a document outlining the process of good practice, the section 136 procedure established a time frame in which decisions and actions were sought. In practice, the Inspector recognised that the bureaucracy involved, while minimised, meant that officers were involved in the process for approximately two to three hours, rather than the expected half an hour. However, while good practice was attempted, the formalised procedure noted that, in reality, this would be difficult. It also recognised the key features of accepted practice, acknowledging the resource problems implicit in the assessment of the mentally ill. The notion of an appropriate place of safety was

endorsed, with the anticipation that a police station would only be used in certain, exceptional circumstances.

6.5 Issues in Joint Procedures

However, the Inspector highlighted the fact that several factors had to be considered to provide a context for the practical reality of this area of joint working. He felt that the police were often viewed with suspicion and mistrust by other services and individuals within the community. This effected not only their relationship with local agencies, but also the reaction of individuals to the police, including those with mental illness. At the time of interview, the Inspector considered that the situation was improving, with mutual trust and reliance developing. This was partly as a response to the perceived success of the section 136 procedure. As this area of practice was being implemented, more informal arrangements of communication were being utilised. The Inspector commented that if he suspected an individual was mentally ill, he could contact the local community mental health team and suggest that they undertake an assessment informally, rather than formally approach services.

Several issues occurred as a result of the development in this informal communication system. These included the overall ability of the police to detect mental illness, the confidence of others in the police diagnosis and the rights of the individual and their confidentiality. From an organisational point of view, such informal systems provided the police with certain benefits. The emergence of informal networks gave the officers an opportunity to extend their role in the maintenance of social order by becoming involved, albeit superficially, in the treatment of this perceived difficult group of individuals.

One further possible consequence of this network was the possible additional methods of obtaining information regarding certain individuals in the community. This method of gaining information was a key feature that emerged from the Inspector's perspective of the police encounters with the mentally ill. It was also seen by him as a key requirement for working with this group. Additionally it had relevance for how the police operated in society generally, through the amassing of information and knowledge of their community. The possibility of using other agencies for gaining information was seen as a consequence, potentially both negative and positive, of increased interagency co-ordination.

6.6 The Training of Police Officers

This force responded to the growing numbers of mentally ill people coming into contact with their officers by implementing a system of training. This training identified the core skills needed. These included the detection of illness and developing awareness of the available strategies for the police when dealing with the mentally ill. This training was recognised as an objective at a wider policy level within the force. It became an essential training strategy for a significant proportion of the force's officers. The training package offered consisted of a distance learning pack, which included a booklet and video.

The Inspector commented that this use of this style of teaching had obvious benefits and deficits. It enabled officers to assimilate the information at their own pace and in their own time. As the authors of the document additionally pointed out, it would also act as a reference manual and a source of information for any further dilemmas encountered by officers. However, one of the deficits highlighted by the Inspector was that, as the trainers and the police superiors cannot necessarily monitor the actual

training, its implementation and appliance may vary. One concern expressed was that that, while officers have to be trained in a variety of areas and develop various skills, there was a danger that the issue of mental illness was marginalized. It was argued that officers may not receive the benefit of an increased knowledge base.

The main characteristics of the training package consisted of the following areas:

- an outline of the facts surrounding mental illness.
- the relevant context of community care and the increasing likelihood of police encounters with the mentally ill.
- information on local schemes and arrangements.
- the legislation considered important to the police.
- the usual methods deployed when facing someone who is mentally ill, including an awareness of non-verbal behaviour.

While covering a wide range of areas and topics, the nature and style of the training pack meant that officers were provided with a basic understanding and awareness of the necessary information, skills and knowledge pertinent to interactions with the mentally ill.

One issue highlighted by the Inspector was that of violence and mental illness. The issue of violence has been outlined previously by the findings in chapter four. The training provided by this force reiterated this and outlined the possibility of such a situation being faced by officers. Such factors as sudden physical movements, language and tone of voice were incorporated into this description of key factors. Equally important was the need for the police to inform the mentally ill individual of the process and actions of the officer.

The document illustrated the importance of the police in minimising the potential for aggression and violence. For operational officers, the perceived dangers of their work and the unpredictability of the dynamics and the outcomes of the situation were features that were not only stressed within the organisational culture, but also recognised by as inevitable part of all police work. The training package appeared, therefore, to draw on the skills of the police officer already in existence, while also recognising the need for a greater understanding of the position and experience of the mentally ill.

It was acknowledged by the Inspector that the effectiveness of such training schemes was difficult to judge and dependent, to some extent, on the willingness of individual officers to incorporate the key themes of the training into their practice. The distance-learning module provided guidance for police officers rather than a formal outline of instructions. As in other areas of police work, individual determinants of action still took precedence over the organisational expectations and methods. However, in conjunction with the other arrangements in place, the police management of the mentally ill within this specific locality provided a variety of avenues and methods for operational officers to deploy.

6.7. The Development of Mental Health Policy of the Police

It appeared to the Inspector that the various divisions within the police force largely determined their own managerial strategies. The police as a force developed various initiatives centrally, which complemented the work of the individual areas. These included an analysis by the police and the Royal College of Psychiatrists of the use of section 136 [1996/97]. The police, the local ambulance service and social services additionally examined the specific area of the assessment of the mentally ill in private premises. Overall, this police force were continuously examining their response to the

mentally ill and assessing new methods and approaches, within a multi disciplinary framework.

The examination of the service for the mentally ill, amongst other particular groups, occurred in the background of the Macpherson Report into the treatment of the Stephen Lawrence. This report severely criticised the police in several areas, including the existence of institutional racism. The pressure on the police to develop an effective response to key groups and areas within the communities continued to be seen as a necessity. This aimed to rebuild public confidence in the police as a service and ensure effective methods of working within the community. This wider political context cannot be ignored when looking at the police in their many guises and in their encounters with many groups. It has implications for the police, organisationally and individually.

This particularly police force adopted many key areas of policy development, in light of governmental objectives and organisational experience. Several factors contributed to these developments, including the demographic features of the force's region, individual interests and the demand for a more co-ordinated approach. Similarly, other police forces have responded to problems identified in their local communities.

6.8 Police Force 2 - A Multi Layered and Multi Disciplinary Response

This police force covered a large geographical area and faced similar issues as highlighted by Police Force 1. Equally, the force demonstrated a comparable response to the occurrence of mentally ill people in the community. The need for the various policy developments was reinforced by the increasing numbers of the mentally ill who were coming into contact with the police service.

The Inspector interviewed had a particularly interest in the area of mental illness and took the lead in the development of force policy. He highlighted the incidence of police encounters with the mentally ill and viewed this as a characteristic of modern policing. A local study carried out in the latter half of 1993 found that 207 people were detained using section 136 [Roughton, 1994]. The predominance of the black and ethnic community in these figures raised many questions. The figures stated that 13.5 per cent of those detained under section 136 were from black or ethnic minority origin, although they only represented 3.6 per cent of the local population.

For Inspector, the main issues centred on developing strategies to enable officers within their area to effectively deal with the problems associated with the mentally ill. The police force responded to this dilemma in several key areas, notably with the development of diversion schemes within their police stations, involving the expertise of other professionals.

6.9 The Development of Local Diversion Schemes

Within the police force, there emerged a variety of schemes specifically aimed at managing the mentally ill within the community. These included

- a diversion at the point of arrest scheme at certain police stations
- court diversion schemes
- nursing services at bail hostels
- a mental health assessment procedure at a local prison, which was the subject of research by the NACRO [1993a].

The multi-layered response to the experience of the mentally ill in the criminal justice arena encompassed the numerous stages of the system. In essence, the Inspector felt

that the presence of mental illness should be detected, if not in the beginning at the police station, then at the next point on the continuum.

The co-ordination and existence of such schemes, in response to the local experience faced by the mentally ill, occurred within a background of multi agency working and partnerships. At the time of interview, the Inspector considered that the police and other local services generally provided an encompassing and multi faceted answer to those deemed mentally ill. The strategy, additionally, incorporated the management of the mentally ill at the first point of contact. For the Inspector, the key initiative centred on the employment of a diversion scheme at the time of arrest, at the very beginning of the criminal justice system.

The police area incorporated twenty-one divisions or operational command units. This police force in the early 1990s recognised the necessity of the conditions and objectives outlined by both the Reed Report [1992] and the Home Office [1995]. The briefing document produced by a local forensic unit and the police force additionally remarked on further issues that influenced and generated the requirement of co-ordinated services for the mentally ill. These included issues around community safety.

In response to these considerations, the pilot project of a diversion at the point of arrest scheme [DAPA] began in 1992. It was initially based at what was considered their busiest police station, in the south of the city. As result of the success of the scheme, three additional stations in the area benefited from the project. The development of such an initiative involved several processes, including the establishment of a multi agency steering group. As commented by the Inspector, providing a forum for the various agencies, such as social services, police and psychiatry, was an accomplishment. Establishing procedure and consensus from this arena was a further example of the

importance and relevance for a systematic and structured method of managing and caring for the mentally ill.

The pilot project began with two community psychiatric nurses from the forensic unit being placed within the police station. This provided an assessment service for those arrested by the police, regardless of the police diagnosis of mental state. The Inspector noted that the use of DAPA aimed to provide them with an efficient and effective option for dealing with the mentally ill. The scheme further acknowledged the negative aspects of detaining a mentally ill person for any considerable period of time in a police station. The availability of specific mental health knowledge meant that the police role in diagnosing mental illness was minimised. This was in association with a reduction in bureaucratic responsibilities.

However, the difficulties of placing one professional with occupational beliefs, working practices and methods into a different and potentially opposing ideology had implications for the success of any joint working. Overcoming these presumptions involved a variety of approaches. These highlighted the necessity of developing organisational interaction and understanding; from an informal level with the initial sharing of information to the more structured and formally acknowledged emergence of joint schemes, such as DAPA.

The dilemmas posed by the formulation of diversion schemes had further implications for the potential outcome of the project. As a result, the nurses who became involved within the scheme were allowed time to understand the nature of police practice and procedure, in addition to forming relationships with the officers at the station. The Inspector recognised the importance of this approach and acknowledged that it aided the development not only of partnerships, but also joint working practices and styles.

The operational procedure of DAPA involved the nurse assessing those in the custody suite by a variety of means, including observation and standardised documentation. The station installed a special room within the station in order for the assessment interview to take place. This assessment occurred after discussions between the arresting officer, the custody sergeant and the nurse. This was the initial stage of the diversion process. It took into account such factors as the presence of substance abuse and the severity of the mental illness. Following a more formal interview, the procedure examined the availability of health and social care services.

6.10 Practical Issues and the Maintenance of Schemes

Several issues were prominent for the Inspector when discussing the issue of diversion.

The concerns raised by the use of cautioning for young offenders were applied to the mentally ill. These included:

- the involvement of the mentally ill, even superficially, with the criminal justice system
- the issue of the police gate keeping the decision making process
- the appropriateness of the disposals considered.

As with most social policies, a balance of the interests of the client and the organisation was difficult to maintain. However the ideological basis of diversion maintained that the removal of the mentally ill out of the criminal justice system not only served the needs of officers, but also the mentally ill themselves.

During the first year of the scheme, the total number of people interviewed by the community psychiatric nurse was seven hundred and thirty five, or twenty per cent of those arrested by the police. Of these seven per cent, or fifty one individuals, were diagnosed as mentally ill. Thirty three of the mentally ill suspects were not charged.

Instead they were diverted to more suitable facilities. The offences reiterated several known facts about the offending behaviours of individuals with mental illness, with offences being classified as minor. One crucial feature of those in contact with the police was their previous psychiatric history. Of the fifty one people seen as mentally ill, thirty two had prior involvement with mental health services. Therefore, the police were not encountering 'new' groups of mentally ill people.

As noted earlier, the scheme evolved further after the initial pilot project was established in one police station. Other stations additionally had the availability of an on call service. Essentially this response recognised the cost effectiveness of instilling a service which met the demands of the police and encouraged its availability only when needed. The 'on call' service received a steady rate of police referrals between the years 1993/4 to 1995/96. In the final year the scheme seen one hundred and thirty two referrals, which showed a slight decrease. Of these, one hundred and eight had a diagnosis of mental illness. The Inspector commented that the nature and number of the referrals demonstrated their ability to provide a clear diagnosis in their initial assessment of the individual's mental state.

The Inspector felt that the use of DAPA schemes offered a less bureaucratic response to the experience of the police officers in dealing with the mentally ill. One of the additional outcomes of the project was the recognition of the need for such a service [Wix, 1994]. As a result of the success of the scheme further policy objectives emerged, particularly the introduction of training. The scheme formed a central component of the overall answer to the problems experienced by both services and the mentally ill. It also encouraged the policy to be implemented on a national basis.

The policy of the police as outlined above terminated in April, 1997. The police response concentrated on the development of a scheme that covered the city, both North and South, and included the use of local community psychiatric nurses. An evaluation of this change in the direction needed to take place to fully understand the impact of this response on operational officers. Further communication with the Inspector, however, pointed to a reduction in the desire for diversion as a policy direction. The future of such schemes was in doubt, both from an ideological and resource viewpoint. While it demonstrated the possibilities of a formalised and coherent response, additional policy and service implications took precedence over specific, but useful, schemes.

What such schemes as DAPA show are the possibilities of developing a response to a complex social problem. The apparent success of the DAPA scheme, through the number of referrals received and the non-custodial disposals offered, suggested that the operational officers complied with the strategy instigated and organised by their superiors. However, issues around the transfer of policy into practice were felt to be a constant consideration when analysing the effectiveness of any initiative. The implications in ensuring policy administration on the streets pointed to a possible blurring of managerial objectives by street level implementers. These issues, and the role of discretion in the police officers' decision-making procedure, should be borne in mind when considering the relative success of projects.

6.11 Police Force 3 - The Variant Nature of Diversion

This police force in the Northwest has dealt with the recognition and emergence of the problems associated with the mentally ill in the community in the same manner as both

the previous two police forces. However, as with any policy solution, the organisation faced unique characteristics, and while the foundation and basis of such developments emerged from the same source, the final product of such decision-making was different.

The initial approach of this force encompassed both training and diversion. However, the implementation of the overall policy direction varied throughout the police area, with no two police stations experiencing similar initiatives. This was the result of numerous factors, such as the decision by managers to target the scheme in specific areas. The presumed response from the management included the following aims and illustrates their understanding of the dilemmas facing operational officers.

The Chief Inspector responsible for developing objectives in the police organisation claimed that, in 1991/92, they were one of the first forces to implement a policy regarding mentally disordered offenders. There were two components involved in these directives; diversion and training through distance learning. The training instigated by the police demonstrated similarities with that compiled by the police force 1. In addition to covering the main characteristics of mental illness, it highlighted the roles of other professions and the process of diversion.

The need for training was highlighted in a study of forty-eight officers, obtained through a local police station. The study showed that the officers, while having a basic understanding of their powers, procedures and the appropriateness of police involvement, required further information and knowledge. This was in conjunction with wider developments in inter agency co-ordination. This initiative was seen as one method of minimising the negative relationships between professions that some officers identified. A positive growth in professional understanding was a consequence of developing a policy of diversion.

The diversion schemes developed were similar to those identified within police force 2, such as DAPA schemes concentrating on diverting the mentally ill upon arrest. These schemes were situated in certain stations in the city centre and across the more rural areas. The Chief Inspector noted that the use of these schemes predominantly emphasised those who had committed an offence. For those who had broken the law, the distinction between options appeared simple and clear. The choice for officers was either diversion or prosecution. However, the Chief Inspector additionally recognised the significant number who come into contact with the police through non-criminal matters.

For the group of non offending mentally ill individuals, the Chief Inspector saw the only possible solution as the use of section 136 and the subsequent placement in hospital. The prevalent dilemmas, which centred on assessing the most appropriate course of action to take, were minimised. Here, the Chief Inspector felt that the officers in the community did not exercise any discretion.

"We don't leave the mentally ill alone. Police officers don't have any discretion – they either have to charge them or take them to hospital under section 136 – it's as simple as that"

[Chief Inspector, Police Force 3]

According to this officer, therefore, the choices open to operational officers involved the use of hospitalisation or an introduction to the two systems of criminal justice or health, via diversion.

The Chief Inspector envisaged that the officer in the community would not decide what action to take using his/he discretion. Through this policy, the skills and decision-making process in their every day practice of policing were distracted in favour of a guided and structured response. Or as the Chief Inspector stated himself, 'there is no

conflict for the police in fulfilling their roles – organisational demands take precedence over individual interpretation'. The reality of this, however, was debatable. It was dependent on the culture, procedure and practice within individual police units and the successful filtration of these managerial policy objectives to the staff in the community.

When questioned about the potential confusion that might arise between the issuing of instructions and the practical dilemmas created by dealing with the mentally ill in the community, the Chief Inspector held the belief that there was no conflict for the police in fulfilling their roles. The general belief that the officers experiencing and working with diversion schemes both welcomed and benefited from such approaches was reiterated. Emphasis was placed on the development of clear lines of responsibility as indicated in the formal schemes. Equally important for the movement between policy and practice was the assumption, by the Chief Inspector, that the operational police in their force had accepted diversion as a feature of their practice.

6.12 Policy Through to Practice – The Experience of Two Stations in One Force To highlight further the practical elements and characteristics of force policy, it is useful to examine two police stations within the area and their experience of the mentally ill. The police force was divided into seven districts. The two stations, both situated in areas experiencing social deprivation, had differing experiences of training and diversion. It is evident that the variance in the policy and practice of the mentally disordered offenders initiatives was not merely on a national basis, but also on a regional level. The operational officers who participated in this research were stationed

The officers at this station [Station B] in the north of the area covered a beat of six square miles. The activity of the police was mainly conducted through patrols of the

at one of these locations [Station B].

central area. The environment was predominantly housing estates, with a small town centre. The police at this station additionally were aware of the rising numbers of community homes for those with mental health problems. As previous chapters note, these developments carried implications for their involvement with the mentally ill.

The officers were divided into five different shifts. Between six or eight officers were allocated to each unit, who worked together on a regular basis. The station operated a car patrol system, for as one officer noted, they 'cannot afford the luxury of allowing officers to go on the beat'. Another feature of the station was that it no longer housed a custody suite. Offenders were taken to a police station, situated elsewhere in the area, which had purpose built facilities. These features of the station have implications for the officers in their management of specific groups, including the mentally ill.

This station, at the time of interview, did not operate a diversion scheme. This was an obvious consequence of the removal of their custody suite. The options available to officers centred on the use of existing facilities elsewhere in the region. When faced with a mentally ill person, they could transport them to the local hospital, although no set procedure, as demonstrated in police force 1, existed. Additionally, no formal arrangements for inter agency working were in place. The training the police received with regard to mental health issues mainly occurred during their initial period as a cadet. While officers were aware of the distance learning training pack, it was not necessarily available or used. However, personnel from the central diversion scheme in the city were beginning to inform officers of their powers and provide them with accessible tools, such as a card highlighting the details and conditions of section 136. In the widest sense, therefore, officers were required to make their own judgements as to the decisions to make and the action to take.

This illustrated that, while the police as a force advocated certain and specific approaches, for the individual stations and officers the practicalities of managing this group tended to be based on their knowledge of the area. As the officers' perceptions of their work with the mentally ill highlighted, they faced a certain amount of bureaucracy and frustration as to what could be achieved. The comparison of two stations within one police area illustrates the potential effect of diversion and formal arrangements on the work of the officer.

The experiences of the second police station police station contrasted with the lack of a formal approach outlined above. Subsequently, the Inspector at this station reiterated the need for ensuring, where practicable, that policy was considered throughout the force, rather than in certain locations.

This police station was a relatively busy station. For example, in one month between January and February 1997, one thousand prisoners were processed. This particular station identified various methods for dealing with local individuals they seen as experiencing mental illness. Through an arrangement with a local hospital, the development of an appropriate adult scheme and through the use of diversion, the police had mechanisms by which to structure their interactions with the mentally ill. By using a form detailing the implementation of section 136, the officers were also able to monitor the numbers referred.

In 1996, twenty-two people were sectioned. A significant proportion were dealt with either in the community, or through informal/voluntary admission. The numbers appeared to be relatively small. With the establishment of a more formal procedure by arrangement with the hospital, officers tended to spend on average over two hours dealing with this type of situation. This can be compared with the experience of the

officers in the first police station who often anticipated a considerable amount of time, sometimes up to five hours, dealing with the mentally ill within the hospital setting. The Inspector, however, acknowledged that generally section 136 was rarely used. Diversion at the station also involved of the use of a local community nurse, who attended the police station when required. This nurse also provided training to officers, in addition to the force's own training package. The Inspector believed that the officers at the station were more aware and able to detect mental illness as a result.

As with police force 2, the Inspector was involved in various steering groups and committees, ensuring that the police view was put forward and policy formation took account of their opinions and experiences. This formulation of policies also extended to the police assisting in certain elements of the Mental Health Act 1995, with specific reference to the powers to take and convey patients. This was a recurring problem and concern for the Inspector in this station, and arguably throughout the force. This was reiterated by other managerial officers within the force,

"Ideally we shouldn't be involved with the mentally ill, but it is a fact that we are facing more and more mentally ill now."

[Chief Inspector, Police Force 3]

It was a constant concern for the Inspector that while they acknowledged the social problem of the mentally ill in the community, their individual desire to become involved was minimal. Incorporated further within this process were certain political implications. As in many areas concerning the crossover of social problems, and the interlocking of systems, lines of responsibility and accountability appeared to be dominant issues. The Inspector argued that in this area there were 'clear lines of responsibility', mainly due to the existence of such schemes. As the Inspector

acknowledged the formal processes minimised the previous use of the police as an added resource, both as a taxi service and in providing a place of safety.

6.13 The Community Approach

The formal system of diversion accounted for a proportion of the interactions of the mentally ill with the police. It also provided a semblance of organisation and control of police actions. Through the use of a community beat officer system, the Inspector felt that the officers had additional informal contact with the mentally ill. By attending community meetings, officers became aware of mentally ill individuals in the community who were causing concern. The Inspector argued that with the existence of a community approach, problems presented and experienced by the mentally ill individual were managed.

Furthermore, since the interview with the Inspector, crisis intervention teams were developed. He recognised that this had advantages for officers. For while they were

"..in a predicament in looking after people, and not being able to arrest them – we can spend hours looking after people – now with CITs (Crisis Intervention Teams) people can get treatment earlier" [Inspector – Mental Health Liaison Officer, 5 June 1997]

This practice can be seen in light of the intentions of the force management to move towards problem orientated policing throughout the area. The development of community policing in the local area had further consequences for the mentally ill and the management of this group by the police and other agencies. Initial reactions suggested a change in the direction of policing.

On a practical level, it was estimated that fifty-nine officers in a district were to be designated with the remit of managing and dealing with local issues via community policing. It was envisaged that issues regarding mental illness would fall under the

remit of the community officers. It was anticipated that such an approach would fundamentally change the current response of the police towards the perceived special groups within society. As yet it is too soon to fully comprehend this change of policing style and the implications for individuals and officers alike. However, various implications can be sought and provide an theoretical explanation for police-mentally ill interactions in the future.

6.14 Police Force 4 - The Informal Response to a Limited Problem

The experience of the police force 4 was an example of the different levels of emphasis placed on mental illness by different police forces. At the time of interview, the Inspector acknowledged that the police force were at an early stage of policy development. Their overall policy direction and implementation were perceived as both ad hoc and reliant on existing legislation and powers to formulate a response to the mentally ill in their community. This was a reflection of the relatively small numbers of mentally ill people coming into contact with the police. In one area in 1996, of the 8000 arrests, only twenty were classified as section 136. The Inspector, however, recognised that the number of mentally ill in contact with officers was increasing. The Inspector claimed that this increase was due to the failings of community care to sustain the mentally ill within the community.

6.15 The Use of Officer Judgement and Skill

Regardless of the small numbers, some procedures and practices raised issues. When examining the forces' approach to mental illness, the Inspector emphasised the reliance on the skills and discretion of the individual officer. As a working model of managing the mentally ill, the practice of the police force represented and encouraged the skill of the individual officer in negotiating and deciding on the most relevant course of action.

While this police organisation and other forces shared similar experiences, the Inspector argued that the situation in this particular force pointed to the requirement to manage through existing resources, rather than develop new or formal initiatives. Themes highlighted included the use of informal methods. This was advocated as a benefit as it utilised the skills of the individual police officer, instead of the relying on the organisational policy of the force.

The Inspector highlighted several practical elements which distinguished this police force from other forces. The police station remained in use as a place of safety. The Inspector acknowledged the inappropriateness of this practice, but recognised that the use of a police cell for the mentally ill was an existing requirement and necessity. Few local hospitals could or would accept what they seen as 'difficult cases'. More fundamentally, there were few places or emergency beds within the area. The perspective of the police in implementing section 136 in particular was focused on the care of the individual. The Inspector recognised the police role in the care and protection of the mentally ill, particularly their responsibility to 'look after' this group. This caring role places the police at the forefront of responsibility for the local mentally ill. As the Inspector stated,

"We don't just throw them out, we have a responsibility to look after people. The police station is the only place available to bring them. I expect that this will rise as community care fails."

[Inspector, Police Force 4]

The Inspector noted various additional effects. The wider policy consequences of these actions included a decrease in the use of sections two, three, and four of the Mental Health Act in the area, accompanied by limited access to further specialist health resources. What also emerged was the subsequent existence of the 'revolving door syndrome' for the mentally ill. However, a constant feature throughout this examination

was the relatively limited formal involvement of the police with the mentally ill. This should be seen as providing a context in which to assess their policies and actions.

The Inspector commented both on the importance of the role of the custody sergeant in the decision-making process and the overarching relevance and predominance of experience as a police tool. He saw the police having to develop methods by which to negotiate their environment and deal effectively with the problems encountered. Within this framework, specific roles and identities are instigated and played out.

To the Inspector experience was considered a prerequisite within the decision making process of a police officer. It aided in their ability to determine mental illness. The custody sergeant embodied the necessary experience to influence and assist in both deciding the presence of mental illness and the course of action to take. One characteristic of this police force, therefore, was their maintenance and usage of the existing skills of their police officers, recognising the necessity of individual power. Of all the police forces examined, this particular force acknowledged the prevalence of the police officer as decision maker and implementer,

"We rely on our officers and their knowledge and experience. We can't train them in everything."

[Inspector, Police Force 4]

At the time of interview the Inspector did not see the need for a formal policy approach towards the mentally ill, especially in the short term. He argued that the favoured approach emphasised the role of the police officer in deciding their own policy on a local level. On a wider level, the policy direction of the police was characterised as response to local demands, characteristics and needs. However, as highlighted by Community Care in 1997, a growing awareness was emerging within this police force. The discussion noted the nature, scale and scope of the dilemmas resulting from the

wider policy of community care and the experiences of the mentally ill, as viewed from a police perspective. As with other police forces, the formalisation of policy and objectives may result from this identification.

6. 16 Police Force 5 - The Dilemmas of Diversion

Police force 5, through the use of diversion at their police stations, developed a structured approach in their response towards the mentally ill. The approach was focused at the beginning of the court process and included the introduction of localised multi agency procedures. The dynamics and processes of diversion have been examined in relation to both the police force 2 and police force 3. The answer offered by this particular force carried several of the same details, objectives and connotations as highlighted by these forces. However, there are several issues particular to this force, which highlight their own experience and characteristics in the operation of the organisational response to the mentally ill. Operational officers from this police force also participated in the research [Station A].

The Inspector outlined the demographic features of the police area. This police force covered a diverse area, both urban and rural, with certain regions experiencing levels of unemployment and deprivation, in addition to incorporating the more prosperous commuter districts. The police stations within the area, therefore, faced a variety of situations. Equally, they responded to this diversity with differing policing styles, including the use of community policing. Two examples of the numbers processed in two local, but different police stations demonstrate the busy nature of the police force. On average, 3,000 prisoners in one station and 10,000 in the another were processed in one year.

The Inspector recognised the importance of prioritising police work in the face of such demands. One of these demands concerned the regular involvement of the police with the mentally ill. Any policy decisions and formulation of procedure for the police towards the mentally ill occurred within the Community Affairs department of the force. This department, established in August 1998 and based in police headquarters, dealt with matters pertaining to the issues of race relations, homophobia, youth justice and mental health. Their management of these areas was under review at the time of the research, so this assessment of the position of the police force was reliant on previous interviews of managerial personnel.

6.17 The Limitations of a Formal Approach

At the station level, [specifically one station where the operational officers were interviewed] the management of the mentally ill became a matter of attempting to match priorities with resources. While the formal use of section 136 was limited [often used on average eight times a year] there existed an informal network of regulars encountered by the police. Whilst the Inspector was aware of and knew this regular group, he believed that the officers could not provide adequate services or access the relevant and necessary agencies to assist these individuals fully. As the schemes such as diversion developed, questions occurred as to the application and usefulness of these procedures. The Inspector commented that the officers had a tendency to access the local community psychiatric nurse only when they believed it was appropriate, and when they were convinced of the presence of mental illness.

While the Inspector reinforced the claim for police accuracy in determining mental illness, if there was an element of doubt, he argued that officers tended to charge the individual with an offence. He held the opinion that regardless of the existence of the

diversion scheme at the local station, the individual, if mentally ill, would be dealt with later in the system. This opinion, if taken to the extreme, went against the ethos of diversion, as outlined in government policy. The Inspector went so far as to argue that he 'could not see much use for the nurse' within the scheme. The officers in the interviews echoed this in their reluctance to utilise formal procedure. Their experience of the diversion scheme was limited. This particular admission demonstrates that, while policy may have the support higher up the police hierarchical scale, for the individual officer practical concerns often overrode the implementation of force policy.

Arising from the restricted use of the scheme, there were several multi agency issues and concerns. On a local level, the Inspector perceived a lack of interagency communication, in addition to the limitation of interagency working and ideals.

Compared to the notions of sharing information outlined by police force 1, for this police force the fundamental problem between the agencies was the limited availability of information. While the Inspector felt that they were able to give information to other services, these services did not reciprocate. He felt that this was mainly on due to issues of confidentiality. The system was seen as 'ad hoc' and did nothing to help the police in the fulfilment of their roles and tasks.

6.18 Resource Issues

Resource related issues emerged from this dilemma for the police. The Inspector believed that there existed an expectation by other services of the police taking the lead role in any local scheme. The Inspector felt that this mean that many local agencies were expecting the police to take the initiative and manage the local problem of the mentally ill in the community. By association, there was a requirement that such a position involved a commitment to resources. This was reflected in the apparent

reluctance of the Inspector to engage with other agencies. As demonstrated earlier [reiterated by Thomas, 1994], the blurring of professional boundaries and accountability transpired into the ad hoc nature of agency co-operation.

The Inspector argued that the police aimed to distance themselves from mental health issues. The reluctance to become over-involved arose from a variety of reasons:

- the resources required.
- the overall priorities of the police as a force.
- the mandate of the Chief Constable and its effect on the operations of the police.
- their assertion that they were insufficiently trained to deal with mental illness.

This myriad of assertions subsequently led to the conclusion that the police officers managed the mentally ill in the community against a background of resource restriction and professional isolation. For the Inspector, it was the individual skills, methods and coping mechanisms of the operational officers which were used, rather than the overall policy direction of the force.

While the scheme of diversion was a resource available, when officers needed to decide on action to take other factors proved more influential. These included their knowledge of the community and the use of their skills as an officer. The Inspector questioned the effectiveness of such schemes for the officers in his station. This raises a further issue regarding transferring policy directed by management to the practice of the individual officer. It was felt that the police managers needed to ask several questions. These centred on the ability of diversion and training to fit into the every day world of their officers and the relevance of the organisational response to the experiences of the operational police.

The comments of the Inspector in this police force echoed the concerns and initial feelings of many officers throughout this research, both managerial officers and the officers on the beat,

"We try and distance ourselves from mental health issues, we are not trained, it's not our role to be involved with the mentally ill. We should charge them and let them progress through the system and get picked up later."

[Inspector, Police Force 5]

The organisational response, while being formulated to address issues systematically and provide officers with some structure, did not appear to alleviate their concerns and their desires to remain distance from the social problem of mental illness. The reluctance of the police to use more formal schemes and opportunities demonstrated the complexity of the situation and the dilemmas that subsequently arose.

6.19 Conclusion

In describing and detailing the five police forces' organisational response, certain themes emerge;

- the filtration of government policy down towards the individual forces
- their interpretation of these objectives
- the reality of such practice and the perspective of officers of its relevance
- the limitations of formal policy.

The responses of these individual forces illustrate the various organisational methods available to the police in dealing with the mentally ill. A significant determining factor can be seen as the perception of the scale of the situation and its implication for the operational police.

The two examples that illustrate the diversity of the experiences and their differing solutions are police force 1 and police force 4. They recognised the importance of matching responses to local conditions. However, one of the central areas in examining the issue of police involvement with the mentally ill concerned the development of training. Two police forces [police force 1 and police force 3] developed packages aimed at informing officers of their role and powers. Police force 5, however, raised issues over the extent of training and the need to recognise the variety of training police officers undertake in the course of their careers. In contrast, police force 4 at the time of interview, had no formal training schemes established, with the exception of the specialised role of the custody sergeant.

When examining this area specifically, one needs to assess several factors and features involved in the training of the police officer. Considerations include an examination of the aims of training, its expectations and its emphasis. One has to ask if it can be seen as stressing the translation of these expectations to the officer on the beat. In shaping the individual within the force, and providing a consistent response throughout practice, the most obvious and accessible method is training. This has relevance not only for their involvement with the mentally ill, and the role and tasks they perform in this capacity, but also in the development of and adherence to the 'police function'.

The obvious conflicts between policy and practice are evident in the every day interactions of the police with the mentally ill. While each force identified approaches to take with this particular group, very few could highlight methods of ensuring the fulfilment of policy. What still remains is the reliance on the individual officer to complete tasks and resolve situations, whilst adhering to the general philosophy of the force.

Chapter Seven: Discussion - The Police Experience of Mental Illness

7.1 Introduction

Any examination of the police relationship with the mentally ill is essentially one that highlights the conflicts, dilemmas and solutions available to the police in their every day practice. What becomes apparent is that the police have to manage a vulnerable group of individuals through the application of their knowledge. Encompassed within the framework of discretion, the style and type of action and behaviour is one that offers a myriad of possibilities. Central to these is the police understanding of their interactions and their experience.

The practice of the police aims to predict and determine the actions of the mentally ill.

All subsequent reactions can be seen to lead from this concern. On the individual level,
the dominant functions for police officers centre on their ability to negotiate their safety
and provide a short-term response. This is against a background of unpredictable
behaviours and outcomes.

In order to survive in the role of police officers, and to act according to expectations, it is assumed that officers develop a 'habitus'. Bourdieu's habitus, and its relation to practice and role identity, can be summarised as 'a unifying principle' [Pinto, in Shusterman, 1999, page 100] that aims to unite the practice of the individual. The notion of 'habitus' argues that the police are provided with the methods and skills to negotiate their environment. The problems raised include the dilemmas around care and control, the predominance of risk and the bureaucratic problems of managing the mentally ill within the community. The police have to organize such concerns in light of additional demands on their time and expertise. It is argued that their habitus provides a blueprint to understand behaviour and decide on action.

The police do not act in isolation of wider society. They operate within the current social context, with its objectives and policy implications, including mental health and community care. These effect the overall development of the police habitus. Whether encountering the mentally ill or involved in the management of offenders, the police officer as a law enforcer or peace maintainer is instilled with core beliefs, attitudes and methods of acting.

What Bourdieu implies in his description of habitus is that the identity of the agent and their subsequent practice carry central characteristics. These are maintained, modified, and re-established with each experience. In the specific interaction between the police and the mentally ill, certain aspects of the police habitus are brought to the fore. While they exist in the general role of the police officer, for this particular situation, they take on additional significance.

Within the officer's decision-making process, there is a reliance on their individual judgement and subjective understanding of the current situation. However, the police officers' ability to change their environment is limited. This specifically concerns the organisational and societal demands, which may be opposed to the individual needs and practice of the officer. There is added frustration at the policies, objectives, and guidelines presented.

The distinct features of the police habitus suggested by this research can be categorised as follows:

- The Preoccupation with Violence
- The Notion of Unpredictability
- The Development of a Strategy of Least Time
- The Use of Discretion
- The Individual Officer and the Organisational Demands
- The Concern for the Welfare of the Mentally Ill

7.2 The Preoccupation with Violence

One of the main issues to emerge from the research regards the link between violent behaviour and the mentally ill. The definition of violence should be addressed. As with risk, it has many different meanings. In the majority of studies looking at the association between violence and the mentally ill, it is homicide that attracts attention [Szmukler, 2000]. It is clear that when looking at homicides, the number of inquiries has increased the perception of the violent mentally ill individual. This is regardless of the decrease in the numbers of murders by this group. However, there is some link between mental illness and violence [Taylor, 2001, Walsh et al 2002], This link remains small when compared with figures outlining other acts of violence [Royal College of Psychiatrists, 2003]. As the Royal College of Psychiatrists [2003] confirms, only a minority of mentally ill individuals can be seen as violent.

The findings in this study outline that officers maintained this belief in the violent mentally ill regardless of the reality of violence. There are persistent views which effect how the mentally ill are seen. In light of the emphasis on risk [Munro et al, 2000, Taylor, 2001], this perception of and influence on the police (as well as other professionals) cannot be ignored. The officers in this study typified their interaction with the mentally ill as being one that is characterised by the risk of violence. The universality of their comments and the firm belief in the connection between violence and the mentally ill influenced how they enter the encounter. They were prepared not only for making difficult decisions in managing the situation, but also for the possibility of violence. The past experiences of the police suggested to them that the likelihood of violence increases when they encounter an individual with mental illness.

The reasons why officers maintained this assertion often included the unpredictable behaviour of the mentally ill. In some cases, officers had previous experience and knowledge of violent acts by the mentally ill. The belief in actual and potential violence is not unique to the interactions and the relationship of the police with the mentally ill. The presumption of danger and expectation of harm are important features of the characteristics [refs]. As a result of previous experience, the experience gained through various situations and the routine developed, an established set of beliefs is maintained. As the officers in the study demonstrated, their first consideration is one of safety, primarily of self, and then of others. In order to do this they believe in the necessary contemplation of danger and violence. The main objective for the officer is to return home safely [Bayley and Bittner, 1984].

The police maintained a heightened sense of awareness and a general suspicion towards events in the community. This includes a concern towards potential harm. To place in context this preoccupation with violence is to recognise that the police,

"...have an acute sense of where danger lies and what kinds of situations cause them the greatest difficulty in deciding what to do."

[Bayley and Bittner, 1984, page 37]

Previous research [e.g.Menzies, 1987] confirms that the association with violence and the mentally ill continues to be a feature of the police world of mental illness.

The assumption of violence and mental illness as a direct relationship is not a feature confined to the group of officers interviewed. The police officer's experience is one dominated by unpredictable occasions and violent intentions [Ruiz, 1993]. Within this police perspective, to have a mental illness is to increase the propensity for violence. This connection has its foundation in common assumptions surrounding the nature and consequence of mental illness. This has been reiterated by recent moral panics around

community care. Nor is it a new concern [Scheff, 1966]. Regardless of the limited truth in these foundations of violence and mental illness [Boyd et al, 1994], it remains a part of the belief system of the police.

Why should the police maintain this association? The connection with danger is part of being a police officer. The notion of danger within the police world is a complicated, interactive culmination of events. It is not merely consigned to individuals and specific encounters. Behind such notions are specific issues including role identity and isolation that sets the police officer apart from the public they serve. Through the wearing of a uniform and the provision of powers, the police officer is singled out as a visible and definable social agent.

Their ability to understand mental illness, while effective in determining the presence of a disorder, also carries with it negative assumptions around those labelled mentally ill. Via experience and through ideas around mental illness, the police continue to make this link. This can lead to caution when faced with certain situations and an expectation that a violent incident may result. It is part of the police belief system. One should not be surprised by its predominance in the arena of activity that encompasses the police and the mentally ill.

7.3 The Use of Force and the Mentally Ill

Amongst the skills of the police is the ability to determine and use appropriate force.

Force as a recognised part of policing distinguishes officers from other professional groups. In prioritising the possible actions to take, force enters the police decision making process. This is part of their role and within their powers. While the police do not always use force, it is their capacity to do so that is relevant. For, while the police have many aspects to their role in the community, it has to be recognised that the police

are seen primarily as law enforcers who maintain the capacity to use force [Waddington, 1993]. A significant group of officers interviewed acknowledged that the issue of force remained an issue, even if the individual concerned was perceived as mentally ill.

How the police decide to use force is complex, whether it is from an abstract standpoint or an analysis of situational criteria [Hunt, 1985]. Entering an unpredictable situation requires the officer to further examine the various options available to them, including the issue of force. Determining the level of force required in dealing with the mentally ill is an important issue. The type of force used varies from the hands on restraint to the use of CS gas.

For certain officers in this study, the use of CS gas was a viable option. Two groups of officers noted this area in particular. These officers operated in an area which contained community houses for those with mental illness. What is significant is that will a majority of officers interviewed experienced similar environments, but only two groups raised CS gas as an issue. This may be due to the experiences and attitude of the officers.

When examining decisions on the appropriateness and application of force on the mentally ill, more serious considerations are apparent. It can be viewed within the police preoccupation with violence. The officers felt that they needed to do what was required to ensure the re-establishment and continuation of order. If the use of force is one of the more effective options, then it is considered.

What is evident from officers is that, through the use of force, the dynamics of the encounter change. The police were no longer 'humouring' the mentally ill and calming

the situation down via conversation or in their role as 'professional persuaders'. Their primary role as law enforcers and controllers was maintained. Rather than moralising on the need for force, officers were responding to the demands of the situation. The distinction between the groups of mentally ill identified subsequently influenced the use of force.

While all mentally ill could be seen as potentially violent, this distinction between the groups of mentally ill manifested in the use of force. For several officers this resulted in the use of CS gas. The use of force and CS gas was seen as appropriate for those viewed as potentially violent. Thus, the police identification of individuals allows for the sanctioning of force. The use of CS gas by the police on the mentally ill is still a current and relevant issue [Bowler et al, 2000]. The situation summarizes the problem of policing the mentally ill. Attempting to control the situation involves

"...the need to balance respect for personal autonomy, both morally and legally, as against the need to control quickly and safely."

[Bowler et al, 2000, page 16]

The appropriateness of this form of control has been questioned as an operational method [Police Review, 1999]. The Police Complaints Authority has equally raised issues regarding the use of CS gas on the mentally ill. The nature and organisation of police forces means that the variance in such actions makes the examination of practice difficult. One also has to recognise that the responsibility for its use lies with the individual officer and their interpretation of events [Bowler et al, 2000]. The police, however, felt that in their everyday encounters with a variety of social groups it was a necessary addition.

The police exhibited various dilemmas in their interactions with the mentally ill. The use of CS gas in particular adds further to the confusion over the care or control role of the police when dealing with the mentally ill. The fact that the police demonstrate sympathy and an awareness of the vulnerabilities of the group did not pose contradictory objectives for the officers. Whilst they acknowledged the experience of the mentally ill in the community and the treatment they received, the police retained their need to control the situation faced. The need to contain the situation and the course of events are requirements to be met first.

7.4 The Symbolic Mentally Ill

The police had a workable stereotype of the mentally ill individual they were likely to come into contact with. What is created in the police mentality is the 'symbolic mentally ill'; the unpredictable, vulnerable, but potentially violent individual. Although deserving of sympathy, the individual had to be judged with caution. For while the police treat each incident individually, there is an all-encompassing image that serves to concentrate their efforts and provide a blueprint for the possible courses of action.

Regardless of the police knowledge and awareness of individuals within their community, each is assumed to have this potential, from those considered meek and quiet to those with an existing reputation for violence. It is far from clear how the effect of alcohol or drugs influences the police perception of the mentally ill – those identified by the police were seen as primarily as mentally ill. This carries with it various conflicting attitudes.

The mentally ill are interpreted as occupying two extremes, the meek and quiet, and the outwardly aggressive. The attitudes of the officers also displayed extremes in categorisation, seeing them both as 'victims of the community' and as potentially

violent 'nutters'. Polarising the mentally ill helps to categorise behaviour. For these officers, their categories of mental illness provided them with reliable generalisations. The 'symbolic mentally ill' is a character that appears at the forefront of the police decision making process.

The issue of drug and alcohol use were surprisingly absent from many discussions within the group of officers. The reasons why the officers did not raise these issues may be various. These include the perception of officers as to the nature of mental illness and those individuals they encounter. There additionally may be issues regarding the type of questions asked, which focused primarily on mental illness

7.5 The Notion of Predictability

To understand the police world is to recognise the need for order and predictability. For the police, to establish a routine is to minimise the unpredictability of events. This leads to a need for predictable behaviour patterns and outcomes. In practice, where possible, the police develop a model of working that recognises similar patterns, behaviours and routines. Thus, the ideal for the police in their interactions with the mentally ill [and with other social groups] is to know the dynamics and the circumstances of the encounter, prior to their start of the process. The reality is, however, different.

What the police do not like is the occurrence of disorder and unpredictability, in events and in behaviour. The likelihood of encountering such incidents is, however, at the centre of being a police officer. The reason for their involvement in social situations includes the characteristics they dislike the most; the presentation of disorder and 'unusualness'. It is this aspect that officers recognise as adding difficulties in managing the mentally ill in the community.

The police develop knowledge of their area and become familiar with both their surroundings and its residents. This was apparent through interviews with the operational officers, who developed knowledge of the local characteristics and the features of their community, including the local mentally ill. In doing so, they are establishing the parameters of order and familiarity. In achieving this, they can then be witness to anything which becomes an 'out of ordinary' event. Having information and knowledge helps to predict the course and outcome of events.

The police depend on gaining information as quickly and reliably as possible. Limited knowledge of events is not confined to interactions with the mentally ill. It includes the majority of citizen-evoked encounters faced [Waddington, 1993]. The citizen calls for assistance subsequently form the majority of the police interactions. They often involve non-criminal activities [Waddington, 1993]. The police in this study remarked on the type of calls received, including those related to domestic matters. The role of the police officer in these situations is continually one that defines them as

".... the servants of the public, with one consequence being that the social troubles they oversee often have little to do with the criminal law."

[Black, 1971, page 1091]

In professional and police evoked situations, the police have ideas around the expectations of the contact and action to be proposed. In the citizen encounters the police are reliant on the attitude of the individual, their reasons for assistance and their expectations of police action. Several determinant factors influence the situation, including the demeanour of individuals, the relationship between the caller and the 'offender', and the demographic characteristics of the offender population. In order to negotiate these competing demands, the police develop methods of control and routine.

The police routine includes varying and diverse encounters, facing an array of different problems [Waddington, 1993]. The officers in this study pointed to their role in assisting individuals in seemingly trivial, non-policing related calls. Their involvement with the mentally ill falls potentially in-between this myriad of possibilities, as a potentially criminal matter arising from a social dilemma. Via a process of examining past experiences, previous methods of managing and their expectation of future events, the officers form a basis for present action [Ruiz, 1993]. As Emerson remarks the police decisions and

"... assessments of the 'seriousness' of particular cases tend to be made in relation to the kinds of cases regularly encountered in that particular setting."

[Emerson, 1983, page 428]

For the police and their involvement with the mentally ill, their internal scale of judgement involves a reliance on previous interactions and outcomes. This includes not only the severity of the behaviour, but the likely effect on the police officer in terms of risk and in time spent at the incident. The requirement to establish a form of order where the dynamics of the situation are unknown serves the officer in the current situation. The prime predicament for the police in this study remains the unpredictable features of the encounters.

7.6 Police Methods in 'Making Sense' of Situations

The frequent assertion by officers interviewed of 'not knowing what they were going into' means that to make sense of the encounter and to prepare for any eventualities, the police develop their own method of understanding. The ultimate aim for the officer is the restoration of order. This means that, for officers, the situation follows a similar pattern and the outcomes can be predicted. As Bourdieu acknowledges, one of the outcomes of the habitus is the construction of predictability. That the establishment of

routine, ritual and predictability are police requirements highlights that their belief system and practice has within it a disposition for order.

However, the added dynamic to this situation is the possible presence of mental illness. The use of stereotypes of individuals and behaviours serves as a guide to the characteristics of the encounter. The police classification of mental illness involves a plethora of descriptions, mostly describing external and physical features. The reliance on generalised concepts of mental health and mental illness has further significance when those lay concepts are utilised by social agents such as the police. The physical manifestation of distress and the possibilities of harm to both self and others has come to represent to the police [and other social agencies] a definition of mental illness. It is considered deviant by virtue of its visibility and its abnormality. This, in police terms, is seen as something 'not being right'.

The placing of the mentally ill into groups is illustrated by Teplin et al [1992]. It is also not a feature uncommon to police interactions generally [more refs, McConville et al, 1991]. This links to the ideas presented in the notion of the 'symbolic mentally ill'.

Officers viewed the mentally ill as eccentrics or characters, those who were to be humoured, the 'quiet' mentally ill who display outward signs of distress, and the more violent individual, who the police felt either has been or will be violent.

The police labelling process can be seen as a method to determine subsequent action.

Outsider confirmation, via the initial call for assistance, other professionals or the individual's family members, also provides some presentation of order. It provides the officer with a rationale as to the likely course of events. This does not mean that the encounter suddenly becomes predictable. In association with their previous experience,

the officer is informed of the initial features. The expectation then is, conversely, one of unpredictability. This is with regard to behaviour, time and the availability of options.

The police often describe the difficulties they face [Waddington, 1999], accentuating the dilemmas posed by various social encounters. The police officers interviewed here were no different. However, in making sense of the police world, their reliance on extremes, in emphasising the methods involved in interpreting and defining actions, should be noted. It defines their need to categorise. To ignore this would be to ignore the subjects' interpretations of their world.

The recurring encounters with individuals known to the police offers a form of predictability and order. The realisation that the police constantly face the same mentally ill individuals should not be surprising. It demonstrates the characteristics of the system, of the inadequacies of mental health provision and the use of the police as a twenty four hour service. This is a major feature of the police-mentally ill encounter [Green, 1997].

In one sense situations become familiar and predictable are within the options available to the officers within this study. For while the social situation and its characteristics may be unpredictable to officers, the possible recourses for action are not. The uncertainty in accessing appropriate treatment means that, while officers are aware of what should be done in practice there are limited options. An informal approach versus formal responses is the main choice for officers. From this divide, officers decide on the action to take. This often means using discretion to determine what will prove most effective for individual officers.

7.7 The Development of a Strategy of Least Time

A picture has developed that recognises police officers use various mechanisms to assist them in dealing with the mentally ill in the community. This includes their understanding of violence, their perception of the mentally ill and the use of categories, and the establishment of order. The next stage proves to be a major concern for the police.

When entering encounters, the police have to decide on a series of options. This is obvious. The process in determining and deciding action is not. The police can enact a series of roles, as peace keeper or order maintainer [Waddington, 1993]. They can perform numerous tasks, formally acknowledging their role as social agents and following guidelines and designated procedures, or informally deciding action from the situation around them. This study suggests, however, that one of their prime concerns is to discover and develop a strategy of least time. This means that the officers in this study sought to end the situation as quickly as possible. They do this by adopting various approaches.

As 'professional persuaders', the police can use their social skills to encourage the individual to cease their behaviour, to return to their home and to terminate their involvement with the police. For the police in this role, it is their ability to 'manipulate' the situation that helps resolve the situation.

Their role as professional persuaders implies that the police embark on action that will result in a quick and effective solution. This particular role calls upon their skills to encourage people to comply and to determine the situation with the minimum of disruption. Persuading people to conform and return saves the police from the bureaucracies of hospitalisation, the justification of arrest and the sanctions of their

supervisors. Through the use of humour and conversation, they are attempting to 'calm the situation down' while also exhibiting choice as to the options to employ.

7.8 Priorities in Developing a Strategy of Least Time

Additional priorities are dependent on the interpretation of the officer. One of these considerations is the extent of officer involvement. Developing methods that result in the least time spent at the incident is important. This does not diminish the need to effectively resolve the situation, but rather recognises the time and organisational constraints involved in managing the mentally ill in the community. Finding ways out of this predicament becomes a priority.

Hospital as the most formal and logical place to take the mentally ill becomes the last resort for these officers. The bureaucratic nature of the formal response to the mentally ill deters the police officer from utilising the Accident and Emergency department at the local hospital. The formal approach as a means of managing the mentally ill is relegated to those cases where no other action is possible. The findings from this study are confirmed by Green [1997], Patch [1999] and Teplin et al [1992].

The informal disposals are those officers use most frequently, not only as a consequence of their discretion but also as a result of the difficulties within the system. The difficulties in inter agency working between the police and mental health professionals in particular points to a reluctance to become formally involved in the sanctioning of mental health disposals. Such difficulties involve the perceptions of each profession towards their role and responsibilities. The police remarked that they were often seen as 'dumping' people in the local hospital and leaving the health professions to deal with the situation.

The consideration of available resources enters the plans of the police. Thinking of what can be done at the start of the interaction is not unusual. As McConville et al [1991] remark, police officers often consider what actions to take prior to a deliberation of powers. In terms of the police with the mentally ill, this takes the guise of weighing up of the options that exist within the formal mental health system. The potential professional conflict that arises in the meeting of two social agencies has as much to do with practical considerations, rather than the ideological differences between the police and health and social services [Thomas, 1994]. The effects of hospitalisation and the limited co-operation of medical agencies indicate a prolonged and uncertain course of events for the police. The recourse to the strategy of least time aims to avoid these dilemmas and ensures the quickest way to exit the encounter, even if the options left are either informal or the utilisation of the criminal law.

The officers interviewed noted the reluctance of the medical profession to admit to hospital manifested itself in the diagnosis of personality disorder. This was a particular occurrence for officers in one station. The current situation regarding individuals with personality disorder centres on the need to establish effective services within the mental health system [Department of Health, 1999]. Again, this is within the context of risk and dangerousness [Mullen, 1999]. There are various issues around personality disorder, which have been highlighted over the last five years. With the predominance of specific cases, such as that of Michael Stone, questions have been raised regarding the notion of treatability and personality disorder. Under the existing Mental Health legislation if a psychiatrist deems those with personality disorder as untreatable, no services or care can be accessed. The consequences for the individual are various, with a deterioration of their illness. One other possible scenario is that such individuals return to the attention of the police.

The main concern for the officers was the use of this diagnosis on individuals they seen as in need of help, but who were denied it due to this label. Officers found out about this diagnosis when they escorted individuals to the local hospital. Officers felt it was the result of a lack of resources that led to the use of the personality disorder label, rather than issues of treatability. To the police, they were individuals in need of assistance, who then remained the responsibility of the police.

Officers look at the consequences of the actions they take, both for the present and for the future. More importantly for the officers is the perception that once the encounter has begun, they will be required to deal with the incident for the remainder of their shift. The informal options identified by officers in this study included persuasion, deescalation techniques and existing police powers such as breach of the peace. The use of breach of the peace highlights that when all else fails a return to existing police powers is available [Green, 1997, Teplin et al, 1992]. It may also prove to be the first act to take, as officers know the parameters of the powers and its influence in the course of events. It provides the officer with some indication of the length of time they will be involved with the mentally ill. It primarily serves in offering a police solution to a social problem, where the alternatives on offer are finite. The police, however, question the appropriateness of this disposal.

The above rationale illustrates the police development of a strategy of least time, by using their existing powers and skills. These prove to be significant police methods of managing the mentally ill in the community. All this occurs in an arena where others question their presence and the police question what they can actually achieve.

7.9 The Use of Discretion

Different components contribute to the notion of discretion; the power of the individual officer to choose action from an array of possibilities; discretion as embodied in the law; and discretion as individual judgement [Reiss Jr, 1984]. Its importance within this research involves not only understanding what the police do, but also what cannot be achieved. Discretion implies that individual judgement also extends to the roles to be acted. To be a 'social carer' suggests a an extension of the police role. However, in questioning this, one is also questioning the function of policing generally.

Discretion is an essential part of the police role. It begins in their training as officers. Essentially it means that officers develop a reliance on their individual judgement and flexibility. How they use their discretion depends on the appraisal of the situation, in assessing the options, the appropriateness of the various methods available, and the likelihood of their success as a method of management. Discretion involves an acknowledgement of the particulars of the social situation and the interpretation of its features [Reiss Jr, 1984].

What this study indicates is that the police have limits to their use of discretion. By the nature of many incidents, the officers find that their discretion has restrictions. They have not only to satisfy the individual's need for resolution, but also to achieve this in a manner that is appropriate to outsiders' expectations. Thereby in deciding on the course of action, officers have to consider the external appearance of the encounter. This is in association with their appraisal of what can be done within the constraints of resources. It was particularly evident from this police perspective that external features of the interaction contribute to their judgements of action. These include

- The context of the encounter
- The combination of the internal and external consideration of priorities
- The immediate situation and the safety of self
- The outsiders' observations
- The police perspective of mental illness.

Equally relevant is how the police perceive their involvement with the mentally ill.

They have to consider both the control of the situation and the welfare of the mentally ill.

7.10 The Police as 'Social Carers'

One factor that is involved in the use of discretion is the dilemmas which arise from the care and control of the mentally ill. This is particularly pertinent when considering the position of the police as social agents within a socio-medical framework. They are often required to fulfil a social care role, maintaining the welfare of the mentally ill while managing unpredictable behaviour. Their involvement in social problems centres on their ability to negotiate their world and to conform to their duty of protection and assistance. The fact that the police cannot ignore this aspect of their work means that the role of discretion continues to play a significant part. This is through the choices made in classifying the individual and in deciding what to do next

The roles the police fulfil in these encounters are numerous and involve both clear and less obvious aspects. Officers are performing the role of 'informal social carers'. This is, however, from the standpoint of law enforcers. The police noted the social circumstances surrounding the mentally ill, noting their social exclusion and their placements within the community. The police see the mentally ill as 'victims', being vulnerable to a variety of social problems. The fact that the police have to manage the

mentally ill at times of crisis further complicates the police perception. The dichotomy between the care and control aspects of policing is demonstrated in these encounters with the mentally ill, [Stephens and Becker, 1994]. The police exhibit sympathy and understanding for the vulnerability of the mentally ill, their place in society and the problems of community care. This is in apparent conflict with the emphasis on violence and dangerousness and the need to establish order. The dilemma of care and control remains.

Through the use of discretion officers have to decide whether to act informally or formally. The preferred solution of taking the mentally ill to a mental health setting proves difficult to achieve. Police discretion in their encounters with the mentally ill also centres on the police need to find a solution in the short term. This involves their development of a strategy of least time, the officers' ability to enact the role of professional persuaders and to use their powers to end the incident.

However, officers still operate within the restrictions of the social context and social system. This implies that, while their choices may be endless in theory, in practice they are restricted to what is practical, legal and achievable. What officers tend to find in their encounters with the mentally ill is that they 'go beyond the law'.

Using a criminal power when the evidence of a criminal act is limited and informally applying the law becomes an essential part of police work. This is especially when the recourse to other more appropriate alternatives is restricted. In bending the law, the police attempt to find a solution, both for themselves and the mentally ill. Making the incident fit their available powers suggests the dilemmas posed in operating within a socio-medical context and determining a social conclusion. This also recognises the vulnerability of the mentally ill.

7.11 The Difficulties with Discretion

The notions of 'Catch 22' dilemmas and the belief in 'we cannot win, no matter what we do' characterises the overall police perception of their encounters with the mentally ill. The police often have difficulties in deciding what to do with the mentally ill and the resources available. The officers recognise that they are open to observation and criticism regarding the extent and nature of their actions. Within the context of mental illness, this manifestation of police control of a distressed individual increases their concern for accountability and acceptance. In justifying their answers and responses, the police acknowledge that they have to resort to what they consider the best available option. This may mean that to society, to their superiors and to themselves they go beyond the law.

The police attempt to understand mental illness and their role can be summarized in one phrase; 'what can they do'. The limitation of options that the police consider appropriate and necessary leads to a frustration towards the situation itself. In line with the police expectations, the practical solutions available often do not meet the requirements of the officers. Discretion suggests that the police have a multitude of options at their disposal to provide a solution to the social problem posed. In reality, discretion is used in the first stages of the interaction, in judging the actions and behaviours of individuals and deciding in the next course of action. When it comes to the final stage, the police find they reach a 'dead end'.

What emerge from these encounters are the police as first aiders and as street corner psychiatrists, [Green, 1997, Teplin et al, 1992]. The continuation of the police providing 'psychiatric first aid' to a group of distressed individuals is a result of several differing components;

- The police organisational structure and its twenty four hour function
- The expectation of the public in calling the police
- The diversity of the police role
- The recognition of the police as one of the dominant 'social work' agencies in the community [Bean et al, 1991].

One cannot deny the role of the police in the fulfilment of social orientated tasks in the community. However, their mechanisms for management continue to evoke dilemmas, both for them and for the police organisation.

7.12 The Short Term Response to the Mentally Ill

A short term solution is the best the police can envisage in their interactions with the mentally ill. When we are asking the police to deal with the mentally ill in the community, it is assumed that they will provide an immediate response to an immediate situation. It is not expected that they will provide a long term answer to the continual problem faced. What this means for the police, however, is that the recurring nature of encounters remains. There is a frequent resort to the short term, 'sticking plaster' solution. The use of their discretion is a response to the situation faced, rather than an attempt to solve the situation completely and finally.

In deciding on the way forward in dealing with the mentally ill, the police take on the role of social commentators and reformers, attempting to solve the social problem both ideologically and practically. Asking what the police need to clarify their role in relation to the mentally ill is not to assume that their role will be diminished or that they will cease to be a community 'treatment resource' [ref]. The solutions of the police point to a change in the wider system to meet their needs and those of the mentally ill. The variety of possible strategies suggested covers both the criminal justice and mental health systems. The overall emphasis, however, is in adherence to the need to formulate

and function within a strategy that resolves the situation as quickly as possible. The outcome here is the appropriate placement of the mentally ill within treatment services and the 'freeing' of the officer to continue what they see as their primary role of law enforcer.

7.13 The Function of Discretion

What does discretion achieve for the police? It provides a basis for understanding the circumstances and dynamics of the event, in planning officers' individualised, but strategic formulation of action in relative autonomy and invisibility. This is central to understand any 'habitus' of the police. In this respect, discretion as a central and established component of the police role is almost an automatic mechanism or strategy for considering action.

However, this exists within the arena of rule enforcement and social expectations. The conflict for the police can be seen in the consequences of utilising their existing powers for an overt social problem. The more organisational conflicts that emerge say much about the difficulties in combining the ideology of mental health policy with the practicalities of implementation. The resulting picture is one that highlights the need of the police to continue and reinforce their role as 'psychiatric first aiders'. The police maintain that this approach is based on 'common sense' and 'intuition'. Their assessment of any situation revolves around the 'tried and tested' skills of a police officer.

A major issue regards the use of discretion by operational officers and the potential conflict with the expectations of the managerial officers dictating policy and guidelines. How this role fits with the wider structures including the police organisation itself is debatable. The organisational expectations of individual officers have to be seen in

context of the reality of practice and the unpredictability of managing the mentally ill.

As the divide between the expectations of the organisation and the practice of the officer illustrate, influencing and altering the practice of the officer is far from easy.

7.14 The Individual Officer and the Organisational Demands

One of the most pertinent issues arising is the connection of the individual officer to the aims and objectives of their force. In Bourdieu's terms, the 'habitus', beliefs and choices has to be placed in the context of the wider social 'field'. It is the result of 'feeling the game' [Bourdieu in Shusterman, 1999]. For the police officers, their habitus has certain elements which are in conflict with the police organisation. It is important to understand the connection between the two.

The work of Chan [1996] supposes that the 'cultural practice' of the police is the relationship between the habitus and the field. Within these frameworks there are not only the social and legal requirements of the police, but also the organisational environment surrounding their actions. It could be argued that the police habitus is a result of the following combination:

- their individual outlook
- the process of learning via experience
- the characteristics of the social situation/s
- the organisational arrangements that dictate 'what they should do'.

In reality the likelihood of tension increases when what they should do is not necessarily what they can do, or actually achieve in practice.

7.15 The Policies of the Police

What the police should do is indicated by the policies and procedures they receive from governmental guidelines and the police management. An important consideration

centres on the transferring of policy into practice, placing the organisational demands onto the practice of the officer. Whilst officers may be 'trained' via formal means in the essential components of police procedure, how they act and perceive in practice may be different.

Waddington [1993] points to the importance of policy interpretation, especially when officers are faced with numerous demands. Policy is often viewed as coming from the 'top down'. By establishing the ideological and practical dimensions of police practice, the police organisation is directing officers to the main notions of current practice. This may be seen as an attempt to formalise their behaviour. The police management are aiming to control the actions of their officers, ensuring consistency in approach and behaviour. However, the officers have to organise their working practice in such a manner that reflects their need to achieve organisation and solution in the face of dilemmas. Therefore,

"how the police respond to the diversity of demands made upon them is decided almost exclusively by the lowest echelons in the organisation."

[Waddington, 1993, page 156]

What is important is the appearance of any differences between the organisation expectations and the practice of the officer. Towing the organisation line may not always be possible. While acknowledging policy as procedure and guidelines, it is the presentation of practice that should be recognised [Grimshaw et al, 1987].

Can the policies and requirements of the organisational arena be imposed in practice?. The officers within this study displayed an appreciation of the core elements of the police role. While officers were aware of the policy dictating how they manage the mentally ill in the community, such as guidelines on the use of Section 136 and the various local schemes, in practice they continued to act informally. The predominant

issues remain the use of common sense and the skills used in every day policing. For some of the police management, the establishment of policy and procedure decreased the use of individual discretion by officers. What appears apparent is that the schemes and policies in practice were considered 'unworkable' by the police when dealing with a difficult situation quickly. It is clear that different forces and their policies produce differing responses, styles and outcomes [Reiner, 1992b], a conclusion supported by these findings. While policy may aim to be consistent and uniform, for individual officers their own assessment and negotiation of the environment remain dominant.

7.16 The Influence of Police Culture

The influence of the police culture on practice has often been held accountable for the ability of the police to decide and implement their own course of action. The power of the police culture facilitates officers in altering the policy of the organisation. Within the culture of the police emphasis is placed on the individual and collective beliefs and working practice. The findings of this research suggest that this culture, while not being dominant, has certain significance for the reality of police officers in the area of mental illness.

The culture of the officers in this study comprised the following characteristics:

- a recognition of the vulnerability of the mentally ill
- the inadequacies of the mental health system and the problems with multi agency working
- a belief in the unpredictable nature of the mentally ill, including the possibility of violence.
- the use of humour, both in dealing with the mentally ill and describing their encounters with them
- an adherence to the key methods of police work, including discretion and existing police powers

• a belief in a 'common sense' approach to managing the mentally ill, including their assessment of their problems.

For the officers in this study, there are several key considerations. These suggest a reliance on their own individualised dispositions as opposed to a reliance on the organisational structures. The lack of adherence to the formal schemes shows an awareness of the practicalities of policing a social problem in a law enforcing structure. Where schemes are in existence, the reluctance of the officers is such that they are not considered a viable option. For the individual officer, their use of the scheme is considered ad hoc and demonstrates a variance within the force. Officer knowledge and judgement continues to be prevalent. This is regardless of the managerial officers considering discretion to be a thing of the past in the police interactions with the mentally ill. Attempts at training officers to offer a uniform response do not diminish the use of informal approaches. Again, the inconsistent nature of such schemes means that the control of officers' actions through training becomes minimal.

The officer involvement in the decision making process continues to be limited [Goldsmith, 1990]. For while the officers remain social commentators in discussing the causation and solution of social problems, their role as policy makers is restricted to commenting on the options available and operating on a short term level.

Managing the mentally ill informally centres on methods of negotiation. The organisational response to the mentally ill and the development of an interagency approach are minimised. While remaining an organisational solution, for those operating in the community such responses have little impact. The reality for officers is that the solutions to the dilemmas posed have to remain with their own individual discretion and habitus.

7.17 'Managing the Police Officer'

Throughout this debate, we revisit the autonomy of the officer and its place within the organisational structure. The police are fundamentally decision makers [De Lint, 1998]. This has relevance for the area of discretion. However, it also has to be recognised that officers have additional powers. For,

"although always a discretionary decision maker, the constable has only recently become a chooser."

[De Lint, 1998, page 281]

Officers are often portrayed as agents of the bureaucratic process. However, the police are, in reality, the choosers of action. This insight is not new. What de Lint [1998] suggests is that this capacity is maximised through the use of formal training. By influencing, or in de Lint's [1998] terms 'shaping the officer', training could focus on their decision making process, alongside the imparting of information on the legislative and power structures in place.

Within this, situational factors, individual experience and interpretation are formalised and categorised [de Lint, 1998]. It organises the responses of officers on an individual level with an organisational structure. Conversely, the police culture maintains its presence and provides a relevant point of reference for officers. For any training would

"...attempt to rationalise judgement and structure decision, making police discretion consistent with organisational missions."

[De Lint, 1998, page 289]

This process is one of analysis, looking at the various stages in interactions and their role within them. Subsequently they are converted to the organisational demands. However, the dilemmas with this approach centre on the ability to quantify its success or failure. De Lint's [1998] proposals indicate mechanisms by which to inform the

officers of the process involved in their social interactions. Alternatively, this attempts to 'quantify the qualitative appreciation of the event' [de Lint, 1998, page 298].

In assessing the individual response to organisational demands, themes emerge that point to differing layers of interpretation. The issue of the transition of policy into practice suggests that while the organisation directs officers in their working style, in reality the officers formulise their individualised response to the complex and diverse social interactions. This can be illustrated through the examples of the police forces in this research. While the police forces identified incorporated policy specifically for the police practice with the mentally ill, their use is not only limited but also differed from station to station. This finding is reinforced by Goldsmith's [1990] argument, which remarks on the influence of police culture in defining responses to such situations involving the mentally ill.

The joining of the internal preferences of the officer with external governance is often fragmented. In answer to these dilemmas, Goldsmith [1990] argues for the development of negotiated rulemaking. Suggestions put forward not only highlight the positive nature of discretion, but also its inclusion within a formal system of rule development.

The main component of this argument acknowledges the advantages of a system where the operational officers have input into rules that direct their actions. This increases the likelihood of compliance. For,

".. It has been suggested......that rules which are internalised by those subject to them are more effective, because they become part of the subject's modus operandi."

[Goldsmith, 1990, page 97]

The promotion of joint rulemaking clarifies areas of police work that exhibit complexity and confusion. This serves as the organisational management of discretion.

However, for the individual officer it can be seen as providing further guidance and accountability in the administration of their tasks. The model allows for the possible development of differing styles of policing, including a more proactive approach. The implementation of the Crime and Disorder Act 1998 suggests that such an approach is increasingly likely, with its emphasis on community responsibility and joint partnerships.

The distinction between police tasks and the dilemmas arising from social problems enforces the argument for rule formation. Warnings, however, have to be made as to the extent of rules and the situations to be included. The end result may culminate in a curtailing of the police craft. Making the police response more predictable may not take into account or acknowledge the essential unpredictability of the majority of police encounters.

The arguments put forward by de Lint [1998], Goldsmith [1990] and Grimshaw et al [1987] recognise the diverging viewpoints and perspectives of the organisation and the individual officer. De Lint [1998] and Goldsmith [1990 suggest different approaches in an attempt to join the organisational aims with the practice of the officer. There needs to be an acknowledgement of the necessity of discretion and its role in the completion of police tasks. Equally, the need for the organisation to encompass the wider policy, social and political considerations has to be placed in context.

However, the police context is restricted to the demands, needs, and characteristics of the organisation. The police operate within a much wider context. These influence the actions and the role of the police. The reality of the police world recognises the social policy context in which they operate and its impact, both in the present and future.

7.18 The Reality of the Police World: Policy and Practice.

The police go into encounters with the mentally ill armed with a specific 'habitus', which places emphasis on the use of their discretion and the priorities in their strategy of least time. This informs the individual of their alternative options. It also develops into a way of thinking and being. Certain characteristics are core in the construction of the police role. Officers additionally recognise the importance of a service mentality and the inclusion of the police in diverse social problems. In doing so they are also conceding the importance of various social skills in communicating and managing crisis situations [Golden, 1982].

What does this mean for the police? In having such methods, the police have a selection of possible alternatives at their disposal. They resort to these particular features when managing their encounters with the mentally ill. How the police negotiate the social problem of mental illness and its visible social causalities involves potentially contradictory elements;

- as law enforcers
- as protectors of people and property
- as social reformers and social commentators.

In justifying their action, they are recognising their role as 'first aiders'. They do this through the use of the standard police response of self safety and in constructing their own understanding of the social process of mental illness. This behaviour tends to fall into the two divides of informality and the formal sanctioning of behaviour. The police were also frustrated by the inadequacies of the social system that brought the mentally ill to their attention. This division has potential conflict for the conscience of the police officer in the enactment of their role and through their use of force.

Throughout this, the police additionally attempt to avoid the bureaucracies and limitations imposed via the formal response to the mentally ill. The mentally ill as both victims and perpetrators confuses the situation. In seeing them as both, the police are justifying their resort to authority and their role as social commentators, a role usually assigned to Chief Constables [Loader and Mulcahy, 2001].

When reiterating their difficult position when dealing the mentally ill, they are explaining to themselves and each other that such encounters are entered reluctantly. People call the police because it is felt that they are the most suitable agency to deal with specific social problems [Shearing et al, 1977]. The police in this study confirm the frequent calls for assistance as a result of their duty to respond and to protect others. This element of the police world translates into the likelihood of their continued presence in the dilemmas inherent within the community. As the officers remark, when in doubt about a situation and the potential difficulties that may arise, the first response of the community is to call the police.

With their inability to refuse help, the police have to develop strategies and mechanisms that ensure their safety and the achievement of solutions. The avoidance of social problems, therefore, becomes impossible. Developing a framework of understanding, even unconsciously, is a prerequisite for formulating action. To call the police is to acknowledge that the situation is not only viewed as a crisis, but that a return to some form of order is required. Framing action within a law enforcement role, commonly associated with the police, subsequently requires the use of what is termed 'special procedures' and, in the words of the officers within this study, the 'bending of the law to get the job done'.

The spontaneity inferred by Bourdieu, in defence of the predeterminism of habitus, exists in terms of the officers' insistence in the individuality of cases and the need to respond to each incident as it arises. The foundation involved in such cases, to prepare for the interaction and to predict actions, is processed via the strategies developed. In dealing with the mentally ill, therefore, the officers go into the encounters with the anticipation of longevity, of frustration and of the difficulties in formally exiting the encounter.

The roles of the police in dealing with the mentally ill in the community acknowledge the reference made to existing and established police methods and skills. They use these skills and complete tasks under the guise of problem solvers. In conceptualising the modern role of the police, one has to question the expectations that are present, in society and in the police force itself. We return to the dilemma posed by Morgan and Newburn [1997] and Reiner [1992b] and reiterated by Waddington [1993]. We need to question whether the police are a force to exert authority and to maintain the status quo, or a service to administer solutions to a variety of dilemmas, social and legal, deviant and problematic.

The 'symbolic backdrop' of their interactions is important in understanding the functions of the police. This involves the wider context of policing, including the community itself. It influences their interaction with groups, especially those not defined as criminal. It also goes some way to explaining why the police resort to their endorsed methods of management, even with the mentally ill.

In intervening in the social problems of society, the police are offering a short-term solution to a long-term dilemma. The appropriateness of such involvement may prove to be irrelevant. Regardless of concerns over the criminalisation of the mentally ill

[Patch et al, 1999, Teplin, 1992b], society's reliance on the police and the concerns over their involvement, they maintain the function of restoring order. They are maintaining their position as 'frontline mental health workers [Green, 1997].

The policy context for the police involvement points to the various recourses to action that should be adhered to and sought, via wider policy objectives such as community care and the circulars introduced to formalise the police action towards the mentally ill. In examining the wider context, it is apparent that the police experience of mental illness is one possessed with difficulties. The interaction between the officer and their environment exhibits conflict in aims and objectives, between the subjective considerations of the individual and the external structures and guidelines.

The social policy surrounding mental illness has involved various policy objectives, notably the NHS and Community Care Act 1990 and the move towards deinstitutionalisation. This has been cited as providing a cause-effect influence on the numbers of police-mentally ill encounters. The wider context of mental health policy demonstrates a failure to cater for and design practice that corresponds to the expectations of society. The ideological basis and the expectations of community care, in particular, was an attempt to 'normalise' the mentally ill and avoid the stigmatisation of mental illness.

With the reform of the Mental Health Act, the suggestion for compulsory treatment in the community, the introduction of human rights legislation and the proposals for tighter management of personality disordered individuals, further examination of mental health policy is envisaged. The criticism levelled at this policy generally, and community care specifically, has been a result of the visible distress of the mentally ill and the manifestation of public fear in the cases of homicide by the mentally ill. The

expectations of the police in their interactions with the mentally ill arise from these external ramifications of mental health policy. In taking on the role of psychiatric first aiders, they are also involved in the external confirmation of the mentally ill as a social problem. Again, we return to the notion of the 'symbolic mentally ill' which is not only a police construction, but also a public image.

7.19 The Role of the Police Culture

This police culture is often seen as a negative and corrupting force [Chan, 1996]. Its ability to influence the actions of the police has been witnessed. Waddington [1999], in defence of the police culture, points to it as providing the police with a frame of reference. He sees the culture of the police as explaining to officers their role and experience, rather than the action taken. The police culture should, therefore, be appreciated. It encapsulates the 'essence' of policing and offers explanations of behaviour to officers themselves. It should be seen as 'the expression of common values, beliefs and attitudes within a police context' [Waddington, 1999, page 293].

The police use of discretion operates within cultural context. this will inform how that discretion is exercised. The issues raised when looking at the police culture include those concerned with gender and race. The role of women within the police has tended to be marginalized, with a limited number reaching the higher ranks of the force. The 'Gender Agenda' calls for action in this area that will fundamentally alter the cultural basis of the police. One other important issue concerns the police and race relations. The McPherson report highlighted the incidence of institutional racism and called for the police to develop strategies both in terms of race and gender. The challenge for the police force today is to adapt its 'canteen culture' to become more inclusive.

Culture is transformed, in Bourdieu's term, into the habitus of the police. This belief system is developed in practice. Waddington [1999] asks us to look at what the police are saying when referring to their culture, rather than its effect on actions and behaviours. The sensibility [Shearing and Ericson, 1991] in being a police officer is portrayed through the use of stories and the expressions of the police culture itself. The existence of police culture is not a new phenomenon, but rather is a constant and universal feature of the police world. It is one that appears stable within the constant change that has been a feature of the police domain.

Culture alone does not explain the actions of the police. Waddington [1999] argues that it should not be seen as a subverting force, corrupting the intentions of policy for the preferences of the individual officers. What this particular analysis implies is that the practice of the police is not necessary an extension of the statements made under the realms of the canteen culture. The police are not necessarily wholly responsive to the demands of the particular situation faced, but they are sensitive to the features and the possible outcomes of such events. The police are the 'playthings of situational constraints' [Waddington, 1999, page 302].

7.20 The Practice of the Police

It is apparent that a dichotomy remains. It centres on what the police can do and what they practice, with the emphasis being placed on the extent of the role of culture or habitus in influencing action. To achieve a more 'balanced' view, one has to recognise that in any profession there is a core set of beliefs which guide action. The police are no different. The features that differentiate the police from others are the visible nature of their role, the presumption of authority, the legality of their position, and the repercussions of action. The habitus serves the police officer as it serves any other

social agent, by providing strategies of action. Whether it is defined as a habitus within this specific theoretical framework or as a culture, the end result is a reference for action and for understanding.

The police, while experiencing possible constrictions on their practice, rely on their experience as police officers to manage the social interactions they encounter. Training provides some semblance of an organisational and consistent response. However, in reality, as highlighted by the police in this study, the social features of the encounter cannot be solely catered for in any training package. The similarities of the encounters themselves provide a blueprint for action. These blueprints also acknowledge the individuality of situations. What becomes important is the confirmation of the police culture within these encounters and their specific habitus in relation to their interactions with the mentally ill.

One has to remember the political and social context of policing, especially within a climate of performance indicators and efficiency drives. The police remain an economic service, providing a facility for their consumers [Sullivan, 1998]. This illustrates the need for the police to provide 'value for money'. This has to be achieved whilst also adhering to the aims of law and order, within specific budgets and designated objectives. Community policing can be seen as a move towards making the practice of the police structured to the needs of the community. The current emphasis on community partnerships, recognising the limitations of the police in managing crime in particular, signals the development of a proactive approach. The experiences of the community sergeant highlight the possible benefits of such an approach. It also clarifies the role of the police in relation to the community and recognises the importance of discretion as a working tool.

However, there is a danger when examining the police response to the mentally ill of emphasising the morality of officers in these interactions. In assuming that their action is either 'right or wrong', there may be a tendency to view them as primarily coercive agents of control, even when it is suggested that they are in fact more tolerant than many sections of society itself [Waddington, 1999]. Attaching labels is to attempt a subjective and moral definition of that behaviour. As Haines and Sutton [2000] remark, criminological theory involves judgementalism, dissecting morality from empiricism. The moral and subjective classification of mental illness can equally apply.

In describing and diagnosing mental illness we are in fact deciding on what is normal and what is not. To involve the police in such process is to recognise them as social agents at a crucial point in society. The police act in the way they do as a result of the defining features of their environment. While the issue of mental illness and the police response in the community poses specific problems, repeated issues remain, including the unpredictable nature of the encounter itself.

The police are a service with several defining features:

- The power and ability to use force
- The inability to refuse to assist
- The twenty four facility
- The powers to search, arrest, and detain individuals.

While the police are seen as distinct, one needs to question the extent of any difference with other professions, including social work, psychiatric nursing and probation. Each has the capacity to use controlling aspects that the police may see as solely their remit. From my own observations and perspective there are similarities between police officers and mental health professionals, specifically in how they manage the

environment. This includes the negotiation of safety, the emphasis on risk and the use of 'black humour' to survive their shift. The management of working environments may require similar characteristics to maintain a sense of balance.

7. 21 The Continuation of Police Encounters with the Mentally III

What has been demonstrated by this study is that the police encounters with the mentally ill, the police perceptions, their actions and the options available to officers have continued to remain the same. Regardless of social policy and organisational changes, the interactions of the police with the mentally ill continue to exhibit the same features and dynamics. The police retain their role as 'psychiatric first aiders', at the forefront of social problems, maintaining their discretion to decide and resolve the social problems they encounter.

As Bourdieu demonstrates, the habitus provides individuals with strategies and considerations for choice. This particular habitus, although centred on the police interaction with the mentally ill, can be transferred and utilised in describing the police world of encounters. While this study does not claim to decipher the entire world of the police, it does aim to provide indications of possible courses of action taken by the police and their rationale for such behaviour. It offers one perspective in which to view police behaviour and the tasks and roles they perform.

Reference has been made to the relevance of previous research in confirming these findings. These studies have indicated that the police possess certain roles and utilise possible routes of action in their interactions with the mentally ill. To try and distinguish the police actions with the mentally ill from their interactions with other groups in society, one has to remember the uniqueness of the police role, as a twenty four hour 'trouble-shooting service' [Shearing et al, 1977]. Whatever the basis and the

circumstances of the police interaction with individuals, the expectations of the police include their ability to respond to any crisis and any situation.

A review of the research into the specific notions of policing the mentally ill has, thus, tended to demonstrate the police categorisation process of labelling behaviour [Teplin et al, 1992], identified the problems faced by the police and their subsequent response [Mathews Jr, 1970], and offer explanations as to how they determine action and their use of discretion. The findings of this research aim to show that, whatever the social climate, the issues facing the police in their interactions with the mentally ill continue; with similar problems, similar processes and similar solutions.

In using Bourdieu, an effort has been made to try and explain why. Combining the wider social environment with the individual 'habitus' of the officer has highlighted the dilemmas in placing organisational demands and expectations onto the practice of the officer. The wider context of mental health not only includes the police official response, but also social policy. This provides a framework for understanding the reasons as to why the police continue to be involved in the social problems of society. The police-mentally ill interaction process is influenced by both internal and external factors; the officers' individual beliefs, the conception of their role, the influence of outsiders to the process, the expectations of others and the contextual arena of the police organisation. It is anticipated that not only will these features continue to play a fundamental part, but that this particular encounter, between the police and the mentally ill, will continue as long as the police have a role within society.

Chan [1996b] highlights the issues for criminology generally in responding to the reflexivity of Bourdieu's perspective, in determining the location of the criminologist in the field of criminology. In describing the individuality of the police encounter with

the mentally ill, this study has documented Chan's description of the logic of the field, in recognising the position of the researcher in the process, the reliance on existing social theories, and the relevance for policy of such research.

The development of criminological thought since the Thatcher years of government has demonstrated a tendency to deconstruct the elements of crime and crime control, focusing on the divergence of the wider issues through to the new realist perspective, and the emphasis on marginal groups and situations [Jefferson and Shapland, 1994]. The police, however, continue to have a place within the wider examination of the actions of public agencies. For ,

"The end of the decade saw no let up in the activity in the arena of criminal justice, with the police, the prisons and the courts the subject of public concern and official scrutiny."

[Jefferson and Shapland, 1994, page 283]

This research has, therefore, continued the process of examining the police as a social agency and as part of the wider social structure. The examination of individual experiences offers a snapshot of the police experience and an appreciation of the context of their actions.

7.22 Limitations of the research

This research has highlighted the experiences of the police and their response to the mentally ill in the community. However, there are certain limitations to the study which place the findings in context. These involve methodological and practical issues.

The obvious limitation of the research was the difficulty in obtaining direct observation of police practice. The advantages of witnessing the practice of the police would have enabled a more thorough insight, examining how the police and the mentally ill interact.

Additionally, this cannot be a comparative study. Its aim was to examine the experiences of officers in a specific geographical location. It was a picture of police involvement with the mentally ill. No quantitative data was accessed, primarily as a result of lack of monitoring. This also means that no comparisons can be made. The reliance was on interviews with various police officers. The conclusions are therefore limited to this group of officers. Additionally, this would make the study difficult to replicate.

The areas examined in the research attempted to provide an overview of the police experience. However, one omission was the issue of substance use and mental illness. Whether this was due to the perceptions of the officers interviewed or the questions asked at interview is difficult to ascertain. As a result, this topic was not discussed. Given the complexity of mental illness and the prevalence of substance abuse, this was an interesting exclusion.

Having recognised the limitations of the research, it is evident that future methodologies would benefit from

- Random selection to test out the hypothesis that the increase in mentally ill in the community has facilitated a change in roles and practice of the police.
- Direct participant observation of police practice
- A greater number of interviews with female officers to examine any gender differences in approaches to the mentally ill.
- The involvement of service users
- Investigations concerning the role of drug and alcohol use

7.23 Recommendations

From this research, it is evident that there are several issues that have significance for the police. One issue is also clear; the police will remain central in the care of the mentally ill in the community. The recommendations suggested include both local policy suggestions and practical considerations. They recognise that the police require more effective mechanisms to manage the mentally ill in the community. What recommendations that can be made include the following:

1. Training in mental illness. Whilst training was seen as minimal in terms of content and impact, there is a strong argument for developing a consistent training package that addresses several key areas for the police. This is a difficult issue. The police are expected to understand a variety of different areas. Any training package has to recognise the local experience of the police.

Training could include the following areas:

- Changes to the Mental Health Legislation. It is anticipated that there will be changes to the 1983 Mental Health Act. There is also the possibility that these changes will involve the police in several areas, especially when managing those considered high risk. If the reforms are implemented, the relevance for the police will include developments in new services, the use of compulsory powers in the community and concerns around risk.
- The Classification of Mental Illness, including personality disorder. There appeared to be some confusion as to the different types of mental illness presented by individuals in the community. While the police should not be trained as clinicians, there is a case of aiding officers in the identification of illness rather than relying on more general concepts.
- *The Presentation of Mental Illness*, including the signs of distress. For officers, the main dilemma centred on the behaviour of the individual and their

understanding of mental illness. Two of the training packages examined considered this area as important. This will assist officers in understanding the difficulties and complexity of mental illness.

- The Effect of the Police on the Individual. Equally important is the development of self awareness in officers as to the effect of their involvement on individuals. Officers did exhibit some understanding of their influence on others' behaviour. This is an area that would facilitate the police further in their approach to the mentally ill.
- Cultural and Gender Issues. It is important that any training package takes into
 account the cultural context and issues relevant to the experience of ethnic
 minorities. There are various concerns over the inclusion of ethnic minorities
 and the criminal justice and mental health systems.
- *Drugs and Alcohol*. The impact on mental illness. Mental illness is a complex disorder. When this is exacerbated by the use of drugs and alcohol, the situation is further complicated. It is important that officers are aware of this issue and the possible issues that arise.
- Local Schemes and Local Agencies. It was apparent that the actions of officers were in response to local problems and dilemmas. Often they obtained information and knowledge through their various experiences. There may be a need to inform and train officers in the relevant issues in the local community, including the various schemes, local hospitals and hostels.

There are various mechanisms to achieve this. The development of diversion schemes has provided officers with links into service with specialist knowledge. This may be a

mechanism for training officers in the basic areas of mental health and increasing knowledge regarding local developments. There is also in argument for developing the use of distance learning packs. However, whilst these may be practical for the police, the monitoring of training is difficult. Any training initiative should begin whilst officers are still on probation and undergoing their basic education.

- 2. The Development of Local Schemes. Whilst some officers did not recognise or use the existing schemes, there is some evidence that well co-ordinated projects aid the police and the mentally ill in the criminal justice system [Rix 1994]. These include mental health liaison officers and local forums. It is evident that the police play a significant part in the management of the mentally ill in the community. Any approach should recognise the complexity of the situation and the need to coordinate responses. Any local scheme or project should also knowledge the skills of the police and the importance of their contribution. Lessons can be learnt from the development of juvenile panels, multi agency risk panels and other established forums which link a variety of agencies from the public sector. Such schemes, however, should not rely on the personalities of the participants to develop the project or maintain its success, but rather be based on robust policies and guidelines.
- 3. Resources for the Placement of the Mentally III. One of the main dilemmas for the police was the placement of the mentally ill once they came to the attention of the criminal justice system. Several officers remarked on the need for treatment centres for those at crisis point and who could not be admitted to hospital. It appears that the pressure on resources means that local hospitals do not have the facilities to deal with police referrals. Such treatment centres could provide assessment and short term specialist care. For the police, this would be a more appropriate solution that would

allow them to return to the community and for the mentally ill to receive treatment. The solutions available include:

- Specific Section 136 Suites. At the time of the research, and from further correspondence, there are still no dedicated section 136 suites within the local areas. Such a facility would resolve many of the issues pertinent to the police. However, the problem of resources remains. Whilst other suites have been developed elsewhere, in the local areas there are still pressures on psychiatric services.
- Identified Crisis Intervention Teams. Whilst crisis intervention teams have developed in the local areas that were involved in this research, there is an argument for establishing a more structured response and procedure. This would formally establish links between the police and the mental health system.
- 4. Sharing of Agency Information and Resources. In the area of mental illness, the number of services and agencies involved with an individual are numerous. The various changes to mental health policy have meant that responsibility has been fragmented. For the police, sharing information and resources further provides them with a consistent response to the mentally ill. The police highlight the importance in knowing their area and any developments that occur. By sharing information and any relevant resources, the police would be better informed of the options available. Equally, the mentally ill would receive treatment from more relevant and suitable agencies. The solutions available have both procedural and resource implications:
 - Identified Mental Health Link Workers. These workers should be identified both within the police and in local social and health services. The purpose of

developing such posts would be to provide clear lines of communication and demarcation, with both services being aware of roles and responsibilities. The evidence of the more successful schemes [Rix, 1994] recognises that there needs to be a awareness of different professional ideology and methods of working.

- Clear Guidelines on the Issue of Confidentiality. One of the major issues in sharing information between agencies is confidentiality. The police felt that they could not obtain the relevant information that would help them assist the mentally ill in the community. Equally, there are fundamental concerns regarding the disclosure of client information and the relevance of data protection information. There is an argument for the development of a protocol that both protects the individual, but also aids the police.
- Local directory of resources and policies. As mental health policy has changed significantly over the last twenty years, the development of new systems, initiatives and schemes has been prolific. The police acknowledge the need to know the situation in their local area. Yet they are often unaware of local service planning and the establishment of new units. One solution may be a process of involving and informing the police of local policies and services. This has the advantage of providing further access to appropriate resources. The police may also then be aware of the type and numbers of mentally ill in the community and be more prepared for any issues that arise.

There are wide implications for the treatment of the mentally ill in the community.

These include awareness and perception of mental illness, its presentation and the experiences of the mentally ill through their contact with the police. The police handle difficult situations with a variety of methods.

7.24 Conclusion

What is clear is that the police have a pragmatic response to the care and management of the mentally ill in the community. When dealing with the various incidents that arise, the police have to assess the situation, manage any risk and face the dilemmas of dealing with vulnerable individuals. The police in this study recognised the complexities of mental illness. How they manage and negotiate these difficulties is by relying on their skills as officers. It is the nature of police work that is highlighted in their involvement with the mentally ill. The decisions made by these officers are characterised by their understanding of the community, the local resources and an awareness of the need to re-establish order. The police resort to making pragmatic decisions because that provides them with the most appropriate method available.

Police Interview Schedule The Mentally Ill and the Police

1998

Group number

Individual officer details: Rank

Length of Service

Role

Areas

- 1. Context and Experience of Mentally ill
- 2. Cues for Mental Illness
- 3. Decision making process
- 4. Perceptions of the encounters and the police role

1. Experience of the Mentally ill - context

Involves examining the frequency, social context the source of referral, location, time of day etc

- 1a) How do you first become involved with the mentally ill how is contact established
- 1b) What is the most frequent time for contact
- 1c) Who provides the main source of contact
- 1d) Where do you see the mentally ill -i.e. the streets/public or private settings, at the police station, or via other agencies
- 1e) What incidents usually determine police involvement or action

2. Cues for mental illness - identification

Involves general impressions, how individuals are defined as mentally ill, what behaviour indicates mental illness

2a) What tells you someone is mentally ill

- 2b) What is the typical behaviour and actions of the mentally ill encountered
- 2c) What skills do you use to make this assessment
- 2d) How is this confirmed

3. The decision making process – factors considered

Involves the first actions of the officers, the police perception of their role in these encounters, actions taken and factors taken into account

- 3a) What do you do first
- 3b) What are priorities
- 3c) Who is contacted
- 3d) The process of the encounter what happens next
- 3e) What action is taken and the rationale used
- 3f) How do you see your role in these situations

4. The police perceptions of their interactions and their role

Involves how they view their involvement with the mentally ill, dilemmas, solutions and their overall perception of this area of police work

- 4a) Have the numbers of mentally ill people who come into contact with you increased and what has the effect of this increase been
- 4b) Have the interactions/situations with the mentally ill changed if yes, how
- 4c) Are there any dilemmas in your involvement with the mentally ill what are they
- 4d) What do you think you should do with the mentally ill
- 4e) What are your management techniques
- 4f) What training or schemes help you in your involvement with the mentally ill
- 4g) Solutions what would make the situation easier or better to deal with

Police Interview Schedule Police Response to the Mentally ill – Managerial Officers

Length of Service

Role

Areas Covered:

- 1. The Police Involvement with the Mentally ill context
- 2. The Policy of the Force for Dealing With the Mentally ill
- 3. The Practical Implementation of Policy

1. The Police Involvement with the Mentally ill - context

- 1a) Has the police involvement with the mentally ill increased
- 1b) In what areas do the police deal with the mentally ill
- 1c) What effect has wider social policy had on the actions and practice of the police service
- 1d) Do you see any changes in this police involvement
- 1e) What do you see as the main role of the police in these situations
- 1f) What should be the police role when dealing with the mentally ill

2. The Policy of the Force for Dealing With the Mentally ill

- 2a) What is the policy of the police service in dealing with the mentally ill
- 2b) What specific schemes have been developed by the force
- 2c) How do these schemes work
- 2d) What is the rationale for such schemes
- 2e) How is the scheme monitored and evaluated
- 2f) What have been the benefits and drawbacks for such schemes/policy developments
- 2g) What do the police perceive as the outcomes for such policy objectives

3. The Practical Implementation of Policy

- 3a) How do you view the effectiveness of the schemes
- 3b) What has been the effect on multi agency working
- 3c)How has police practice changed
- 3d) What do you see as future developments in this area for the police service

BIBLIOGRAPHY

Adams, D, [1997], Custody Pressure, Police Review, 5 December, 16-17

Adams, D, [1998], Clockwise, Police Review, 25 June, 22-24

Adler, J, [1996], Fear is a Four Lettered Word: Safety and Resource Issues for Probationary Constables, Issues in Criminological and Legal Psychology, 25, 39-46

Ahmad, W [1995], Book Review: Health, Race and Ethnicity: Making Sense of the Evidence, SHARE, Issue 11 [May] 9-10

Ainsworth, P, Pease, K, [1987], Police Work: Psychology in Action, London: British Psychological Society/Methuen

Alibhai, Y [1988], Broken and By Passed: Mental Health and Ethnic Minorities, New Society, 6th May, 2-3

Allen, D, [1996], Women on the Edge, Community Care, 5-11 Dec, 24-25

Alty, A, [1996] Seclusion: Exploring the Myths and Misconceptions, OpenMind 79, Feb/March, 18-19

Anshel, H, M, [2000], A Conceptual Model and Implications for Coping with Stressful Events in Police Work, Criminal Justice and Behaviour, 27, 3, 375-400

Appleby, L [2000], Safer Services: Conclusions from the Report of the National Confidential Inquiry, Advances in Psychiatric Treatment, 6, 5-15

Arcuri, A, Gunn, M, Lester, D, [1987], Measuring Police Discretion, Perceptual and Motor Skills, 64, 774

Ashworth, A, [1995], The Criminal Process: An Evaluation Study, Oxford: Clarendon Press

Audit Commission, [1996a], Detecting A Change: Progress in Tackling Crime, London: Audit Commission

Audit Commission, [1996b], Streetwise: Effective Police Patrol, London: Audit Commission

Audit Commission, [1996c], Tackling Patrol Effectively: Management Handbook, London: Audit Commission

Audit Commission, [1998], The Doctor's Bill: The Provision of Forensic Medical Services to the police, London: Audit Commission

Backer-Holst, T, [1994], A New Window of Opportunity: The Implications of the Reed Report for Psychiatric Care, Psychiatric Care, 1,1, March/April, 15-18

Baker, E, [1997], The Introduction of Supervision Registers in England and Wales: A Risk Communications Analysis, The Journal of Forensic Psychiatry, Vol 8, No 1, May 15-35

Bardsley, B, [1987] Flowers in Hell: An Investigation into Women and Crime, London and New York: Pandora,

Barnes, M, Bowl, R, Fisher, M, [1990], Sectioned: Social Services and the 1983 Mental Health Act, London: Routledge

Barnes, R, [1995], Support Systems, Community Care, 12-18 January, 28-29

Bayley, D, Bittner, E, [1984], Learning the Skills of Policing, Law and Contemporary Problems, 47, 4, 35-59

Bean, P, Bingley, W, Bynoe, I, Faulkner, A, Rassaby, E, Rogers, A, [1991], Out of Harm's Way: MIND's Research into Police and Psychiatric Action under Section 136 Of the Mental Health Act, London: MIND Publications

Bean, P, Mounser, P, [1993], Discharged from Mental Hospitals, Basingstoke: MIND/Macmillan

Bean, P [2001] Mental Disorder and Community Safety, Basingstoke: Macmillan

Bebbington, P, Stephens N, Clarke, B, [1997], A Gateway to Efficiency: The Role Of the Psychiatric Emergency Clinic, The Evidence, Spring, Macmillan Magazines

Beckett, I, [1998], Command and Patrol, Police Review, 2 Jan, 24-25

Belfrage, H, [1998], A Ten Year Follow Up of Criminality in Stockholm Mental Patients: New evidence for a Relation Between Mental Disorder and Crime, British Journal of Criminology, Vol 38, No 1 Winter, 145-155

Benyon, J, [1984], Scarman and After: Essays Reflecting on Lord Scarman's Report, the Riots and their Aftermath, Oxford: Pergamon Press

Berkley, M, [1995] People with Mental Health Problems Arrested in Oxford : A Review of Needs for Services, Oxford : The Elmore Committee

Bhat, A, Carr-hill, R, Ohri, S, [1988], Britain's Black Population : A New Perspective, Aldershot : Gower

Bickman, L, [1974], The Social Power of a Uniform, Journal of Applied Social Psychology, 4,1, 47-61

Bittner, E, [1967a], The Police on Skid Row: A Study of Peace Keeping, American Sociological Review, Vol 32, No 5, October, 699-715

Bittner, E, [1967b], Police Discretion in Emergency Apprehension of Mentally III Persons, Social Problems, 14, 278-292

Blaauw, E, Vermunt, R, Kerkhof, A, [1996], Treatment in Police Custody: Organisational Characteristics and Performance of Custodial Divisions in Dutch Police Organisations, Criminal Justice and Behaviour, Vol 23, No 4, Dec, 553-571

Black, D, Reiss, J, [1970], Police Control of Juveniles, American Sociological Review, 35,1, 63-77

Black, D, [1971], The Social Organisation of Arrest, Stanford Law Review, 23, 1087-1111

Blom-Cooper, L, Grounds, A, Guinan, P, Parker, A, Taylor, M, [1996], The Case of Jason Mitchell: Report of The Independent Panel of Inquiry, London: Duckworth

Boast, N, Chesterman, P, [1995], Black People and Secure Psychiatric Facilities, Patterns of Processing and the Role of Stereotypes, British Journal of Criminology, Vol 35, No 2, Spring, 218-234

Boesch, E, [1997], Reasons for a Symbolic Concept of Action, Culture and Psychology, 3,3, 423-431

Bonovitz, Caldwell, J, Bonovitz, S J, [1981], Diversion of the Mentally Ill into the Criminal Justice System: The Police Intervention Perspective, American Journal of Psychiatry, 138, 7, 973-976

Borum, R, Deane Williams, M, Steadman J H, Morrissey, J, [1998], Police Perspectives on Responding to Mentally III People in Crisis: Perceptions of Program Effectiveness, Behavioural Sciences and the Law, 16, 393-405

Borum, R, [2000], Improving High Risk Encounters Between People with Mental Illness and the Police, The Journal of the American Academy of Psychiatry and the Law, 28, 332-337

Bourdieu, P, [1977], Outline of a Theory of Practice, London: Cambridge University Press

Bourdieu, P, [1985], The Genesis of the Concepts of Habitus and of Field, Sociocriticism, 2,2, 11-24

Bourdieu, P, [1990], In Other Words: Essays Towards a Reflexive Sociology, Cambridge: Polity Press

Bourdieu, P, [1991], Language and Symbolic Power, Cambridge, Massachusetts: Harvard University Press

Bourdieu, P, [1998], Practical Reason, Cambridge: Polity Press

Boyd, W, [1994] Confidential Inquiry into Homicides and Suicides by Mentally Ill People: A Preliminary Report, London: Published by Steering Committee of Inquiry

Bracken, P, Thomas, P, Mental Health Legislation: Time for a Real Change, OpenMind, 90, March/April, 17

Bradley-Taylor, M, [1996], The Mental State, Police Review, 17th May, 18-21

Bratby, L, [1998a], Preventive Measures, Police Review, July 13, 24-26

Bratby, L, [1998b], Responding to Treatment, Police Review, August 28, 22-23

Bratby, L, [1998c], Fact Finding Team to Help Develop Crime and Disorder Partnerships, Police Review, September 25, 12-13

Bratby, L, [1999], Fragmented and Confused Police Training Needs National Strategy, Police Review, April 23, 5

Bratton, W, [1996], How We Cleaned Up New York, The Sunday Times, November 24, 2

Brown, D, [1990], Black People, Mental Health and the Courts, London: NACRO

Brindle, D, [1995], Force Urged on Mentally ill in Community, The Guardian, 17th January, 1

Bryan, B, Dadzie, S, Scafe, S, [1988], The Heart of the Race: Black Women's Lives in Britain, London: Virago

Buchanan, A [ed], [2001], Care of the Mentally Disordered Offender in the Community, Oxford: Oxford University Press

Burns, L, [1995], Care in Crisis, Community Care, Nov 30-Dec 6, 1

Burns, T, Leibowitz, J, [1997], The Care Programme Approach: Time for Frank Talking, Psychiatric Bulletin, Vol 21, No 7, 426-429

Busfield, J [ed], [2001], Rethinking the Sociology of Mental Health, Oxford: Blackwell

Butler, A, J, P, [1992], Police Management, Aldershot: Dartmouth

Bynoe, I, [1992], The Role of Criminal Justice Agencies in the Diversion of Mentally Disordered Offenders, London: MIND publications

Campbell, D, [1996], Courts Not Using Mental Tests to Save Money, The Guardian, 11th July, 1

Capstick, L, [1992], Black and Minority Communities and Mental Health, MIND Information Publications

Carpenter, M, Hannon, R V, McCleery, G, Wanerling, J A, [1988], Ethnic Differences in Seclusion and Restraint, The Journal of Nervous and Mental Disease, Vol 176, No 12, 726-731

Carty, P, [1996], Cell Mate, The Guardian, 20 March, 7

Chambliss, W, Liell, J, [1966], The Legal Process in the Community Setting : A Study of Local Law Enforcement, Crime and Delinquency, 12, 310-317

Chan, J, [1996a], Changing Police Culture, British Journal of Criminology, 36,1, Winter, 109-134

Chan, J, [1996b], The Future of Criminology: An Introduction, Current Issues in Criminal Justice 8,1, 7-13

Cherrett, M, [1996a] Mentally Disordered Offenders: Part One, Police Review, 3 May, 30-31

Cherrett, M, [1996b], Mentally Disordered Offenders : Part Two, Police Review, 10 May, 28

Choongh, S, [1998], Policing the Dross: A Social Disciplinary Model of Policing, British Journal of Criminology, Vol 38, No 4, Autumn, 623-634

Clark, J, [1965], Isolation of the Police: A Comparison of the British and American Situations, The Journal of Criminal Law, Criminology and Police Science, Vol 56, No 3, 307-319

Cobb, A, [1996a], Homicide and Suicide - The Boyd Inquiry Reports, OpenMind, 79, Feb/March, 4

Cobb, A, [1996b], Making Sense of the Process and Effects of ECT, Nursing Times, Vol 92, No 10, 6 March, 32-33

Coffey, M, [1995], Supervision Registers and Mental Health Problems, Nursing Times, Vol 91, No 28, 12 July, 36-37

Coffey, M, [1996], Supervised Discharge, Nursing Times, Vol 92, No 26, 26 June, 50-53

Cohen, D, [1988], Forgotten Millions: The Treatment of the Mentally III - A Global Perspective, London: Paladin/Grafton Books

Colombo, A, [1997], Understanding Mentally Disordered Offenders, A Multi Agency Perspective, Aldershot: Ashgate

Commission for Racial Equality, [1995], The Race Implications of the Mental Health (Patients in the Community) Bill: A Position Paper

Community Care, [1995a], Call for Better Practice to Help Mentally ill People, 23 Feb-1 March, 2

Community Care [1995b], Scots' Care Programme Approach Criticised, 7-13 Dec, 2

Community Care [1996a], Mental Health Death Toll, 14 Dec- 4 Jan, 1

Community Care [1996b], Study Defends Offender Diversion Schemes, 15-21 August, 3

Community Care [1997], Psychiatric Patients Putting Too Much Pressure on Police, May29-June 4, 6

Cope, R, [1990], Mentally Disordered Offenders from Ethnic Minorities, Hampshire Probation Library

Cordner, G, [2000], A Community Policing Approach to Persons with Mental Illness, The Journal of the American Academy of Psychiatry and the Law, 28, 326-331

Corry, P, [1999], Duty of Care, Police Review, June 25, 23-24

Crandon, L, [1997], Measure for Measure, Police Review, October 17, 22-23

Crepaz-Keay, D, [1997] A Mental Health Manifesto, OpenMind, 86, 5

Cummings, E, Cummings, I, Edell, L, [1965], Policeman as Philosopher, Guide and Friend, Social Problems, 12, 3, 276-286

Curra, J, [2000], The Relativity of Deviance, London: Sage

Curtis, S, [1989], Juvenile Offending: Prevention through Intermediate Treatment, London: Batsford

Davis, R, [1995], Inquiry into Inquiries, OpenMind, June/July, 75, 6-8

Davis, S, Thornicroft, G, Leese, M, Higginbottom, A, Phelan, M, [1996], Ethnic Differences in Risk of Compulsory Psychiatric Admission Among Representative Cases of Psychosis in London, British Medical Journal, 312, 2 March, 533-536

Dawson, J, Romans, S, Gibbs, Ratter, N, [2003], Ambivalence about Community Treatment Orders, International Journal of Law and Psychiatry, 26, 243-255

De Gale, H, Hanlon, P, Hubbard, M, Morgan, S, [1993], Improving Practice in the Criminal Justice System: A Training Manual, Northern Curriculum Development Project, Leeds: CCETSW

De Lint, W, [1998], Regulating Autonomy: Police Discretion as a Problem for Training, Canadian Journal of Criminology, July, 40,3, 277-304

De Vaus, D, A, [1996], Surveys in Social Science, London: UCL Press

Deahl, M, [1996], Insanity Plea, Times Higher, June 21, 18

Delaney, B, [1996], The Mentally III in the Community, Liverpool: Merseyside Police Force

Dell, S, Grounds, A, James, K, Robertson, G, [1991], Mentally Disordered Remanded Prisoners: Summary of Report to the Home Office, London: Home Office Research and Planning Unit

Denzin, N, Lincoln, Y, S, [1994] Handbook of Qualitative Research, London: Sage

Dey, I, [1993], Qualitative Data Analysis, A User Friendly Guide for Social Scientists, London and New York: Routledge

Dickens, D, Fontana, A, [1994], Postmodernism and Social Inquiry, London: University College London Press/The Guildford Press

Dixon, D, Bottomley, K, Coleman, C A, Gill, M, Wall, D, [1989], Reality and Rules in the Construction and Regulation of Police Suspicion, International Journal of the Sociology of Law, 17, 185-206

Dobson, R, [1995a], Moving Away from Victorian Values, Community Care, 5-11 October, 11

Dohrenwend, B, Bernard, W, Kolb, V, Lawrence, D, [1962], The Orientations of Leaders in an Urban Area toward Problems of Mental Illness, American Journal of Psychiatry, 118, 683-691

Donaldson, R, [1999], Market Forces, Police Review, 26 March, 26-27

Dotter, D, Roebuck, J, [1988], The Labeling Approach Re-examined: Interactionism and the Components of Deviance, Deviant Behaviour, 9,1, 19-32

Doyle, H, Delaney, W, [1994], Police Referral of Compulsory Admissions; A Comparative Study, Irish Journal of Psychological Medicine, 11, 3, September, 116-119

Duff, P, [1997], Diversion from Prosecution into Psychiatric Care: Who Controls the Gates?, British Journal of Criminology, 37,1, Winter, 15-34

Dutt, R, [1994], Report of Black Mentally Disordered Offenders Conference, London: Race Equality Unit

Dyson, J, [1996], Rubbing Out the Black Marks, The Guardian, May 7, 13

Easterday, L, Papademas, D, Schorr, L, Valentine, C, [1977], The Making of the Female Researcher: Role Problems in Fieldwork, Urban Life, 6, 3, October, 333-348

Eastman, N, Mullins, M, [1999], Prosecuting the Mentally Disordered, The Journal of Forensic Psychiatry, 10,3, 487-501

Eaton, L, [1996] Schizophrenic Murder Fears are Unfounded, Nursing Times, 92, 8, February 21, 9

Elliot, M, [1995], Care Management in the Community: A Case study, Nursing Times, 91, 48, November 29, 34-5

Ely, P, Denney, D, [1987] Social Work in a Multi-Racial Society, Aldershot: Gower

Emerson, R, [1983], Holistic Effects in Social Control Decision Making, Law and Society Review, 17,3, 425-455

Emsley, C, [1991], The English Police: A Political and Social History, New York and London: Longman

Fabianic, D, A, [1979], Authoritarianism in Criminal Justice Literature, Journal of Police Science and Administration, 7, 1, 53-64

Fair, J, [1997], Inquiry Slams Services for 'Failure to Care', Community Care, 13-19 March, 7

Fanti, G, [1993], Masking the Problem, Community Care, April 29, 17

Farrel, E, [1997], The Complete Guide to Mental Health, London: Vermilion

Farrington, D, Gunn, J, [1985], Reactions to Crime; the Public, the Police, Courts and Prison, Chicester: John Wiley & Sons

Feldman, P, [1978], Criminal Behaviour : A Psychological Analysis, London: John Wiley & Sons

Feldman, P, [1993], The Psychology of Crime, Cambridge : Cambridge University Press

Fernando, S, [1991], Mental Health, Race and Culture, Basingstoke and London: MIND Publications/Macmillan

Fernando, S, [1992], Roots of Racism in Psychiatry, OpenMind, Oct/Nov, 59, 10-11

Fernando, S, [1995], Mental Health in a Multi Ethnic Society, London : Routledge

Fielding, N, G, Fielding, J, [1991], Police Attitudes to Crime and Punishment; Certainties and Dilemmas, British Journal of Criminology, Winter, 31,1, 39-53

Fielding, N, [2002], Theorizing Community Policing, British Journal of Criminology, 42, 147-163

Finn, M, Stalans, L, [1997], The Influence of Gender and Mental State on Police Decisions in Domestic Assault Cases, Criminal Justice and Behaviour, 24, 2, 157-176

Finn, P, Sullivan, M, [1989], Police Handling of the Mentally III; Sharing Responsibility with the Mental Health System, Journal of Criminal Justice, 17, 1-14

Fitzgerald, M, Muncie, J, [1983], System of Justice: An Introduction to the Criminal Justice System in England and Wales, Oxford: Blackwell

Fletcher, H, [1992], Mentally Disordered Offenders - The Reed Report, NAPO News, 42, October, 6

Ford, R, Rose, D, [1997a], Into the Wilderness, Community Care, 28 August-3 September, 2-4

Ford, R, Rose, D, [1997b], Heads and Tails, Health Service Journal, 6 November, 28-29

Foucault, M, [1967], Madness and Civilisation: A History of Insanity in the Age of Reason, London: Routledge

Foucault, M, [1977], Discipline and Punish: The Birth of the Prison, London: Penguin

Francis, E, [1993], Psychiatric Racism and Social Police: Black People and the Psychiatric Services, in Inside Babylon; The Caribbean Diaspora in Britain,

Francis, E, [1996], Community Care, Danger and Black People, OpenMind, April/May, 80, 4-5

Francis, J, [1993a], Black and Blue, Community Care, January 7, 20-21

Francis, J, [1993b], Worth the Hard Work, Community Care, Feb 11, 18-19

Francis, J. [1993c], Meeting the Challenge, Community Care, July 29, 20-21

Freeman, R, Roesch, R, [1989], Mental Disorder and The Criminal Justice System: A Review, International Journal of Law and Psychiatry, 12, 105-115

Gallant, M, J, Kleinman, S, [1983], Symbolic Interactionism vs. Ethnomethodology, Symbolic Interactionism, Spring, 6,1, 1-18

Galliher, J, [1971], Explanations of Police Behaviour, Sociological Quarterly, 12, 308-318

Gardener, J, Pugh, R, [1996], Threatened Eviction and Hospital Readmission from Community Homes, Nursing Times, September 4, 12

Garland, D, [1992], Criminological Knowledge and Its Relation to Power: Foucault's Genealogy and Criminology Today, British Journal Of Criminology, Autumn, 32, 4, 403-423

Gauntlett, N, Johnson, N, Navano, T, [1995], Meeting the Mental Health Needs of Ethnic Minority Groups, Nursing Times, October 18, 36-37

Gelder, M, Gath, D, Mayou, R, [1994], Concise Oxford Textbook of Psychiatry, Oxford: Oxford University Press

Gheverghese, G, J, Reddy, V, Searle-Chatterjee, M, [1990], Eurocentrism in the Social Sciences, Race and Class, April-June, 31, 4, 1-26

Gibbons, D, [1994], Talking about Crime and Criminals: Problems and Issues in Theory Development in Criminology, London: Prentice Hall

Gibbons, S, [1997a], Searching Questions, Police Review, February 21, 14-16

Gibbons, S, [1997b], PCA Proposals to Improve Care of Detainees, Police Review, March 7, 4-5

Giddons, A, [1982], Sociology: A Brief but Critical Introduction, Basingstoke: Macmillan

Giles, J, [1998], The Myth of Madness, Police Review, March 6, 24-25

Gillig, P, Dumaine, M, Stammer, J, Hillard, J, Grubb, P, [1990], What Do Police Officers Really Want From The Mental Health System, Hospital and Community Psychiatry, June, 14, 6, 663-665

Glassner, B, Hertz, R, [1999], Qualitative Sociology as Everyday Life, London : Sage

Glasman, D, [1995], Agencies Fear for Clients in Custody, Community Care, 20-26 July, 5

Glover, G, R, Leese, M, McCrone, P, [1999], More Severe Mental Illness is More Concentrated in Deprived Areas, British Journal of Psychiatry, 175, 544-548

Godwin, T, [1997], Mentally Disordered Offenders - The Role of the Police Service, A Paper Delivered to the National Conference on Mentally Disordered Offenders - Current Developments, University of York, 29-30 September

Goffman, E, [1961], Asylums: Essays on the Social Situation of Mental Patients and Other Inmates, London: Penguin

Goffman, E, [1963], Behaviour in Public Places, Notes on the Social Organisation of Gatherings, New York and London: The Free Press

Golden, K, M, [1982], The Police Role: Perceptions and Preferences, Journal of Police Science and Administration, 10,1, 108-111

Golding, J, [1996], The Care Programme Approach, OpenMind, April-May, 80, 10

Goldman, H, H, Adams, N, H, Taube, C, A, [1983], Deinstitutionalisation: The Data Demythologised, Hospital and Community Psychiatry, 34, 129-134

Goldsmith, A, [1990], Taking Police Culture Seriously: Police Discretion and the Limits of Law, Policing and Society, 1, 2, 91-114

Goldstein, H, [1963], Police Discretion: The Ideal versus the Real, Public Administration Review, September, 23, 140-148

Goldstein, J, [1960], Police Discretion Not to Invoke the Criminal Process: Low Visibility Decisions in the Administration of Justice, The Yale Law Journal, March, 69, 4, 543-594

Goodwin, S, [1990], Community Care and the Future of Mental Health Service Provision, Aldershot: Avebury

Gorman, J, [1992], Conditions in Prison, London: MIND Publications

Gough, M, [1995], Informed Consent, Community Care, 16-22 February, 17

Gray, R, Smedley, N, Thomas, B, [1997], The Use of Section 136: A Review of Research 1972-1996, Psychiatric Care, 4, 2, 62-66

Greater Manchester Mentally Disordered Offenders Co-Ordination Group [1996], Responding to the Needs of Mentally Disordered Offenders and Others with Similar Needs: Developing an Effective Multi Agency Response in Greater Manchester, North West Regional Health Authority

Green, T, [1997], Police as Frontline Mental Health Workers: The Decision to Arrest or Refer to Mental Health Agencies, International Journal of Law and Psychiatry, 20, 4, 469-486

Grimshaw, R, Jefferson, T, [1987], Interpreting Police Work: Policy and Practice In Forms of Beat Policing, London: Allen and Unwin

Guite, H, [1994], Editorial: Violence and Psychiatric Patients: A View From The Community, Criminal Behaviour and Mental Health, 4, 245-252

Gunn, J, Robertson, G, Dell, S, Way, C, [1978], Psychiatric Aspects of Imprisonment, London: Institute of Psychiatry/Academic Press

Gunn, J, [1991], Meeting the Need, Hampshire Probation Library

Gunn, J, Maden, A, Swinton, M, [1991], Treatment Needs of Prisoners with Psychiatric Disorders, British Medical Journal, 303,

Gurney, N, J, [1985], Not One of the Guys: The Female Researcher In A Male Dominated Setting, Qualitative Sociology, Spring, 8, 1, 42-62

Hackney, A, [1993], Truth About Drugs, Nursing Times, February 24, 66-68

Hackney Social Services/Health Authority and Metropolitan Police, [1995], Mental Health Act Procedures Section 136, Hackney Borough Agencies

Hadley, R, Clough, R, [1996], Care in Chaos: Frustration and Challenge in Community Care, London: Cassell

Hadley, T, Goldman, H, [1995], Effect of Recent Health and Social Service Policy Reforms on Britain's Mental Health System, British Medical Journal, 311, 1556-1558

Haines, F, Sutton, A, [2000], Criminology as Religion?: Profane Thoughts about Sacred Values, British Journal of Criminology, Spring, 40, 146-162

Halpern, D, [2001], Moral Values, Social Trust and Inequality, Can Values Explain Crime?, British Journal of Criminology, 41, 236-251

Hamilton, I, Roy, D, [1995], The Care Programme Approach at Work in Mental Health Care, Nursing Times, December 20, 35-7

Handy, C, [1985], Understanding Organisations, London, Penguin

Hanewitcz, B, W, Fransway, L, M, O'Neil, W, M, [1982], Improving the Linkages between Community Mental Health and the Police, Journal of Police Science and Administration, 10, 2, 218-223

Harrison, G, Halton, A, Owens, D, [1989], Severe Mental Disorder in Afro-Caribbean Patients: Some Social, Demographic and Service Factors, Psychological Medicine, 19

Harrison, K, [1997a], Lock them up, Community Care, 9-15 January, 24-25

Harrison, K, [1997b], Falling Short, OpenMInd, January/February, 83, 12-13

Harrison, K, [1997c], Lords Defeats Crime Bill, OpenMind, March/April, 84, 4

Harrison, K, [1997d], Taking Liberties, Community Care, 28 Aug-3 Sept, 8

Hart, S, Hare, R, [1996], Psychopathy and Antisocial Personality Disorder, Current Opinion in Psychiatry, 9, 129-132

Health Service Journal [1997], Communication Failures Led to Deaths, Health Service Journal, 17 July, 5

Heape, N, [1991], Mentally Disordered Offenders, South West Association of Chief Officers of Probation

Heginbotham, C, [1996], In-patients and Impatience, The Guardian, February 28, 6-7

Hemmings, A, [1997], Calls for Help, Health Service Journal, 12 June, 34-35

Henderson, G, McCollan, A, [1997], Building Bridges, Community Care, 23-29 October, 2-3

Hennessy, S, [1995], Living on the Edge - Individual and Community Need Associated with People Presenting with Personality Difficulties Living in the Cheltenham Area: A Report for the Cheltenham Locality Planning Group, South West Region: NSF

Herbst, K, Gunn, J, [1991], The Mentally Disordered Offender, London: Butterworth-Heinemann in association with the Mental Health Foundation

Herman, D, Opler, L, Felix, A, Valncia, E, Wyatt, R, Susser, E, [2000], A Critical Time Intervention with Mentally III Men; Impact on Psychiatric Symptoms, The Journal of Nervous and Mental Disease, 188, 3, 135-140

Hester, S, Eglin, P, [1992], A Sociology Of Crime, London and New York: Routledge

Hewitt, C, [1997], Discharge of Duties, Community Care, 28 Aug-3 Sept, 6

Hicks, J, [1997a], Dispelling the Dixon Myth, Police Review, June 6, 16-17

Hicks, J, [1997b], Policing at the Crossroads, Police Review, June 13, 24-25

Hicks, J, [1997c], Moving Forward, Police Review, June 20, 24-25

Higgins, R, Richardson, A, [1993], The Right Approach, Community Care, July 15, 20-21

Hill, R, Shepherd, G, [1997], Joint Resolutions, Health Service Journal, April 10, 30-31

Hills, G, [1993], Diverting Tactics, Nursing Times, January 6-12, 24-27

Hirst, J, [1996], Mental Health Meltdown, Community Care, 10-16 October 10-11

Hirst, J, [1997], Primary Concerns, Community Care, 18 Dec-7Jan, 8-9

Hirst, J, [1998], Community Scare, Community Care, 29 Jan- 4 Feb, 10-11

HMSO, [1995], Mental Health (Patients in the Community) Act 1995, London: HMSO

Hochstedler, E, [1987], Twice Cursed? The Mentally Disordered Criminal Defendant, Criminal Justice and Behavior, 14, 3, 251-267

Holley, H, L, Arboleda-Florez, J, [1988], Criminalisation of the Mentally Ill, Part 1 Police Perceptions, Canadian Journal of Psychiatry, March, 33, 81-86

Hollin, C, [1989], Psychology and Crime: An Introduction to Criminological Psychology, London, Routledge

Home Office [1992], Race and the Criminal Justice System: A Home Office Publication under Section 95 of the Criminal Justice Act 1991, London: HMSO

Home Office, [1995], Home Office Circular No 12/95, Mentally Disordered Offenders, London: Home Office

Home Office and Department of Health, [1995], Mentally Disordered Offenders Inter Agency Working, London: HMSO

Home Office Research and Statistics Department, [1996], Operation of Certain Police Powers Under PACE; England and Wales 1995, Issue 12, London: Government Statistical Service/Home Office

Home Office, [1997], Police and Criminal Evidence Act 1984, Codes of Practice, London: HMSO

Hood, R, [1992], Race and Sentencing - A Study in the Crown Court : A Report for the Commission for Racial Equality, Oxford : Clarendon Press

Horn, R, [1996], Negotiating Research Access to Organisations, The Psychologist, December, 551-554

Horn, R, Hollin, C, [1997], Police Beliefs about Women Who Offend, Legal and Criminal Psychology, 2, 193-204

Howe, S, [1998], Dangerous Persons Index Needed Says Father of Killed PC, Police Review, April 24, 4-5

Hudson, B, [1997], Local Differences, Health Service Journal, September 18, 31-33

Hunt, J, [1985], Police Accounts of Normal Force, Urban Life, 13, 4, 315-341

Hunt, L, [1995], Pioneering Venture is Prime Example of Good Practice, The Independent, June 15, 6

Hurst, P, Shields, G, O'Mahony, P, [1997], Beat Regeneration, Police Review, June 12, 16-18

Huxley, P, Reilly, S, Harrison, J, Hadi, M, [1997], Information Breakdown, Health Service Journal, June 5, 28-29

Inner London Nacro Race Issues Advisory Committee, [1995], Interim Report 1993/94, London: NACRO

Inoue, S, [1980], Interactions and Interpretations in Everyday Life, Studies in Symbolic Interactionism, 3, 1-24

Institute of Race Relations, [1993]. Community Care : The Black Experience, London, Institute of Race Relations

Institute for Study and Treatment of Delinquency and MIND, [1994], Values for Change - Mental Health Services in a Secure Environment : Report on Conference, London : MIND

Janus, S, Bess, B, F, Cadelen, J, Greenwald, H, [1980], Training Police Officers to Distinguish Mental Illness, American Journal of Psychiatry, 137, 2, 228-229

Jefferson, T, Shapland, J, [1994], Criminal Justice and the Production of Order and Control; Criminological Research in UK in the 1980s, British Journal of Criminology, Summer, 34, 3, 265-285

Jenkins, C, [1997a], The Mersey Beat, Police Review, August 29, 16-17

Jenkins, C, [1997b], Safe Keeping, Police Review, October 31, 19-20

Jenkins, C, [1998a], Medical Checkup, Police Review, March 6, 14-15

Jenkins, C, [1998b], Most of the Force's Custody Officers Have No Formal Training in the Role, Police Review, November 27, 5

Jenkins, C, [1999a], Community Safety Schemes are Failing to Tackle Local Problems, Police Review, February 12, 11

Jenkins, C, [1999b] South East JBBs Criticise CS Report, Police Review, July 16, 5

Jenkins, R, [1992], Pierre Bourdieu, London: Routledge

Joseph, N, Nicholas, A, [1972], The Uniform : A Sociological Perspective, American Journal of Sociology, 77, 719-730

Jessor, R, Colby, A, Schweder, R, [1996], Ethnography and Human Development: Context and Meaning in Social Inquiry, Chicago and London: The University of Chicago Press

Johnstone, L, [1989], Users and Abusers of Psychiatry: A Critical Look at Traditional Psychiatric Practice, London: Routledge

Jones, T, Newburn, T, Smith, D, [1996], Policing and the Idea of Democracy, British Journal of Criminology, Spring, 36,2, 182-198

Jones. T, Newburn, T, [2002], The Transformation of Policing? Understanding Current Trends in Policing Systems, British Journal of Criminology, 42, 129-146

Judd, M, C, Smith, E, R, Kidder, L, H, [1991], Research Methods in Social Relations, Fort Worth, Chicago, San Francisco: Holt, Rinehart and Winston

Keeble, P, Metcalfe, C, Riley, T, Waterson, J, Winter, D, [1997], Cross Purposes, Health Service Journal, May 8, 28-29

Kerrigan, T, [1997], Professional Hostages, Community Care, Aug 28-3 Sept, 7

Keys, R, [1997], Right Approach, Health Service Journal, Sept 25, 28-29

Kirk, S, Kutchins, H, [1994], The Myth of the Reliability of DSM, The Journal of Mind and Behaviour, Winter and Spring, vol15, 1, 2, 71-86

Kirk-Smith, M, [1996], How to Design an Effective Research Study, Nursing Times, July 10, 40-41

Kleinman, A, [1987], Anthropology and Psychiatry: The Role of Culture in Cross Cultural Research on Illness, British Journal of Psychiatry, 151, 447-454

Kliman, L, [1998], Lethal Weapon, The Guardian, August 5, 8-9

Klinger, D A, [1996a], Quantifying Law in Police-Citizen Encounters, Journal of Quantitative Criminology, 12, 4, 391-415

Klinger, D, A, [1996b], Bringing Crime Back in: Toward a Better Understanding of Police Arrest Decisions, Journal of Research in Crime and Delinquency, 33, 3, 333-336

Kwakwa, J, [1995], Alternatives to Hospital-based Mental Health Care, Nursing Times, June 7, 38-39

Laberge, D, Morin, D, [1995], The Overuse of Criminal Justice Dispositions: Failure of Diversionary Policies in the Management of Mental Health Problems, International Journal of Law and Psychiatry, 18, 4, 389-414

Lafave, W, [1962], The Police and Non-enforcement of the Law, Part 1, Wisconsin Law Review, January, 104-137

Lagos, J, Perlmutter, K, Saexinger, H, [1977], Fear of the Mentally ill: Empirical Support for the Common Man's Response, American Journal of Psychiatry, 134, 10, 1134-1137

Lamb, R, H, Weinberger, L, [1998], Persons with Severe Mental illness in Jails and Prisons: A Review, Psychiatric Services, 49, 4, 483-492

Lart, R, Swyer, B, [1997], Bridging the Gap, Community Care, 1-7 May, 30-31

Launer, M, [1996a], A Paper Trail of Tragedy, The Guardian, February 28, 6-7

Launer, M, [1996b], Positive Thoughts for Negative Minds, The Guardian, June 5, 6-7

Laurance, J [2002], Pure Madness: How Fear Drives The Mental Health System, Faculty of Public Health Medicine Annual Scientific Conference, King's Fund

Lawson, W, Yesaviage, A, J, Werner, P, [1984], Race, Violence and Psychopathology, Clinical Psychology, 45, 7, 294-297

Leff, J, Kuipers, L, Berkowitz, R, Eberlein-Vries, R, Sturgeon, D, [1982], A Controlled Trial of Social Intervention in the Families of Schizophrenic Patients, British Journal of Psychiatry, 141, 121-134

Leff, J, [1997], Care in the Community: Illusion or Reality?, Chicester and New York: John Wiley and sons

Leff, J, [2001a], The Unbalanced Mind, London: Weidenfeld and Nicolson

Leff, J, [2001b], Why is Care in the Community Perceived as a Failure, British Journal of Psychiatry, 79, 381-383

Lehane, M, Rees, C, [1996], What the Papers Say, Nursing Standard, April 3, 22-23

Lemert, E, [1974], Beyond Mead: The Societal Reaction to Deviance, Social Problems, 21, 457-468

Lerman, P, [1984], Policing Juveniles in London: Shifts in Guiding Discretion 1893-1968, British Journal of Criminology, April, 24, 2, 168-184

Lester, D, Pickett, C, [1978], Attitudes Toward Mental Illness in Police Officers, Psychological Reports, 42, 888

Liebling, A, [1990], Psychiatric Admission Rates, Criminal Justice Matters, Summer, 4, 10

Linehan, T, [1996a], Hidden Danger, Nursing Times, February 21, 16-17

Linehan, T, [1996b], Media Madness, Nursing Times, April 3, 31

Littlewood, R, Lipsedge, M, [1981], Acute Psychotic Reactions in Caribbean-born Patients, Psychological Medicine, 11, 303-318

Littlewood, R, Lipsedge, M, [1989], Aliens and Alienists: Ethnic Minorities and Psychiatry, London: Routledge

Littlewood, R, [1991], Against Pathology; The New Psychiatry and Its Critics, British Journal of Psychiatry, 159, 696-702

Livingstone, S, [1996], Prison Link to Increase in Crime, The Times Higher Educational Supplement, April, 9

Loader, I, Mulchany, A, [2001], The Power of Legitimate Naming, Part 1- Chief Constables as Social Commentators in Post War Britain, British Journal of Criminology, 41, 41-55

Loader, I, Mulchany, A, [2001], The Power of Legitimate Naming, Part II – Making Sense of the Elite Police Voice, British Journal of Criminology, 41, 252-265

Logsdail, S, Ellis, K, [1999], Police Training for the Management of Dangerous Patients, Psychiatric Bulletin, 23, 46-48

Lucken, K, [1998], Contemporary Penal Trends, Modern or Postmodern?, British Journal of Criminology, 38, 106-123

Lundman, R, J, [1974], Routine Police Arrest Practices: A Commonweal Perspective, Social Problems, 22, 127-141

Lustgarten, L, [1987], The Police and the Substantive Criminal Law, British Journal of Criminology, Winter, 27, 1, 23-30

MacDonald, I, [1999], Back in the Neighbourhood, Police Review, July 9, 24-25

Macmillan, J, F, [1987], Contact with the Police in Early Schizophrenia: Its Nature, Frequency and Relevance to the Outcome of Treatment, Medicine, Science and Law, 27, 3, 191-199

Mapp, S, [1995a], Stuck in the Past, Community Care, 16-22 February, 16

Mapp, S, [1995b], Mental Divisions, Community Care, 5-11 October, 27

Matthews Jr, A, [1970], Observations of Police Policy and Procedures for Emergency Detention of the Mentally ill, Journal of Criminal Law, Criminology and Police Science, 61, 283-295

Mastrofski, S, Parks, R, [1990], Improving Observational Studies of Police, Criminology, 28, 3, 475-496

Mastrofski, S, Worden, E, R, Snipes, B, J, [1995], Law Enforcement in a Time of Community Policing, Criminology, 33, 4, 539-563

Mayhew, P, Elliot, D, Dawds, L, [1989], The 1988 British Crime Survey, Home Office Research Study III, London: HMSO

McBride, A, [1990], Mental Health Effects of Women's Multiple Roles, American Psychologist, 45, 3, 381-384

McCann, G, McKeown, M, [1995], Identifying the Needs of Relatives of Forensic Patients, Nursing Times, June 14, 35-37

McConville, M, Sanders, A, Leng, R, [1991], The Case for the Prosecution: Police Suspects and the Construction of Criminality, London: Routledge

McConville, M, Sanders, A, Leng, R, [1997], Descriptive or Critical Sociology: The Choice is Yours, British Journal of Criminology, Summer, 37, 3, 347-358

McCord, W, McCord, J, [1956], Psychopathy and Delinquency, London and New York: Grune and Stratton

McGovern, D, Cope, R, [1987], The Compulsory Detention of Males of Different Ethnic Groups, with Special Reference to Offender Patients, British Journal of Psychiatry, 150, 505-512

McGurk, J, B, Thornton, D, Williams, M, [1987], Applying Psychology to Imprisonment: Theory and Practice, London: HMSO

McLaughlin, E, Muncie, J, [1996], Controlling Crime, London: Sage in association with The Open University

McNaught, A, [1987], Health Action and Ethnic Minorities, London : Bedford Square Press

Meehan, J, A, [1995], From Conversion to Coercion: The Police Role in Medication Compliance, Psychiatric Quarterly, 66, 2, 163-184

Melossi, D, [2000], Changing Representations of the Criminal, British Journal of Criminology, 40, 296-320

Mendelson, F, E, Frost, C, [1994], An Alternative to the Panel Scheme for the Diversion of Mentally Disordered Offenders - The Southampton Facilitator Approach, Psychiatric Bulletin, 18, 39-40

Mental Health Unit, [1996], Mentally Disordered Offenders, Sentencing and Discharge Arrangements: A Discussion Paper on a Proposed New Power for the Courts, Home Office and the Department of Health, London: HMSO

Menzies, J, R, [1987], Psychiatrists in Blue: Police Apprehension of Mental Disorder and Dangerousness, Criminology, 25, 3, 429-453

Metropolitan Police, [1994], Understanding Mental Disorder - Distance Learning Module, London: Metropolitan Police

Meyer Jr, J, [1972], Social Aspects of Peace keeping: Handling Personal Crisis Situations, Police, 16, 3, 41-45

Miller, D, R, [1992], Economic Factors Leading to Diversion of the Mentally Disordered from the Civil to the Criminal Commitment Systems, International Journal of Law and Psychiatry, 15, 1-12

MIND, [1995a], Civil Admission to Hospital, MIND Rights Guide, London: MIND Publications

MIND, [1995b], Mental Health and the Police, MIND Rights Guide, London: MIND Publications

MIND, [1996], MIND's Policy on People with Mental Health Problems and the Criminal Justice System, London: MIND publications

Mitchell, M, Munro, A, [1996], The Influence of the Occupational Culture on How Police Probationers Learn to Deal with Incidents of Sudden Death, Issues in Criminological and Legal Psychology, 25, 47-53

Moir, A, Jessel, D, [1995], A Mind to Crime, London : Channel 4/Michael Joseph

Monahan, J, Caldeira, C, Friedlander, D, H, [1979], Police and the Mentally ill: A Comparisons of Committed and Arrested Persons, International Journal of Law and Psychiatry, 2, 509-518

Monahan, J, [1984], The Prediction of Violent Behaviour: Toward a Second Generation of Theory and Policy, American Journal of Psychiatry, 141, 1, 10-15

Morgan, R, Newburn, T, [1997], Time to Change Direction, Police Review, March 14, 16-18

Morgan, R, Newburn, T, [1998], The Future of Policing, Oxford : Clarendon

Morrall, P, [2000], Madness and Murder, London: Whurr Publishers

Morris, N, Torny, M, [1979], Crime and Justice : An Annual Review of Research, Volume 1, Chicago and London : The University of Chicago Press

Morris, N, Torny, M, [1990], Between Prison and Probation : Intermediate Punishments in a Rational Sentencing System, Oxford : Oxford University Press

Morris, S, Steadman, J, H, Veysey, B, [1997], Mental Health Services in United States Jails: A Survey of Innovative Practice, Criminal Justice and Behaviour, 24, 1, 3-19

Mullen, P [1999], Dangerous People with Severe Personality Disorder, British Medical Journal, 319, 1146-1147

Muncie, J, McLaughlin, E, [1996], The Problem of Crime, London: Sage in association with the Open University

Munro, E, Rumgay, J [2000], Roles of Risk Assessment in Reducing Homicides by People with Mental Illness, British Journal of Psychiatry, 176, 16-120

Murphy, E, [1991], After the Asylums; Community Care for People with Mental Illness, London: Faber and Faber

Murray, J, D, [1989], Review of Research on Re-offending of Mentally Disordered Offenders, Home Office Research and Planning Unit, Paper 55, London: HMSO

Murray, V, Walker, W, H, Mitchell, C, Pelosi, J, A, [1996], Needs for Care From A Demand Led Community Psychiatric Service: A Study of Patients with Major Mental Illness, British Medical Journal, 312, 1582-1586

Murrell, E, M, Lester, D, Arcuri, A, [1978], Is the 'Police Personality' Unique to Police Officers, Psychological Reports, 43, 1, 298

NACRO, [1989], Some Facts and Findings about Black People in the Criminal Justice System, London: NACRO

NACRO, [1991], The Resettlement of Mentally Disordered Offenders, London : NACRO

NACRO, [1993a], Mentally Disordered Prisoners at Winson Green, London: NACRO

NACRO, [1993b], Community Care and Mentally Disordered Offenders, Policy Paper 1, London: NACRO

NACRO, [1993c], Criminal Justice Act 1991, Section 95, Report of the NACRO Race Issues Advisory Committee, London : NACRO

NACRO, [1994], Diverting Mentally Disordered Offenders from Custodial Remands and Sentences, Policy Paper 3, London : NACRO

NACRO, [1995], Mentally Disturbed Prisoners, Policy Paper 4, London: NACRO

NAPO, [1993], Acting Upon the Reed Report Conference, NAPO News, 4 National Practitioners Group, [1995], An Investigation into the Operation of Section 136 of the Mental Health Act 1983 in the West Midlands, NPG Newsletter, July

Nelken, D, [1994], The Futures of Criminology, London: Sage

Neocleous, G, [2000], Social Police and the Mechanisms of Prevention, British Journal of Criminology, 40, 710-726

Neuberger, J, [1997], Safety Limits, OpenMind, January/February, 83, 10-11

Newman, P, [1998], Safety Conscious, Police Review, December 18, 22-23

Newman, P, [1999], Frontline Deterrent, Police Review, March 12, 22-23

NHS Management Executive, [1993], The Health of the Nation, Mentally Disordered Offenders, London : Department of Health

Noak, J, House, M, [1997], Assessment of the Risks Posed by People with Mental Illness, Nursing Times, January 1, 34-36

Norfolk, A, G, [1997], Fitness to be Interviewed - A Proposed Definition and Scheme of Examination, Medicine, Science and Law, 37, 3, 228-234

Nzegwu, F, [1993], Black People and Health Care in Contemporary Britain, England: Institute of Black Research

Ogden, J, [1997], Revolving Door, Community Care, 10-16 April, 10-11

Olfson, M, [1987], A Characterisation of Police Referred Psychiatric Emergencies American Journal of Forensic Psychiatry, 8, 3, 5-13

O'Malley, P, [1996], Post Social Criminologies; Some Implications of Current Political Trends for Criminological Theory and Practice, Current Issues in Criminal Justice, 8, 1, 26-38

OpenMInd, [1995], Care not Coercion, Policy, Parliament and Legal Update, OpenMind, October/November, 77, 4

O'Strow, M, J, [1981], Culture as a Fundamental Dimension of Experience: A Discussion of Pierre Bourdieu's Theory of Human Habitus, Human Studies, 4, 3, 279-297

Parsons, T, [1968], The Structure of Social Action, Volume 1, New York, The Free Press

Patch, C, P, Arrigo, A, B, [1999], Police Officer Attitudes and the Use of Discretion in Situations Involving the Mentally ill: The Need to Narrow the Focus, International Journal of Law and Psychiatry, 22, 1, 23-35

Pedler, M, [1999a], Mental Health Act Review, OpenMind, January/February, 95, 4

Pedler, M, [1999b] Dobson's Christmas Present, OpenMind, January/February, 95, 4

Penal Affairs Consortium, [1994], The Path to Community Prisons, London: NACRO

Penal Affairs Consortium, [1995a], Prison Overcrowding, London: NACRO

Penal Affairs Consortium, [1995b], A Joint Manifesto for Penal Reform, London: NACRO

Peryer, D, [1995], Comment : Care not Control - Combating Coercion, OpenMind, October/November, 77, 5

Phillips, R, M, Wolf, S, A, Coons, J, D, [1988], Psychiatry and the Criminal Justice System: Testing the Myths, American Journal of Psychiatry, 145, 5, 605-610

Phillips, T, Smith, P, [2000], Police Violence Occasioning Citizen Complaint, British Journal of Criminology, 40, 480-496

Pilgram, D, [1999], Badly in Need of Help - Psychopathy and Treatability, OpenMind, January/February, 95, 7

Piliavin, I, Briar, S, [1964], Police Encounters with Juveniles, American Journal of Sociology, 70, 206-214

Pitts, J, [1988], The Politics of Juvenile Crime, London : Sage

Pizanias, C, [1996], Habitus: From the Inside Out and the Outside In, Theory and Psychology, 6, 4, 647-665

Pogrebin, R, M, [1986], Police Response for Mental Health Assistance, Psychiatric Quarterly, 58, 66-73

Police Review, [1996a], Discrimination in Criminal Justice System Highlighted, August 16, 8

Police Review, [1996b], Destruction of Old Files hampers Risk assessment, Police Review, September 13, 15

Police Review, [1996c], Calls for Mental Health Liaison Officers in All Forces, Police Review, November 5, 6

Police Review, [1997a], Claims of Poor Policing Bias Dismissed, Police Review, July 25, 4

Police Review, [1997b], New Home Office Key Objectives Announced, Police Review, November 7, 5

Police Review, [1998], Concern Over Offenders Using Mental Illness to Evade Justice, Police Review, April 17, 8

Police Review, [1999], PCA Concern over Effect of CS Spray on Those Taking Anti-psychotic Drugs, Police Review, July 16, 5

Powell, D, D, [1990], A Study of Police Discretion in Six Southern Cities, Journal of Police Science and Administration, 17, 1, 1-7

Prins, H, [1980], Offenders: Deviants or Patients? , London: Tavistock Publications

Prins, H, [1986], Dangerous Behaviour, The Law and Mental Disorder, London: Tavistock Publications

Prins, H, [1996], Can the Law Serve as the Solution to All Social Ills? The Case of the Mental Health (Patients in the Community) Act 1995, Medicine, Science and Law, 36, 3, 217-220

Prior, C, [1997], Forecasting the Future, Community Care, 23-29 October, 10-11

Prior, C, [1998], Future Imperfect ?, Community Care, 26 February-4 March, 22-3

Prison Reform Trust, [1985], Prison Medicine: Ideas on Health Care in Penal Establishments, London: Prison Reform Trust

Pugh, M, G, [1986], The Good Police Officer: Qualities, Roles and Concepts, Journal of Police Science and Administration, 14, 1, 1-5

Pullinger, H, [1985], The Criminal Justice System Model: The Flow Model, Home Office Research and Planning Unit, Paper 36, London: HMSO

Rabinow, P, [1991], The Foucault Reader, An Introduction into Foucault's Thought, London: Penguin

Rack, P, [1982], Race, Culture and Mental Disorder, London: Routledge

Radzinowicz, L, Wolfgang, E, M, [1977], Crime and Justice: Volume II The Criminal in the Arms of the Law, New York: Basic Books

Raine, A, [1993], The Psychopathology of Crime : Criminal Behaviour as a Clinical Disorder, San Diego : Academic Press

Reaside Clinic and West Midlands Police, [1992], Diversion at the Point of Arrest: A Proposal for a West Midlands Service, Reaside Clinic/West Midlands Police

Reed, B, [1980], Issues and Trends in Police Discretion : A Tool for Police at Every Level of Authority, The Police Chief, November, 54-59

Reed, J, [1992], Review of Health and Social Services for Mentally Disordered Offenders and others requiring similar services, Home Office and Department of Health, London: HMSO

Reed, J, [1996], Psychopathy - A Clinical and Legal Dilemma, British Journal of Psychiatry, 168, 4-9

Reedy, M, W, [1997], Against Constructionism: The Historical Ethnography of Emotions, Current Anthropology, 38,3, 327-351

Reiner, R, [1992a], Chief Constables, Oxford: Oxford University Press

Reiner, R, [1992b], The Politics of the Police, New York and London; Harvester/Wheatsheaf

Reiss Jr, J, A, [1984], Consequences of Compliance and Deterrence Models of Law Enforcement for the Exercise of Discretion, Law and Contemporary Problems, 47, 83-122

Rhydderch, R, [1999], Nina Mackay Killing Was 'Avoidable', Police Review, April 23, 11

Rickford, F, [1995a], Law and Disorder, Community Care, 8-14 June, 18-19

Rickford, F, [1995b], Over Dose, Community Care, 5-11 October, 16-17

Rickford, F, [1995c], Contrasting Opinions, Community Care, 30 November - 6 December, 2-3

Riordan, S, Wix, S, [1999], Diversion of Mentally Disordered Offenders - To What and When?, British Journal of Forensic Practice, 1, 3, 23-26

Ritchie, H, J, Dick, D, Lingham, R, [1994], The Report of the Inquiry into the Care and Treatment of Christopher Clunis, North East and South East Thames Health Regional Authorities, London: HMSO

Ritzer, G, [1992], Sociological Theory, New York: McGraw-Hill

Rix, J, B, K, [1997], Fit to Be Interviewed by the Police?, Advances in Psychiatric Treatment, 3, 33-40

Robertson, G, [1988], Arrest Patterns among Mentally Disordered Offenders, British Journal of Psychiatry, 153, 313-316

Rock, P, McIntosh, M, [1974], Deviance and Social Control, London : Tavistock Publications

Rock, P [1998], Murderers, Victims and 'Survivors', The Social Construction of Deviance, British Journal of Criminology, 38, 185-200

Rofman, S, E, Askinazi, C, Fant, E, [1980], The Prediction of Dangerous Behaviour in Emergency Civil Commitment, American Journal of Psychiatry, 137, 9, 1061-1064

Rogers, A, Faulkner, A, [1987], A Place of Safety: MIND's Research into Police Referrals to the Psychiatric Services, London: MIND Publications

Rogers, A, Pilgrim, D, Lacey, R, [1993], Experiencing Psychiatry: Users' Views of Services, Basingstoke and London: MIND/Macmillan

Rooney, J, [1998], Safety Measures, Health Service Journal, January 15, 26-27

Rose, D, Muijen, M, [1997], Nursing Doubts, Health Service Journal, June 26, 34-35

Ross, H, M, Jacobs, M, Monteleane, M, Olsen, R, Surber, R, Winlder, E, Wommack, A, [1986], Characteristics of Psychiatric Patients who Engage in Assaultive or Other Fear Inducing Behaviours, The Journal of Nervous and Mental Disease, 174, 3, 154-160

Roth, M, Bluglass, R, [1985], Psychiatry, Human Rights and the Law, Cambridge : Cambridge University Press

Roughton, A, [1994], An Investigation into the Operation of Section 136 of the Mental Health Act 1983 in the West Midlands, West Midlands Police

Rubington, E, Weinberg, S, M, [1973], Deviance: The Interactionist Perspective, London: Macmillan

Rudat, K, Barnes, S, [1995], The HEA Health and Lifestyle Survey, SHARE, Kings Fund Centre, 11, 4-5

Ruiz, J, [1993], An Interactive Analysis Between Uniformed Law Enforcement Officers and the Mentally ill, American Journal of Police, 12, 4, 149-177

Russo, N, F, [1990], Overview: Forging Research Priorities for Women's Mental Health, American Psychologist, 45, 3, 368-373

Rutherford, A, [1993], Criminal Justice and the Pursuit of Decency, Oxford : Oxford University Press

Sales, N, G, [1991], A Comparison of Referrals by Police and Other Sources to a Psychiatric Emergency Service, Hospital and Community Psychiatry, 42, 9, 950-952

Sashidharan, P, S, Smyth, M, Owen, A, [1999], PRISM Psychosis Study: Thro' a Glass Darkly - A Distorted Appraisal of Community Care, British Journal of Psychiatry, 175, 504-507

Satsumi, Y, Toshiya, I, Yamauchi, T, [1999], Criminal Offences among Discharged Mentally III People: Determinants of the Duration from Discharge and Absence of Diagnostic Specificity, International Journal of Law and Psychiatry, 21, 2, 197-207

Scarman, Lord, [1986], The Scarman Report : The Brixton Disorders, 10-12 April 1981, London : Penguin

Scheff, J, T, [1966], The Societal Reaction to Deviance: Ascriptive Elements in the Psychiatric Screening of Mental Patients in a Midwestern State, Social Problems, 11, 401-413

Scheingold, A, S, [1984], The Politics of Law and Order, New York and London: Longman

Scott, S, Johnson, P, Thornicroft, G, Marshall, J, Bindham, P, Bebbington, E, [1998], Substance Misuse and Risk of Aggression and Offending among the Severely Mentally Ill, British Journal of Psychiatry, 172, 345-350

Sexton, R, [1996], Bridging the Perception Gap, Police Review, September 13, 28-29

Shapland, J, Vagg, J, [1987], Using the Police, British Journal of Criminology, Winter, 27, 1, 54-63

Shapland, J, [1991], Where Do We Put Them? Coping with Mentally Disordered Offenders, Paper to British Criminology Conference, York University, July 1991

Shearing, D, C, Leon, S, J, [1977], Reconsidering the Police Role: A Challenge of a Popular Conception, Canadian Journal of Criminology and Corrections, October, 331-345

Shearing, D, C, Ericson, V, R, [1991], Culture as Figurative Action, British Journal of Sociology, 42, 4, 481-506

Shephard, E, [1996], The Trouble with PEACE, Police Review, July 26, 14-15

Shephard, G, Hill, R, [1996], Manic Depression : Do People Receive Adequate Support, Nursing Times, June 26, 42-43

Sherman, W, L, [1980], Causes of Police Behaviour: The Current State of Quantitative Research, Journal of Research in Crime and Delinquency, January, 69-100

Shusterman, R, [1999], Bourdieu: A Critical Reader, Oxford: Blackwell

Silver, I, [1999], Role Transitions, Objects and Identity, Symbolic Interaction, 19, 1, 1-20

Sivanandan, A, [1986], Asian and Afro-Caribbean Struggles in Britain, Race and Class, London: Institute of Race Relations

Skellington, R, Morris, P, [1992], Race in Britain Today, London : Sage/Open University

Skolnick, H, J, [1975], Justice without Trial: Law Enforcement in Democratic Society, New York and London: John Wiley and Sons

Smaje, C, [1995], Health, Race and Ethnicity: Making Sense of the Evidence, London: King's Fund Centre

Smith, A, D, Visher, A, C, [1981], Street Level Justice: Situational Determinants of Police Arrest Decisions, Social Problems, 29, 2, 167-177

Smith, J, D, [1997], Case Construction and the Goals of Criminal Process, British Journal of Criminology, 37, 3, 319-345

Smith, P, J, [1990], Police are Best at Community Care of Mentally Ill People in England, Journal of Advanced Nursing, 15, 1117

Snell, J, [1995], Diversion from Prison Part of Care Programmes, Community Care, 5-11 October, 2

Social Services Inspectorate/ Department of Health, [1996], Mentally Disordered Offenders: Improving Services, London: HMSO

Sone, K, [1993], Black Lives, White Perspectives, Community Care, March 25, 14-15

Staite, C, Martin, N, Bingham, M, Daly, R, [1994], Diversion from Custody for Mentally Disordered Offenders, Harlow: Longham

Stanistreet, D, Jeffrey, V, [1996], The Report of the Symposium on 'Suicide

Audit in Practice', Wirral Health Authority/Department of Public Health

Steadman, J, H, Morrissey, P, J, Braff, J, Monahn, J, [1986], Psychiatric Evaluations of Police Referrals in a General Hospital Emergency Room, International Journal of Law and Psychiatry, 8, 39-47

Steadman, J, H, Mulvey, P, E, Monahan, J, Clark Robbins, P, M, Applebaum, S, P, Grisso, T, Roth, H, L, Silver, E, [1998], Violence by People Discharged From Acute Psychiatric Inpatient Facilities and by Others in the Same Neighbourhood, Archives of General Psychiatry, 55, 393-401

Steel, L, [1998], Introducing the Care Enforcers, Community Care, 6-12 August, 8-9

Stein, I, L, Diamond, J, R, [1985], The Chronic Mentally III and the Criminal Justice System: When to Call the Police, Hospital and Community Psychiatry, 36,31, 71-274

Stephens, M, Becker, S, [1994], Police Force, Police Service : Care and Control in Britain, Basingstoke : Macmillan

Stephens, M [2002], A Caring Role for the Police in First Contact with the Mentally ill, The Journal of Primary Care Mental Health, 6, 1, 4-5

Stirling, A, [1995], Windsor's Progress, Nursing Times, May 10, 44-45

Stone, M, [1999], Mind Reading, Police Review, January 8, 26-27

Stradling, S, G, Tuohy, A, P, Harper, K, J, [1990], Judgmental Asymmetry in the Exercise of Police Discretion, Applied Cognitive Psychology, 4, 409-421

Strand, G, J, Smith Benjamin, L, [1997], Resistance to Change in Individuals with Personality Disorder, Current Opinion in Psychiatry, 10, 2, 132-135

Strong, S, [1995a], Sensitive Training on Offer to Police, Community Care, 12-18 January, 8

Strong, S, [1995b], Card Carriers, Community Care, 1-7 June, 7

Sudnow, D, [1965], Normal Crimes: Sociological Features of the Penal Code in a Public Defender Office, Social Problems, 12, 255-276

Sullivan, R, R, [1998], The Politics of British Policing in the Thatcher/Major State, The Howard Journal of Criminal Justice, 37, 3, 306-318

Surrey Mentally Disordered Offenders Project, [1993], MDO Handbook: To Provide a Guide to Those Working with Mentally Disordered Offenders, Guildford: Mentally Disordered Offenders Project

Susskind, L, McMahon, G, [1985], The Theory and Practice of Negotiated Rulemaking, Yale Journal of Regulation, 3, 133-165

Swaithe, G, [1997], Brand Loyalty, Police Review, September 26, 27-28

Swanson, M, [1996], Special Solutions, Nursing Times, April 3, 28-30

Swope, R, [1998], Secured Capital, Police Review, February 27, 22-24

Sykes, E, R, Clark, P, J, [1975], A Theory of Deference Exchange in Police-Citizen Encounters, American Journal of Sociology, 81, 3, 584-600

Symonds, B, [1998], The Philosophical and Sociological Context of Mental Health Care Legislation, Journal of Advanced Nursing, 27, 946-954

Szasz, S, T, [1974], Law, Liberty and Psychiatry: An Inquiry into the Social Uses of Mental Health Practices, London: Routledge and Kegan Paul

Szmukler, G, [2000] Homicide Inquiries, Psychiatric Bulletin, 24, 6-10 Taft Jr, B, P, [1980], Dealing with Mental Patients, Police Magazine, 3, 1, 21-27

Taylor, B, M, [1996], Editorial: Taking Over After The Asylum - The Police and Care in the Community, Criminal Behaviour and Mental Health, 6, 299-303

Taylor, I, Walton, P, Young, J, [1973], The New Criminology: For a Social Theory of Deviance, London: Routledge and Kegan Paul

Taylor, J, P, Gunn, J, [1999], Homicides By People With Mental Illness: Myth and Reality, British Journal of Psychiatry, 174, 9-14

Taylor, P [2001], Mental Illness and Serious Harm to Others, NHS National Programme on Forensic Mental Health Research and Development, HMSO

Teevan, S, [1996], Deaths Caused by Psychiatric Drugs, OpenMind, February/March, 79, 6

Teplin, A, L, [1985], The Criminality of the Mentally III: A Dangerous Misconception, American Journal of Psychiatry, 142, 5, 593-599

Teplin, A, L, Pruett, S, N, [1992], Police as Streetcorner Psychiatrist: Managing the Mentally ill, International Journal of Law and Psychiatry, 15, 139-156

Terry, W, C, [1985], Police Stress as a Professional Self Image, Journal of Criminal Justice, 13, 6, 501-512

Thomas, T, [1994], The Police and Social Workers, Aldershot: Arena

Thompson, A, Sylvester, R, [1998], Care in the Community is Scrapped, The Daily Telegraph, January 17, 1,

Thompson, T, Mathias, P, [1994], Lyttle's Mental Health and Disorder, London: Bailliere Tindall

Tyrer, P, Steinberg, D, [1993], Models for Mental Disorder: Conceptual Models in Psychiatry, Chicester: John Wiley and Sons

Tyrer, P, Duggan, C, Coid, J [2003], Ramifications of Personality Disorder in Clinical Practice, British Journal of Psychiatry, 182, 1-2

Valios, N, [1998], Dobson Confirms Mental Health Law Under Review, Community Care, 22-28 January, 1

Vaughan, S, M, [1992], Problem Orientated Policing: A Philosophy of Policing for the 21st Century, Criminal Justice and Behaviour, 19, 3, 343-354

Vaughan, J, P, Badger, P, [1995], Working with the Mentally Disordered Offender in the Community, London: Chapman and Hall

Vaughan, B, [2001], Handle with Care, On the Use of Structuration Theory in Criminology, British Journal of Criminology, 41, 185-200

Vereker, D, [1992], Over 6,000 Psychiatric Reports on Defendants Last Year, NAPO News, August, 42, 8-9

Vinzant, J, Crothers, L, [1994], Street Level Leadership: The Role of Patrol Officers in Community Policing, Criminal Justice Review, 19, 2, 189-211

Wacquart, J, D, L, [1989], Toward a Reflexive Sociology: A Workshop with Pierre Bourdieu, Sociological Theory, 26-63

Waddington, J. A. P. [1993], Calling the Police, Aldershot: Avebury

Waddington, J, A, P, [1998], More Than Good Citizens, Police Review, January 13, 25-26

Waddington, J, A, P, [1999], Police (canteen) Subculture: An Appreciation, British Journal of Criminology, Spring, 39, 2, 287-309

Waegel, B, W, [1981], Case Routinization in Investigative Police Work, Social Problems, 28, 3, 263-275

Wain, N, [2000], May the Force be with You, Community Care, 31 Aug-6 Sept, 22-23

Wallace, C, Mullen, P, Burgess, P, Palmer, S, Ruschena, D, Browne, C, [1998], Serious Criminal Offending and Mental Disorder, British Journal of Psychiatry, 172, 477-484

Walsh, E, Buchanan, A, Fahy, T [2002], Violence and Schizophrenia: Examining the Evidence, British Journal of Psychiatry, 180, 490-495

Waters, J, [1995a], No Way Out, Nursing Times, Dec 6, 20-21

Waters, J, [1995b], Patients Dying of Faulty Drug Doses, Nursing Times, December 20, 6

Watson, [1995], Guardianship, OpenMind, December/January, 78, 18-19

Wessely, S, Taylor, P, J, [1991], Madness and Crime: Criminology versus Psychiatry, Criminal Behaviour and Mental Health, 1, 193-228

West Cheshire NHS Trust, [1994], The Mental Health Guide to Services in Chester and Ellesmere Port, Chester and Ellesmere Port Community Health Council

West, D, [1990], Forensic Psychiatry and Criminal Policy, Criminal Justice Matters, Summer, 4, 9

White, M, [1997], Courting Disaster, Community Care 18 Dec-7 Jan, 9

Whiteley, P. [1995a], An Issue of Protection, Community Care, 15-21 June, 7

Whiteley, P, [1995b], Death by Communication Breakdown, Community Care, 28 Sept - 4 Oct, 3

Whiteley, P, [1995c], Is there Care in the Community Crisis, Community Care, 5-11 October, 8-9

Whiteley, P, [1995d], Mental Health Reforms likely, Community Care, 23-29 November, 9

Whiteley, P, [1996a], High Death Rate of Psychiatric Patients, Community Care, 5-10 January, 4

Whiteley, P, [1996b], Clunis Report Puts Pressure on Council, Community Care, 25-31 January, 2

Whiteley, P, [1996c], A Change Too Many ?, Community Care, 3-9 October, 14

Whiteley, P, [1997a], Mental Health Alliance to Lobby Government, Community Care, 23-29 January, 1

Whiteley, P, [1997b], Forced Together, Community Care, 13-19 February, 10

Whiteley, P, [1997c], Cash Crisis Blamed for Hike in Mental Health Detentions, Community Care, 20-26 March, 2

Williams, A, [1997], Retaking Control, Police Review, August 1, 18-19

Wilson, J, [1968], Varieties of Police Behaviour, Cambridge, Massachusetts: Harvard University Press

Wilson, J, [1996], Power Games, Community Care, 9-15 May 22-23

Wilson, M, [1997], Printing it in Black and White, OpenMind, May/June, 85, 16-17

Wilt, G, M, Bannon, D, J, [1976], Cynicism or Realism: A Critique of Niederhoffer's Research into Police Attitudes, Journal of Police Science and Administration, 4, 1, 38-45

Wix, S, [1994], Keeping on the Straight and Narrow: Diversion of Mentally Disordered Offenders at the Point of Arrest, Psychiatric Care, 1, 3, 102-104

Wortley, R, [1997], Attributions as a Function of Expertise: The Case of the Police Decision to Arrest, Journal of Applied Social Psychology, 27, 6, 525-538

Yates, M, Williams, J, Working in Harmony, Nursing Standard, April 3, 24-25

Zito, J, Howlett, M, [1995], Both Sides of the Coin, Community Care, 30 Nov - 6 Dec, 4

Zito, J, [1998], Lesson from the Outside, Nursing Times, January 28, 34-35